Neural and Behavioural Effects of Bilingualism

on Selective Attention



Jacqueline Ann Phelps

Lucy Cavendish College

University of Cambridge

This thesis is submitted for the Degree of Doctor of Philosophy September 2022

Declaration

This thesis is the result of my own work and includes nothing which is the outcome of work done in collaboration except as declared in the preface and specified in the text. It is not substantially the same as any work that has already been submitted before for any degree or other qualification except as declared in the preface and specified in the text. It does not exceed the prescribed word limit for the School of Biology Degree Committee.

Abstract

Bilingualism has been shown to modulate the neural mechanisms of selective attention, with differences between monolinguals and bilinguals observed even when they display equivalent behavioural performance in a selective attention task. This suggests that the crucial consequence of learning and using multiple languages might be that it triggers neuroplastic adaptation that allows bilinguals to achieve optimal performance under increased processing demands. This functional plasticity yielding equivalent outcomes (also known as degeneracy) is a common feature in biological systems, allowing flexible adaptation to changing environments.

Yet the exact mechanism by which bilingualism affects selective attention is still not entirely clear. While the currently dominant view suggests that the need for constant management of competing languages in bilinguals increases attentional capacity; another possibility is that this language control may be drawing on the available attentional resources such that they need to be economised to support optimal task performance. Another question concerns the development of this adaptation over time, where the demands of competition and inhibition between co-activated languages might be reconfiguring the patterns of attentional processes right from the onset, such that the effects can be seen by the time children can respond to selective attention tasks. Alternatively, these modifications might have a protracted maturation dependent on the length and intensity of exposure to the demands of bilingualism, in which case they would manifest differently in adults and in children, as well as in speakers with different levels of exposure to L2. Finally, another aspect is to establish the extent to which these modifications might affect attentional processing beyond the language domain, extending to auditory processing more generally. Here I present a series of behavioural and neuroimaging experiments that address these questions.

To investigate whether bilingualism enhances attentional processing or triggers redistribution of the existing capacity, I used EEG to track the neural encoding of

attended continuous speech in monolingual and bilingual children aged 7-12, in the context of different types of acoustic and linguistic interference. Participants attended to a narrative in English while four different types of interference were presented to the unattended ear. The neural encoding of attended and unattended streams was assessed by reconstructing their speech envelopes from the EEG data in each condition, using the mTRF toolbox. Results showed more accurate reconstruction of the attended envelopes than ignored ones across all conditions for both bilinguals and monolinguals. Critically however, there was no evidence of enhanced attentional processing in bilinguals; instead data showed a pattern consistent with redistribution of the available capacity, economised to achieve optimal performance on the selective attention task. The follow up behavioural experiments tested the limits of this adaptation by using a dual task (dichotic listening + visual attention) to further increase processing load. The results over three experiments (on children, and adults with different levels of exposure to L2) showed consistently comparable performance on both tasks for monolingual and bilingual adults, suggesting that bilingual adaptation can accommodate high processing loads. However there were also subtle differences in performance on the secondary (visual) task between the monolingual and bilingual children, and across the two groups of bilingual adults, suggesting that maturation and exposure do exert influence on this functional adaptation. The findings of the final EEG study on auditory processing beyond language domain indicate comparable but attenuated modification of attentional processing in bilinguals, compared to the first EEG study using linguistic interference.

Findings from all experiments are explored in the context of theories of selective attention and bilingualism.

Acknowledgements

I would like to thank my supervisor, Dr Mirjana Bozic, for her guidance and unfailing support throughout this PhD. I would also like to thank my advisor, Dr Tristan Bekinschtein and the members of his lab who have helped me with technical advice on EEG. I am especially grateful to Dr Adam Attaheri for all his time and patience when advising on the mTRF toolbox.

I would also like to acknowledge my family, especially my husband and three children, whose unwavering support and encouragement carried me through this PhD.

Many thanks to all the participants and parents I have recruited throughout this process, for their enthusiastic participation in my experiments.

Finally, I would like to thank Professor Roberto Filippi, who encouraged me to embark on this PhD and for his support throughout.

Table of Contents

LIST OF TABLES	10
LIST OF FIGURES	12
CHAPTER 1: INTRODUCTION	12
1.1 Тне Аім	14
1.2 SELECTIVE ATTENTION	14
1.2.1 THEORIES OF SELECTIVE ATTENTION PROCESSING	15
1.2.2 ATTENTIONAL CAPACITY	17
1.2.3 NEURAL BASES OF SELECTIVE ATTENTION IN ADULTS	18
1.2.4 DEVELOPMENTAL CONSIDERATIONS	18
1.3 BILINGUALISM	19
1.4 DEGENERACY	22
1.5 DOMAIN GENERALITY	23
1.6 THESIS OUTLINE	24
CHAPTER 2: LITERATURE REVIEW	27
2.1 NEURAL BASIS FOR FIRST AND SECOND LANGUAGE PROCESSING	27
2.2 SELECTIVE ATTENTION	29
2.2.1 THE NEURAL BASIS FOR VISUAL SELECTIVE ATTENTION	31
2.2.2 AUDITORY SELECTIVE ATTENTION AND NEURAL ENTRAINMENT	32
2.3 NEUROANATOMICAL AND COGNITIVE DEVELOPMENT OVERVIEW	35
2.3.1 DEVELOPMENT OF SELECTIVE ATTENTION - VISUAL	37
2.3.2 DEVELOPMENT OF SELECTIVE ATTENTION – AUDITORY	37
2.4 NEUROPLASTICITY AND BILINGUALISM	41
2.5 SUMMARY	47
CHAPTER 3: METHODS	49
3.1 INTRODUCTION	49
3.2. DICHOTIC LISTENING TASK	49

3.2.1 Overview	49
3.2.2 DICHOTIC LISTENING AND SELECTIVE ATTENTION	50
3.2.3 TASK DESIGN	51
3.3 EEG	55
3.3.1 Overview	55
3.3.2 EEG IN SPEECH AND ATTENTION TASKS	56
3.3.3 THE USE OF EEG IN THIS RESEARCH	57
3.3.4 Procedure	57
3.3.5 Preprocessing	57
3.3.6 MTRF	59
3.3.7 Speech envelopes	63
3.4 DUAL TASK	64
3.4.1 OVERVIEW	64
3.4.2 The visual task (T.O.V.A.)	65
CHAPTER 4: EXPERIMENT 1	68
4.1 INTRODUCTION	68
4.2 CURRENT STUDY	72
4.3 MATERIALS AND METHOD	76
4.3.1 PARTICIPANTS	76
4.3.2 DESIGN	77
4.3.3 TARGET STREAMS	77
4.3.4 DISTRACTOR STREAMS	78
4.3.5 PROCEDURE	79
4.3.6 DATA COLLECTION AND PREPROCESSING	80
4.3.7 Speech envelopes	81
4.3.8 ANALYSES	81
4.4 RESULTS	83
4.4.1 BEHAVIOURAL COMPREHENSION SCORES	83
4.4.2 EEG DATA	86
4.4.3 THE EFFECTS OF ATTENTION ON SPEECH RECONSTRUCTION ACCURACY	86
4.4.4 RECONSTRUCTION ACCURACY OF ATTENDED STREAMS IN MONOLINGUALS AND BILINGUALS	88
4.4.5 RECONSTRUCTION ACCURACY OF UNATTENDED STREAMS IN MONOLINGUALS AND BILINGUALS	91
4.5 DISCUSSION	92

CHAPTER 5: EXPERIMENT 2	101
5.1 INTRODUCTION	101
5.2 MATERIALS AND METHOD	106
5.2.1 Study 1 Participants	107
5.2.2 Study 2 Participants	108
5.2.3 Study 3 Participants	109
5.2.4 DESIGN	110
5.2.5 Study 1 (children): Procedure	113
5.2.6 Studies 2 and 3 (adults): Procedure	114
5.3 ALL STUDIES: ANALYSES	115
5.3.1 DICHOTIC LISTENING TASK	115
5.3.2 VISUAL TASK	115
5.4 RESULTS	117
5.4.1 Study 1 (Children)	117
5.4.2 Study 2 (HIGH-EXPOSURE BILINGUAL ADULTS)	122
5.4.3 Study 3 (low-exposure bilingual adults)	128
5.5 COMPARISONS BETWEEN STUDIES	133
5.5.1 Analysis by age: Comparison of visual data across Studies 1, 2 and 3	134
5.5.2 Analysis of adult language groups: Comparison of data across Studies 2 and 3	136
5.6 Discussion	141
5.6.1 DIFFERENCES BETWEEN MONOLINGUAL AND BILINGUAL CHILDREN	143
5.6.2 DIFFERENCES BETWEEN ADULT BILINGUALS IN STUDIES 2 AND 3	144
5.6.3 DIFFERENCE BETWEEN MONOLINGUALS AND BILINGUALS ACROSS STUDIES	144
5.6.4 IMPLICATIONS OF THE PRIORITISATION OF THE TASKS	145
5.6.5 The findings in the context of other studies	146
5.6.6 CONCLUSIONS	148
CHAPTER 6: EXPERIMENT 3	150
6.1 INTRODUCTION	150
6.2 CURRENT STUDY	151
4.3 MATERIALS AND METHOD	155
4.3.1 PARTICIPANTS	155
4.3.2 DESIGN	156
4.3.3 STIMULI	157

4.3.4 Procedure	158
4.3.5 DATA COLLECTION AND PREPROCESSING	158
4.3.6 Speech envelopes	159
4.3.7 ANALYSES	159
4.4 Results	160
4.4.1 BEHAVIOURAL COMPREHENSION SCORES	160
4.4.2 EEG DATA	162
4.4.3 The effects of attention on speech reconstruction accuracy	162
4.4.4 EFFECT OF CONDITION IN MONOLINGUAL AND BILINGUAL SCORES SEPARATELY	163
4.4.5 RECONSTRUCTION ACCURACY OF ATTENDED STREAMS IN MONOLINGUALS AND BILINGUALS	165
4.4.6 RECONSTRUCTION ACCURACY OF UNATTENDED STREAMS IN MONOLINGUALS AND BILINGUALS	166
4.4.7 COMPARISON OF SCORES BY CONDITION BETWEEN GROUPS	166
4.5 Discussion	168
4.5.1 The role of attention	170
4.5.2 Encoding Across conditions	171
	476
CHAPTER 7: SUMMARY AND CONCLUSIONS	176
7.1 NEURAL AND BEHAVIOURAL MODIFICATIONS TO SELECTIVE ATTENTION IN BILINGUALS – SUMMARY	Y OF
FINDINGS AND CONCLUSIONS	176
7.1.1 EXPERIMENT 1	177
7.1.2 EXPERIMENT 2	178
7.1.3 EXPERIMENT 3	182
7.2 CONCLUSIONS IN THE CONTEXT OF THE LITERATURE	185
7.3 IMPLICATIONS AND FUTURE DIRECTIONS	192
7.3.1 THEORETICAL IMPLICATIONS	192
7.3.2 FUTURE LINES OF ENQUIRY	194
APPENDIX A	<u> 198</u>
	•
APPENDIX B	214

REFERENCES	218

List of Tables

CHAPTER 4

Table 1:	Experimental Conditions	77
Table 2:	Comprehension scores and standard deviation by condition and group	83
Table 3:	T tests for comprehension scores between conditions in EEG Experiment 1	85
Table 4:	Reconstruction accuracy scores (r) by condition and group	87

Table 5: Experimental conditions 1	.11
Table 6: Study 1 (Children): Percentage correct comprehension scores by group and	
condition1	.18
Table 7: Study 1 (Children): Reaction times (SD) in ms and t tests between groups' log-	
transformed scores1	.20
Table 8: Study 1 (Children): Differences in conditions for monolinguals 1	.21
Table 9: Study 1 (Children): Differences in conditions for bilinguals 1	.22
Table 10: Study 2 (High-exposure bilinguals): Percentage correct comprehension scores 1	.23
Table 11: Study 2 (High-exposure bilinguals:) Differences in conditions for all participants	
	.25
Table 12: Study 2 (High-exposure bilinguals): Reaction times (SD) in ms and t tests betwee	n
groups' log-transformed scores1	.26
Table 13: Study 2 (Adults with high-exposure bilinguals): Differences in conditions for	
monolinguals1	.27
Table 14: Study 2 (High-exposure bilinguals): Differences in conditions for bilinguals 1	.28
Table 15: Study 3 (Low-exposure bilinguals): Percentage correct comprehension scores . 1	.29
Table 16: Study 3 (Low-exposure bilinguals): Reaction times (SD) in ms and t tests betwee	n
groups' log-transformed scores1	.31
Table 17: Study 3 (Adults with low-exposure bilinguals): Differences in conditions for	
monolinguals1	.33

Table 18: Comparison of bilinguals' scores in Studies 2 (High-exposure bilinguals) and 3	
(Low-exposure bilinguals) by condition	141
Table 19: Decrement t test results for Studies 1, 2 and 3 between monolinguals and	
bilinguals	147

Table 20:	Experimental conditionS	156
Table 21:	Comprehension scores and standard deviation by condition and group	161
Table 22:	Reconstruction accuracy scores for monolinguals and bilinguals by condition .	164

List of Figures

CHAPTER 3

Figure 1:	Dichotic listening task experimental procedure	54
Figure 2:	Stimulus reconstruction using the mTRF toolbox	61
Figure 3:	Reconstruction from a sample sentence	63
Figure 4:	T.O.V.A. Experimental Procedure	67

CHAPTER 4

Figure 5:	Dichotic listening task procedure for EEG Experiment 1	30
Figure 6:	Comprehension scores and standard deviation by condition and group	34
Figure 7:	Attended and unattended reconstruction accuracy scores (r) by condition and	
grou	ρξ	38
<i>grou</i> Figure 8:	p٤ Between-group differences in reconstruction accuracy scores per condition ٩	38 90

Figure 10:	Dual task procedure
Figure 11:	Study 1 (Children): Percentage correct comprehension scores by group and
condit	ion 118
Figure 12:	Study 1 (Children): Mean RT times by group and condition 120
Figure 13:	Study 2 (High-exposure bilinguals): Percentage correct comprehension scores124
Figure 14:	Study 2 (High-exposure bilinguals): Mean RT times by group and condition 126
Figure 15:	Study 3 (Low-exposure bilinguals): Percentage correct comprehension scores 130
Figure 16:	Study 3 (Low-exposure bilinguals): Mean RT times by group and condition 132
Figure 17:	Patterns of RTs across conditions for Studies 1, 2 and 3 135
Figure 18:	Patterns of monolingual and bilingual RTs across conditions in Studies 2 and 3

Figure 20:	Comprehension scores and standard deviation by condition and group 162
Figure 21:	Attended and unattended reconstruction accuracy scores (r) by condition and
group	
Figure 22:	Differences between the groups for attended and unattended conditions 167

Chapter 1: Introduction

1.1 The Aim

Bilinguals acquire and use their second language without apparent difficulties. This belies that fact that bilingualism is a major processing demand for the cognitive system; which has been shown to shape the brain, through neuroplastic adaptation in brain structure and modifications in neural processing. However, the development, effects and extent of such modifications are not fully clear.

In the research that follows, I investigate the effects of bilingualism on selective attention in children and adults of differing language exposure, with a series of complementary neural and behavioural experiments based on a naturalistic dichotic listening task. I aim to integrate three distinct but interrelated research questions: are neural and behavioural modifications discernible in childhood or only after protracted maturation or exposure to a second language? Do the increased processing demands of bilingualism enhance attentional capacity, or lead to an economisation of limited attentional resource, in order to optimise performance; and do any such effects evolve over time and experience? And finally, do these modifications affect any group's behaviour beyond the language domain? The answers to these questions will create a more nuanced and fuller understanding of the modifications conferred by bilingualism to selective attentional processes.

1.2 Selective attention

"My experience is what I agree to attend to." (James, 1890) Selective attention is the ability to direct focus to relevant stimuli while ignoring the irrelevant stimuli constantly bombarding the senses. It is a critical process without which the brain would be overloaded with sensory information and unable to prioritise any information. In nature, it has both ecological and evolutionary implications: animals respond adaptively to prioritised information that affects survival and reproduction, such as alertness to predators (Clark & Dukas, 2003). Research on selective attention in the human brain has generated much debate about its bases and mechanisms. I summarise key theoretical positions below.

1.2.1 Theories of selective attention processing

Despite the above definition's origin in the 19th century, research into selective attention took several decades to gain momentum. Research into auditory selective attention accelerated in the 1950s, when researchers focused on selective listening and the "cocktail party" paradigm. Researchers wanted to know how a listener confronted with many sounds (in the same vein as a partygoer in the midst of many conversations) manages to pick out the stream that is most relevant and ignore the unwanted noise. Furthermore, researchers wanted to investigate the difference in processing the target (attended) stream and irrelevant (unattended) stream. Many studies on this topic used the classic 'selective shadowing' task (Cherry, 1953; Broadbent, 1958; Moray, 1959). In these, two different speech streams were played to different ears over headphones, and listeners were asked to concentrate on one target stream and repeat ('shadow') it as quickly as possible. The aims of the studies were twofold: establish which features enabled the differentiation of the two streams and, determine what, if anything, the listeners retained of the message they were not shadowing. The results were clear: in order for efficient shadowing to take place, the streams needed to have clear physical differences, such as being played to different ears, or if not, originate from different locations or have different physical properties such as pitches/genders. Additionally, participants found it extremely difficult to retain any of the non-shadowed (unattended) stream, even when the speech contained anomalies such as a single word repeated in it many times, or even

a switch to a different language (Broadbent, 1958; Moray, 1959). This was true even when volume was kept consistent across the two streams, meaning the content of the second stream would have been noticed if attention had not been diverted away from it. The only exception was the reporting of very obvious physical properties, such as a complete change in the gender or pitch of the voice, the presentation of a loud tone; or silence after the stream ended. These findings led to Broadbent's filter theory (Broadbent, 1958), which not only compared the attentional limits of humans to the central processing limits of a computer, but also proposed a theory of *early selection*, a two stage processing mechanism, in which all physical properties of stimulus are processed in parallel initially, after which, due to constrained capacity, only limited abstract properties of the unattended stream (meaning, identity of spoken words) are extracted. Notably, such abstract properties progress through the filter as a result of physical characteristic, such as pitch or location. This explains why participants noticed changes in the physical properties of the non-shadowed speech, such as a change in the gender of the speaker, but not the content.

This led to a rival *late selection* account (Deutsch & Deutsch, 1963; Norman, 1968), which proposed that unattended streams were fully processed, but that participants' inability to respond to questions was a function of memory, rather than perceptual processing. This was substantiated using indirect methods of testing for the processing of unattended stimuli – such as in the controversial study (Corteen & Dunn, 1974) which paired certain words with electric shocks to stimulate a physiological response. Participants demonstrated some response even when these words were in the unattended stream. In an extension of this, participants produced a discernible response when the unattended words were synonyms of those that had been associated with electric shocks, suggesting some semantic processing (Von Wright, J. M., Anderson, K., & Stenman, U., 1975).

Treisman proposed an alternative *attenuated selection* theory (Treisman, 1960; Treisman, 1969), which suggested that some unattended stimuli are processed rather than filtered out completely, but the inputs are weaker than in target stimuli processing. In Treisman's research, the unattended stimuli that were retained had

specific features: either because they had especial significance to the listener (such as the words associated with electric shocks as described above, or the participant's own name, as discovered by Moray [1959]); or because they had been primed by contextual clues in the attended stream.

Such theories of early selection and late selection extended to the study of perceptual processing in vision, and these seemingly contradictory perspectives reconciled in a later theory proposed by Lavie and Tsal (Lavie & Tsal, 1994; Lavie, 1995), who found tasks with a high perceptual load resulted in early selection processing, and results from tasks with a low perceptual load favoured late selection processing. They concluded that the type of processing depended on the demands of the target task on attentional (in the case of vision, perceptual) capacity. If the target task is high in load, this consumes all capacity, leaving none spare for distractor processing (early selection); whereas if the target task is undemanding, spare capacity is released to process more of the distractor (late processing).

1.2.2 Attentional capacity

All the above theories of selective attention assume the important theoretical position that people have a finite attentional capacity, i.e., that human performance is supported by a limited pool of mental "effort" (attention) (Kahneman, 1973) which can only process a restricted amount of information at any given point (Broadbent, 1965; Clark & Dukas, 2003). This limited processing capacity can be allocated across stimuli (Kahneman, 1973; Navon & Gopher, 1979) to maintain optimal performance as long as total capacity is not exceeded (Moray, 1967). However, increasing competition for the limited capacity resource leads to an attentional bottleneck (Pashler, 1994) and deterioration in performance. According to early theories, the deterioration due to resource competition is not necessarily linear or "calamitous" but follows what is coined "the principle of graceful degradation" (Norman & Bobrow, 1975). Later studies applied engineering and computational processing theories to the allocation of these limited attentional resources, formulating models

such as the multiple resource theory (Wickens, 2008). One application of such models has been to incorporate the acquisition and competence of a second language as a demand on attentional resource (Wickens, 2007). The implications of this underpin the behavioural experiments later in this thesis.

1.2.3 Neural bases of selective attention in adults

Brain activity associated with selective attention is evident in mostly prefrontal areas (Salo et al., 2017). Neural responses to preparatory cues (which signal the stimulus to be attended before the target appears) engage a primarily fronto-parietal network (Bressler et al., 2008; Corbetta & Shulman, 2002). Tests of selective attention, such as the Flanker and Stroop (where the participant must suppress distracting information in order to focus on a target arrow or colour) have identified frontal lobes (Konishi, 2011) and the anterior cingulate cortex (ACC) for the management of conflict resolution (Walsh et al., 2011). Further investigations have suggested that the ACC is responsible for the *monitoring* of attentional conflict, and subsequently sends signals to frontal regions to *resolve* the conflict (Botvinick et al., 2004; Bush et al., 2000; MacDonald et al., 2000).

1.2.4 Developmental considerations

The maturation of selective attention is a key developmental process, hence even subtle modifications to this process have the potential to generate significant consequences. The importance of the development of selective attention reflects findings that it is not only linked to inhibitory control (Walsh et al., 2011) and working memory (Veer et al., 2017), but associated with the development of a variety of skills including speech (Astheimer & Sanders, 2012), metalinguistic skills (Astheimer et al., 2014) and arithmetic (Moll et al., 2015). In fact, selective attention has been proposed as one of the key foundational skills for overall academic success in children (Stevens & Bavelier, 2012; Hampton Wray et al., 2017).

Nonetheless, the neural systems associated with selective attention in adulthood, summarised in the section above, have a notably protracted period of structural development from infancy (Giedd et al., 1999; Gogtay et al., 2004; Sowell et al., 2001; Tsujimoto, 2008). The consequences of this protracted maturation on behavioural performance mean that children display attenuated ability to allocate attention relative to adults. This results in their allocating proportionately more attention to irrelevant stimuli at the expense of relevant stimuli in selective attention tasks (for reviews, see: Ridderinkhof & van der Stelt, 2000; Plude et al., 1994).

Given one of the aims of this study was to investigate the emerging effects of bilingualism on selective attention in development, it was important to establish an age range in which a proficient level of selective attention could reliably be assumed. Auditory selective attention is proposed to have developed by age 3-5 (Stevens & Bavelier, 2012) and auditory dichotic tasks have been carried out on children as young as 4 years old (Hampton Wray et al., 2017). Yet a minimum age of 6 has been recommended (Sanders et al., 2006), reflecting the inconsistent results and high variance in response speed and accuracy in the younger children (Takio et al., 2009). In addition, the established view is that selective attention only stabilises around the age of 7 (Gomes et al., 2007) and reaches maturity by the age 8 or 9 (P. R. Jones et al., 2015). Given these considerations, in the following experiments participants were recruited in the age bracket of 7-12, as this age range not only represents a developmental plateau for selective attention in childhood, but is also likely to generate relatively stable effects whilst ensuring that children can reliably perform a selective attention task.

1.3 Bilingualism

Bilingualism, or proficiency in a second language, has been valued historically for social, economic and political reasons: as a marker of social status, to enable international trade, espionage and diplomacy, and to disseminate cultural and

political ideas across nations, to name just a few. In the last century, it has become a focus for psychologists and neuroscientists, who have taken an interest in the processing of multiple languages, its implications for cognition and neurobiological effects on the brain. Evidence shows that the use of multiple languages leads to parallel activation and competition between them, requiring the speaker to prioritise one and inhibit the non-target language(s) (Green, 1998). These additional processing demands have been shown to shape the brain, through neuroplastic adaptation in brain structure (Mechelli et al., 2004; Klein et al., 2014; Burgaleta et al., 2016; Hayakawa & Marian, 2019; Hämäläinen et al., 2017); connectivity between regions associated with "language" and "control" (L. Li et al., 2015); and modifications in neural processing (Garbin et al., 2010; Luk et al., 2010; Kousaie & Phillips, 2012). These effects have been shown consistently across the lifespan, from childhood (Jasińska & Petitto, 2013; Archila-Suerte et al., 2018; Pliatsikas et al., 2020), into adulthood (Bialystok et al., 2005; Mechelli et al., 2004; Filippi et al., 2011) and old age (Luk et al., 2011; Abutalebi et al., 2015; Frutos-Lucas et al., 2020).

Some of these adaptations in brain structure have been associated with neuroprotective benefits in old age. Cognitive impairment and early stages of dementia are associated with grey matter loss, disruption of white matter and deterioration of the temporal poles and orbitofrontal cortex (Abutalebi & Green, 2016). Studies have shown bilingualism is associated with anterior lobe integrity (Abutalebi et al., 2014), greater myelination of white matter in the frontal lobes and corpus callosum (demyelination is associated with neurodegenerative conditions) and greater grey matter density in the dorsal anterior cingulate cortex, presupplementary motor areas, and temporal, inferior parietal lobe (where reduced grey matter is observable in the early stages of dementia) (see Abutalebi & Green, 2016, for a review).

There is also behavioural evidence that lifelong bilingualism confers neuroprotective effects, through slower cognitive ageing (Bialystok et al., 2012; Abutalebi, Guidi, et al., 2015) and a later onset of dementia (Gold, 2015; Bak & Robertson, 2017; Alladi et al., 2013; Estanga et al., 2017). Additionally, bilingualism has been associated

independently with substantially reduced poststroke cognitive impairment (Alladi et al., 2016; Wood, 2016). These findings have led to the conclusion that bilingualism increases "cognitive reserve", the ability to maintain cognitive functions despite neural damage or brain pathology.

This is in stark contrast to the traditional view that bilingualism led to cognitive overload and "mental confusion" in children (Saer, 1923). This was corroborated by early studies showing that bilingual children performed significantly worse than their monolingual peers in I.Q. tests, arithmetic, reading and verbal intelligence (Jones & Stewart, 1951; Lewis, 1959; Macnamara, 1966; Manuel, 1935; Darcy, 1953), and bilingualism was seen as further disadvantaging children with specific language impairments (W. R. Jones & Stewart, 1951). The results of many of these studies have since been contested, on the grounds that they did not control for environmental factors such as socioeconomic status, immigrant status, language ability; or had used assessments which were not appropriately comparative (Hakuta, 1986). In addition, since a landmark study by Peal and Lambert (Peal & Lambert, 1962), in which bilingual children performed significantly better than monolinguals on both verbal and nonverbal intelligence tests, there have been studies across a variety of age groups and cognitive functions that have proposed a significant advantage shown by bilinguals in performance. One claim is that a natural consequence of increased language processing is enhanced metalinguistic skills (Hakuta, 1986). A review of 102 studies of pre-school children (Barac et al., 2014), however, observed ambivalent results for metalinguistic awareness. They concluded that the most consistent findings of bilingual behavioural outperformance were in studies of non-verbal executive control and theory of mind. The generalisation of bilingual attentional control beyond the language domain is discussed in section 1.5 of this chapter.

However, proposals on the cognitive advantages of bilingualism across the age span have been challenged, with a number of studies and reviews not finding evidence for enhanced behavioural performance in bilinguals (von Bastian et al., 2016; Darcy,

1953) even when using advanced techniques or extensive sample sizes (Samuel et al., 2018; Nichols et al., 2020).

Notably, neural differences between monolinguals and bilinguals have been observed even when they display equivalent behavioural performance (Bialystok et al., 2005; Kousaie & Phillips, 2012; Olguin et al., 2019; Luk et al., 2010). This suggests that bilingualism modulates the neural mechanisms of the cognitive processing required for these tasks, through neuroplastic adaptation that cannot be captured by behavioural tests alone.

1.4 Degeneracy

The examples above, of equivalent behavioural performance contrasting with neural differences observed in groups with different language experience (see also Chee et al., 2004), leads to the concept of degeneracy, defined as structurally diverse components leading to the same output or performing the same function. In contrast to the everyday meaning with negative connotations of decay, degeneracy in the scientific sense is arguably a desirable characteristic, making systems robust and enabling natural evolution (Whitacre & Bender, 2010; Edelman & Gally, 2001). Consequently, it has positive implications in the fields of genetics (Whitacre & Bender, 2010), healthcare (Tian et al., 2011) and biology (Joshi et al., 2013). Given that degenerate systems are functionally plastic (Mason et al., 2015), it is a logical extension to view the brain as a degenerate organ, which must provide reliable outputs despite inherent variability due to individual experience. Additionally, degeneracy is a marker of high-level skill or expertise (Seifert et al., 2016), supporting the hypothesis that the neuroplastic change effected by bilingualism is an example of a degenerate system (Green et al., 2006; Mason et al., 2015; Navarro-Torres et al., 2021).

Furthermore, a key corollary of degeneracy in complex structures, is pluripotentiality, i.e., the generation of different outputs from the same degenerate

systems, depending on the circumstances. The brain, recognised as one of the most complex systems in the universe, has the capability to generate the same output from a variety of structurally different elements; these same elements can also generate different outputs independently (Tononi et al., 1999). This supports the hypothesis that not only does bilingualism result in a degenerate system, but that this system has pluripotentiality in its outputs depending on different contexts. This supports the possibility that systems tuned by language control have potential to extend to other domains.

1.5 Domain generality

One of Bialystok's key arguments (Bialystok, 1992) is that bilingual speakers develop an enhanced capacity for selective attentional control, driven by the experience of processing and producing multiple languages, and that furthermore, this control generalises beyond language control into non-verbal tasks. In support of this, Green and Abutalebi's Adaptive Control Hypothesis (Green & Abutalebi, 2013) details a framework for the mechanisms underlying bilingual control of attention, a process which they explicitly claim leads to "enhanced skills in cognitive control...deployed in nonverbal tasks". In support of Bialystok's hypothesis of bilingual enhanced attentional control, Green and Abutalebi (Abutalebi & Green, 2016) propose that, rather than just enhancing control of linguistic interference, the additional control of language employed by bilinguals is more likely to recruit evolutionary earlier systems, such as subcortical structures and the cerebellum, that underpin general attentional control and thus lead to changes in other non-verbal domains.

A number of behavioural studies have been conducted to investigate these claims, using nonverbal tasks including the Dimensional Change Card Sort (DCCS) task, flanker task, Attentional Network (ANT) task, ambiguous figures and Simon task. A bilingual advantage was found across a variety of studies and age groups when using these tasks (Bialystok & Martin, 2004; Carlson & Meltzoff, 2008; Calvo & Bialystok, 2014; Bialystok & Shapero, 2005; Poarch & van Hell, 2012; Costa et al., 2008), even

including pre-verbal infants, who had been raised in a bilingual environment, demonstrating enhanced selective attention through preferential eye tracking (Comishen et al., 2019; Kovács & Mehler, 2009).

However, this hypothesis of enhanced attentional control extending to non-verbal tasks is not universally recognised, and has been challenged by several vocal critics (Morton & Harper, 2007; Paap & Greenberg, 2013; Paap et al., 2014; Paap et al., 2015; Antón et al., 2014; Gathercole et al., 2014), who argue that any difference between groups can be attributed to confirmation and publication bias (Paap, 2016); or to variables other than bilingual experience (Lu & Proctor, 1995; Namazi & Thordardottir, 2010). Enhanced selective attention in bilingual infants has also been questioned (Kalashnikova et al., 2021). Given the "spotty" (Valian, 2015a) results acknowledged in reviews of the field of bilingual attentional control, the present research aims to explore further the patterns of differences between monolinguals and bilinguals beyond the language domain, by including a study that uses non-linguistic unattended stimuli in a version of the dichotic listening task.

1.6 Thesis outline

The main goal of the research described in this thesis to integrate three distinct but interrelated research projects:

- Establish whether neural and behavioural modifications triggered by bilingualism are discernible in childhood.
- 2) Determine if the increased processing demands of bilingualism enhance attentional capacity, or lead to an economisation of limited attentional resource, in order to optimise performance; and if such effects differ by maturation and language experience.
- Investigate the effect of these modifications on any group's behaviour beyond the language domain.

Chapter 2 reviews the current literature on selective attention, including theories of neural entrainment, and explains the themes introduced in this chapter in more detail, i.e., selective attention, developmental considerations, the influence of bilingualism and the intersection of these topics.

Chapter 3 introduces the methods used in the experiments presented in this thesis. It discusses the behavioural and imaging methods undertaken, including the dichotic listening task, which forms the basis of all my experiments, electroencephalography (EEG) used in the neural studies, and the dual task (auditory and visual) used in the behavioural experiments.

Chapter 4 describes the first EEG study¹, in which monolingual and bilingual children performed a dichotic listening task with linguistic attended and unattended stimuli. Their behavioural performance was assessed with comprehension questions on the target story and their neural responses captured with EEG and then compared to both the attended and unattended speech envelopes using the mTRF toolbox (Crosse et al., 2016). Both groups showed equivalent behavioural responses. In contrast, EEG data revealed differences between the groups, showing that the type of interference significantly modulated the neural encoding of attended speech in monolingual children but not in bilingual children, replicating the results previously observed in adults (Olguin et al., 2019). These results indicate that monolingual and bilingual children exhibit different patterns of neural entrainment to attended speech; and that the neural mechanisms of selective attention in bilinguals are reconfigured by the time children can reliably perform selective attention tasks.

Chapter 5 presents the follow up behavioural experiments which tested the limits of bilinguals' attentional processing capacity by using a dual task (dichotic listening + visual attention) to further increase processing load. The results over three experiments (on children, and adults with different levels of exposure to L2) consistently showed comparable overall performance of monolinguals and bilinguals

¹ A version of this chapter has been published in *Scientific Reports* (Phelps et al., 2022)

on both tasks, suggesting that bilinguals' adapted selective attention capacity can accommodate the high processing load of simultaneous demands in multiple domains. However there were also subtle differences in performance on the secondary (visual) task across the three groups of bilinguals, suggesting that maturation and exposure influence the functional adaptation of selective attention.

Chapter 6 describes the second EEG study, in which monolingual and bilingual children performed the dichotic listening task with non-linguistic unattended stimuli. Their behavioural and neural responses were captured in the same way as in the first EEG study. Again, both groups showed equivalent behavioural responses. EEG data revealed similar but attenuated patterns when compared to the first EEG study. Differences between the groups followed the same trend of lower indices of overall attention for bilingual children. These results support an account of constrained attentional capacity, even when performing nonlinguistic processing, which is economised to secure optimal behavioural performance.

Finally, Chapter 7 reviews the results of all the experiments and draws overall conclusions on the effects on selective attention of bilingualism and influence of maturation and language exposure. It also suggests theoretical implications and future directions.

Chapter 2: Literature Review

In the first chapter, I presented the three overarching research questions for this thesis, all concerning the effect of bilingualism on selective attention. In brief these were: firstly, establish whether and how bilingualism affects selective attention in childhood; secondly, determine whether bilingualism enhances attentional capacity or causes a redistribution of existing resource; and if such effects differ over time and exposure; thirdly, investigate the domain-generality of bilingual modifications to processing; and consider the implications of findings in each line of enquiry. Hence, this thesis explores the interactions between three overarching themes: selective attention; developmental influences; and bilingualism and its effect on processing. I explore the literature on each of these below, with an emphasis on the neural perspective of each theme. I also present some more detailed studies, which are particularly relevant to the experimental chapters that follow.

2.1 Neural basis for first and second language processing

A key assumption, introduced in Chapter 1, is that neuroplastic changes as a result of bilingualism are a consequence of parallel activation and competition between languages, requiring the bilingual speaker to prioritise one and inhibit the non-target language(s) (Green, 1998). This simultaneous activation and subsequent inhibition is considered to be the source of neuroplastic change as a consequence of bilingualism (Baum & Titone, 2014; Bialystok, 2017; Kroll, J. F., Dussias, P. E., Bogulski, C. A., & Kroff, J. R. V., 2012), and has been substantiated with numerous behavioural studies showing co-activation of lexical systems for bilinguals during linguistic tasks (Hatzidaki et al., 2011; Goldrick et al., 2016; Iniesta et al., 2021; Shook & Marian, 2019; Blumenfeld & Marian, 2013). Nevertheless, a persuasive counter argument, dubbed the "declarative/procedural model" (Ullman, 2001; Ullman, 2006), posited that L1 and L2 in the bilingual brain were processed in two distinct neural systems,

supported by different memory systems (declarative or procedural memory) to meet the demands of L1 and L2 on lexicon and grammar. This hypothesis was partly based on cases of bilingual aphasics who recovered just one of their languages, indicating a neural distinction between the languages (Albert & Obler, 1978).

However, subsequent neural evidence showed that L1 and L2 in the bilingual brain were processed by the same neural networks (Perani & Abutalebi, 2005). Researchers found that acquisition of L1 and L2 syntax and grammar recruited the same neural devices, in both ERP (Friederici et al., 2002) and fMRI (Sakai et al., 2004; Tettamanti et al., 2002; Musso et al., 2003) studies. Similar findings emerged from a study using a semantic task (Illes et al., 1999) for Spanish-English bilinguals in their L1 and L2. fMRI revealed similar frontal lobe locations of activations for both languages, and there were no differences in the patterns of activation when words in L1 and L2 were directly compared.

Furthermore, studies across different languages found remarkably consistent left hemisphere brain activation for specific linguistic processing tasks in L2 (Chee et al., 1999; Klein et al., 1999). The similar pattern was found in spite of differences between the languages such as orthography, phonology and syntax, indicating a universal neural basis for L2 processing.

Some neural differences have been discovered, however, which have been attributed to factors such as: age of acquisition, proficiency and exposure to each language, with each factor assumed to vary computational demands. For example, imaging studies on bilinguals who varied by age of acquisition or proficiency (Wartenburger et al., 2003; Briellmann et al., 2004; Sakai et al., 2004) showed that all bilinguals used overlapping areas when performing tasks in L1 and L2 but that additional activation for L2 was only evident in late bilinguals and/or those with low proficiency. The above findings point to a common neural basis for L1 and L2 for semantic, syntactic and grammatical processing, with additional activation according to proficiency and age of acquisition.

The evidence of consistent brain patterns, modulated by factors such as language exposure, have been interpreted as strong evidence of brain plasticity, dependent on specific bilingual experience (Perani & Abutalebi, 2005; Green et al., 2006). This is consistent with the evidence presented in Section 2.4.

In sum, there is neural evidence substantiating the hypothesis that a bilingual's two languages are co-activated. It therefore follows that this co-activation requires the bilingual speaker to inhibit one language and prioritise the other at any given time, causing additional processing costs. Experience of managing these costs is assumed to cause the bilingualism-induced modifications described in Section 2.4, many of which directly affect the domain of selective attention (Bialystok, 2017).

2.2 Selective attention

Selective attention, or the act of choosing to focus on stimulus while filtering out irrelevant inputs, has been a subject of interest to psychologists since at least the 19th century with early experimental work performed in the laboratories of Donders (Donders, F. C., 1868), Titchener (Titchener, 1910), von Helmholtz (as described by Stumpf, 1895) and Wundt (Wundt, W. M., 1912). In 1910, Professor Hicks presented the paper "The Nature and Development of Attention" to the British Psychological Society (cited in Edgell, 1947). Since then, neurobiological studies have established that the rate of encountering environmental stimulus far exceeds the rate of the brain's ability to process information (Itti et al., 2005); and thus attentional mechanisms are needed to allow the brain to focus only on the most essential filtered information at any given moment (Driver & Frackowiak, 2001).

As discussed in the Introduction, early work in the 1950s focused mainly on selective attention in auditory language processing, using dichotic 'shadowing' tasks (Cherry, 1953; Broadbent, 1958; Moray, 1959) which formed the basis of the early, late and attenuated selection theories (Broadbent, 1958; Deutsch & Deutsch, 1963; Norman, 1968; Von Wright, J. M., Anderson, K., & Stenman, U., 1975; Treisman, 1960).

Chapter 1 presented the above key theoretical positions, with an emphasis on auditory selective attention. Selective attention in the auditory domain was not only the focus of much of the research in the mid twentieth century, but is also highlighted given that the primary task in each of the experimental chapters is a dichotic listening task. One of the behavioural studies in this thesis, however, also features a visual task so below I also briefly summarise the literature in visual selective attention.

Much of visual selective attention theory (to which interest shifted after initial focus on selective processing in the auditory domain) is consistent with Broadbent's filter theory (Broadbent, 1965), although the distinction between 'physical' vs 'semantic' properties which are extracted by the information processing filter are more easily applicable to auditory linguistic stimuli (such as words) than natural visual stimuli. Nevertheless, early work on visual selective attention, including 'iconic memory' research (Sperling, 1960) is analogous to Broadbent's filter theory and assumptions of the finite bandwidth of a limited-capacity processor. In short, the 'iconic memory' theory was derived from the observation that participants were not able to report all the numbers they saw when presented briefly with an array of letters (e.g., four rows of three), indicating limited capacity to process them all. When asked to focus on a subset, however, of the same array (such as the top row only), participants were able to report all the relevant letters without apparent difficulty. In subsequent studies analogous to a dichotic listening task, in which participants were asked to concentrate on shapes outlines in one of two colours, a spontaneous memory recognition test revealed accurate memory of attended shapes (outlined in the target colour), but no recollection of unattended shapes (outlined in the irrelevant colour) (Rock & Gutman, 1981). This phenomenon has also been dubbed "Inattentional Blindness' (Mack, A., & Rock, I., 1998), and cited as an example of early selection in visual selective attention. Subsequent studies, however, revealed a relationship between an ignored object on one trial and attended object on the next, suggesting that some features of the ignored trial were indeed retained (Tipper, 1985). These findings were interpreted as an example of late selection, in which features of unattended objects are processed but then actively inhibited (see also

studies included in Monsell, S., & Driver, J., 2000). As summarised in Chapter 1, these contrasting interpretations were combined in a theory proposed by Lavie and Tsal (Lavie & Tsal, 1994; Lavie, 1995), who argued that early selection and late selection need not be mutually exclusive, but were instead a function of the task and its perceptual load. After finding that tasks with a high perceptual load resulted in early selection processing, and that tasks with a low perceptual load favoured late selection processing; they concluded that the type of processing depended on the demands of the target task on attentional (perceptual) capacity.

Another theory developed specifically for the visual modality is that of 'feature integration' (Treisman & Gelade, 1980), in which different aspects of visual stimuli are extracted separately but in parallel, and then integrated to produce multidimensional representations, with all features bound together. This theory was based on the results of visual search tasks, and multiple subsequent studies have tried to identify exactly how visual search is segmented and then reintegrated, with arguments for object-based (grouping by feature similarity) or space-based (grouping by location) models (Driver & Baylis, 1998). Further research into compromised attention as a result of brain damage has illuminated the debate, suggesting that neurological deficits in visual attention appear to be mainly spatial.

2.2.1 The neural basis for visual selective attention

Neural studies of visual selective attention have traditionally used either ERP (event related potentials) or functional imaging. ERP studies have shown a sensory response approximately 100ms after stimulus onset, typically with a larger ERP amplitude for attended stimuli than unattended stimuli, consistent with theories of increased neural activation to direct attention (Mangun et al., 1993). These results have been supplemented by functional imaging methods including PET and fMRI, which have demonstrated that neural activity (demonstrated by blood flow) in areas of the visual cortex are modulated by the demands of a visual selective attention task. One study showed that activation differed within the visual cortex between

colour-related and movement-related judgements for the same stimuli (Corbetta et al., 2007).

Now let us turn back to auditory selective attention and examine its neural basis.

2.2.2 Auditory selective attention and neural entrainment

As summarised in Chapter 1, research into selective attention initially concentrated on behaviour in a dichotic listening task pioneered by Cherry (Cherry, 1953), a version of which is used in the experiments that follow in this thesis. More recently, research has made significant progress in understanding the mechanisms of selective listening which underpin the behavioural outcomes identified by Cherry and others (Cherry, 1953; Broadbent, 1958; Moray, 1959). Studies tracking neural response to natural speech streams have revealed that, during listening, low-frequency neural oscillations entrain to the temporal envelope of speech (Aiken & Picton, 2008), which contains acoustic information necessary for linguistic decoding and perceptual encoding (Zoefel & VanRullen, 2015). Entrainment to the speech envelope has been shown to play an important role for speech intelligibility (Obleser & Kayser, 2019; Zoefel et al., 2018; Power et al., 2016; Ríos-López et al., 2020; Peelle et al., 2013; Millman et al., 2014), while poorer speech envelope encoding has been associated with developmental dyslexia (Power et al., 2016). In addition, analysis of the range of neural oscillations has revealed activity in different brain regions (Kubanek et al., 2013), and different frequency bands (Mai et al., 2016; Attaheri et al., 2020; Di Liberto et al., 2015; Giraud & Poeppel, 2012a), suggesting that different sources of information present in the speech signal, such as higher-level semantic and syntactic cues, engage distinct neural components (Keitel et al., 2018; Ding & Simon, 2014). Such processes can also vary according to stage of development: for example, deltatheta oscillations have been identified as enabling language acquisition in infants (Attaheri et al., 2020a), as well as facilitating perceptual parsing in adults (Doelling et al., 2014).

During a dichotic listening task, preferential tracking of the attended stream, or the 'selective entrainment hypothesis' (Schroeder & Lakatos, 2009; Zion Golumbic et al., 2013; Giraud & Poeppel, 2012b), demonstrates a stronger correlation between neural activity and attended speech envelopes (Aiken & Picton, 2008; Di Liberto et al., 2018; Mesgarani & Chang, 2012; Olguin et al., 2018), reflecting prioritised processing of the attended stream. This is true even in acoustically challenging conditions (Fiedler et al., 2018).

In addition, the type of interference has been shown to modulate the strength of entrainment to the attended stream. This was confirmed by a study (Olguin et al., 2018) in which adults performed a dichotic listening task while attending to a target narrative in English under varied conditions of interference. These included a stream that directly competed with the target in English, a narrative in an unknown language, a non-linguistic stream, and silence. Electroencephalography (EEG) responses for each condition were cross correlated with the attended and unattended speech envelopes. In addition to the expected selective entrainment effect of more robust encoding for the attended envelopes than for the distractors; researchers also found that the type of the interfering stream significantly modulated the strength of correlations, with the most intelligible distractor (English) causing the strongest encoding in both attended and unattended streams. In contrast, the non intelligible distractors caused weaker encoding, revealing a modulation of selective entrainment response according to the type of distractor. Behavioural performance (comprehension) was equivalent across all conditions.

Another dichotic listening task used magnetoencephalography (MEG) to record responses of adults listening to natural speech in two conditions of interference: intelligible or non intelligible strings of syllables (Har-shai Yahav & Zion Golumbic, 2021). Again, there was no difference in behavioural performance (comprehension of the target narrative) across conditions. There was, however, evidence of enhanced tracking of the target stream in conditions of the intelligible distractor (consistent with the results of Olguin et al., 2018). MEG analysis showed neural activation for the intelligible distractor stream in regions associated with linguistic

processing (left interior frontal and posterior parietal regions). The researchers (Harshai Yahav & Zion Golumbic, 2021) concluded that selective attention does not fully eliminate the processing of ignored speech.

Furthermore, experience of bilingualism has been shown to affect these patterns of neural entrainment in different conditions, as shown in a study by Olguin et al. (2019). Here, two groups of bilingual adults (those with a typologically similar L1 and L2 and those with typologically dissimilar languages) performed the same task as the monolingual adults presented in a previous study discussed above (Olguin et al., 2018). When the behaviour (comprehension) and patterns of response of all three groups (including monolinguals) were compared, the groups displayed equivalent comprehension of the target narratives. There were, however, different patterns of responses between bilingual and monolingual adults to the interference conditions of varied linguistic intelligibility and non-linguistic comparators, with monolinguals showing significant variation between the conditions of interference according to intelligibity of the distractor stream (as discussed above), and bilinguals showing a neutral pattern of neural response between all the conditions of interference, with no indications of variation. These differences in patterns of response to conditions of interference between bilinguals and monolinguals were held constant across both groups of bilinguals (with different typological experience of L2). Some additional fine-grained differences between the bilingual groups were revealed, namely distinctions as a result of time-windows analysis, specifically in the context of the two types of linguistic interference. These subtle differences, however, did not impact the main finding of significantly different patterns of response to different conditions of auditory interference between (all) bilinguals and monolinguals, coupled with equivalent performance. To date this has only been explored with adults and the fully matured brain with a protracted exposure to bilingualism. Therefore any such study involving children must take neurodevelopmental considerations into account.

2.3 Neuroanatomical and cognitive development overview

As would be expected, neuroanatomical development is associated with cognitive development (Passler et al., 1985; Crone & Ridderinkhof, 2011; Johnson, 2001). In particular, prefrontal areas have been linked to overall cognitive development (Casey et al., 2005; Majovski, 1989), especially those executive functions associated with selective attention such as inhibitory control (Konishi, 2011; Luna et al., 2001) and suppression of irrelevant information (Casey et al., 1997; Dempster, 1992). The parallel expansion of grey matter in prefrontal regions with age is proposed to correlate with an increase of control of interference (Bunge et al., 2002).

However, the development of prefrontal areas is protracted and non linear. Theories of development in early childhood broadly agree that the area comprising the frontal lobes increases sharply from birth to age 2, followed by a less pronounced growth spurt from about 4 - 7 years of age. From 7 onwards, there is a slow and much less dramatic increase in the size of the frontal lobes until adolescence (Dempster, 1992; Passler et al., 1985; Welsh & Pennington, 1988; Romine & Reynolds, 2005). Although total cerebral volume shows no significant increase after the age of 5, white matter volume increases with age and grey matter volumes increase during childhood and then decrease before adulthood (Durston et al., 2001), before stabilising at approximately age 22 (Dosenbach et al., 2010). In addition, MRI studies (Reiss et al., 1996; Sowell et al., 2001), reveal most prominent expansion in the prefrontal areas of the brain between the ages of 5-17, although growth is not uniformly correlated with chronological age (Sowell, Delis, et al., 2001). Neural studies using ERPs (Segalowitz & Davies, 2004) and EEG oscillations (Thatcher, 1991) agree that the prefrontal cortex is still developing into late adolescence.

As described in Chapter 1, development of selective attention, a core cognitive function, is closely associated with the maturation of the prefrontal cortex. In line with the "bumpy" process of structural maturation described above, development of cognitive functioning is also non linear, and developmental progression has been found to vary across tasks (Davidson et al., 2006). One proposal is that there are

different developmental trajectories for executive functions labelled "basic" as opposed to "complex" (Crone & Steinbeis, 2017) or "hot" and "cool" (Prencipe et al., 2011). Basic/hot executive functions are defined as stimulus-driven and, as well as selective attention, include cognitive flexibility, working memory, inhibition and error monitoring. Complex/cool functions are defined as intentional processes, such planning or emotional regulation. Basic/hot stimulus-driven processing is proposed to have a shorter development, whereas complex or cool tasks show a different range of patterns in prefrontal areas into adolescence. Additionally, children's cognitive functioning is not just a delayed or reduced version of adults' performance. fMRI has revealed that different regions are activated relative to adults when children aged 7-12 perform cognitive tasks (Rueda et al., 2004; Casey et al., 1997). Several reasons for such variance in processing relative to adults have been proposed (Ridderinkhof & van der Stelt, 2000). One explanation is that participants of different ages may interpret, understand or perform task instructions in distinct ways and may have different motivation or endurance for following tasks for extended duration. Another is that as they develop, children may use different strategies to perform the tasks that do not necessarily correspond to age-related changes in processing. Furthermore, cognitive architecture does not have a linear development (as described above); therefore corresponding neural processes may not gradually increase in efficiency, but go through phases of qualitative reorganisation. This means that any measurement through tasks might reflect a qualitative phase change rather than gradual development in attentional processing. Finally, neural results may be influenced by changes in brain maturation that are unrelated to attentional processes, so any patterns shown may simply be a reflection of age-related change that is due to non-attentional factors.

Furthermore, tests of selective attention often tap a single domain – visual or auditory – which, as described below, show overlap in terms of phases of development, but also reveal distinct neural correlates.
2.3.1 Development of selective attention - visual

Studies of selective attention in children have first focused on the visual domain through nonverbal tasks such as (modified) Stroop, Simon and Flanker. Although tests have been performed successfully on children as young as 24 months (Poulin-Dubois et al., 2011), an early review of monolinguals (Comalli et al., 1962) recommended a minimum age of 7 years old, on the basis of reading ability. Typically, studies find that both reaction time and accuracy improve with age between 6 and 11 (Rueda et al., 2004). A comprehensive study of age differences in Stroop performance (Comalli et al., 1962) tested over 200 individuals ranging in age between 7 and 80. Results showed that visual selective attention improved with age from the age of 7 to adulthood, remained stable over the young adults into middle age, and declined for the oldest group tested (65-80, see also Wise et al., 1975).

2.3.2 Development of selective attention – auditory

Reviews have proposed that selective listening, in the same way as visual selective attention, improves through childhood and adolescence to early adulthood, and then declines in late adulthood (Dempster, 1992; Mueller et al., 2008). Comparisons of children between age categories or relative to adults have shown significant differences: for example 10 year olds performed four times better than 7 year olds when attending target words (Geffen & Sexton, 1978). In versions of Cherry's cocktail party task (Cherry, 1953), accuracy of repeating a target word while ignoring a distracting word improved linearly with age across several age groups of children (preschoolers: Hiscock & Kinsbourne, 1977; 3-12 year olds: Hiscock, M. & Kinsbourne, M., 1980; 5-6, 7-8 and 9-10 year olds, Anooshian & McCulloch, 1979; 7 and 9 year olds; Geffen & Wale, 1979; 8, 11 and 14 year olds, Doyle, 1973). Similarly, a study comparing children's (aged 4-11) and adults' performance in a tone-in-noise task, showed comparable performance in quiet conditions, but a significantly poorer performance by children in noisy conditions. Again, performance in noisy conditions was correlated with age and reached adult levels by the age of 9-11 (P. R. Jones et al., 2015).

In terms of neural indicators, the maturational development described above is reflected in neural studies that report step changes in the development of the auditory selective attention system. A study investigating the development of auditory attention using auditory evoked potentials (AEPs) (C. W. Ponton et al., 2000) on 118 participants aged 5-20 found that not only does maturation extend into adolescence, but that amplitude changes are more abrupt and step-like, in contrast to changes of latency that are more gradual. A related study (Ponton et al., 2002) found that different AEP components reach maturation at different ages: some by the age of six (specifically the middle latency response [MLR] components Pa and Pb, P₂, and the T-complex); some that mature consistently 50% per year (N₂); and a third group that mature more slowly (at a rate of 11-17% per year) comprising different components again (the AEP peaks P₁, N_{1b}, and TP₂₀₀). The above findings therefore indicate not only age-related maturation, but also variability across components and a non-linear trajectory.

When tested in the context of a dichotic listening task, different age groups displayed distinct amplitudes and latencies of ERPs (Karns, C. M., Isbell, E., Giuliano, R. J., & Neville, H. J., 2015). In this study, the only comprehension differences were less accuracy in the youngest age group (3-5) but across all others (10, 13, 16 and 18-25) comprehension was comparable while (non-uniform) maturational differences were observed for linguistic and nonlinguistic probes. Crucially, such changes in neural processing have been discerned in the absence of behavioural differences in performance (comprehension), indicating neural modifications that cannot be detected by behavioural tests alone. Such age-related neural differences despite equivalent performance have been found in both non-naturalistic (Mueller et al., 2008) and naturalistic listening tasks (Karns, C. M., Isbell, E., Giuliano, R. J., & Neville, H. J., 2015).

In addition to age, environmental factors have also been shown to exert an influence on the development of auditory attention, as demonstrated in neural responses to auditory tasks. One such environmental variable is socioeconomic status (SES).

Similar to studies investigating selective auditory attention across age groups (Karns, C. M., Isbell, E., Giuliano, R. J., & Neville, H. J., 2015), those examining the impact of SES on selective attention in children have also found different neural (ERP) patterns despite equivalent performance (comprehension), while controlling for age.

In one study investigating the development of auditory selective attention (Hampton Wray et al., 2017), groups of higher and lower SES children performed a listening task while their ERPs were measured and compared at age 4. As a followup, the children in the lower SES cohort were retested a year later. Findings showed that the age of 4, children of higher SES showed a significant effect of attention (higher ERPs for attended than unattended probes), and the children of lower SES showed no such effect (similar ERPs for attended and unattended). However, comprehension scores between the two groups at age 4 were not statistically different. When the lower SES children were retested a year later, at age 5 the attention effect in the lower SES group was comparable to the higher SES group at age 4 to the attended stream. The researchers concluded that not only did this represent a maturational delay of one year between the ages of 4 and 5 for the lower SES group (according to neural, not behavioural indices), but also a "divergent developmental pattern in neural mechanisms" according to SES background (Hampton Wray et al., 2017). This study was supported by others that had found neural differences in the higher and lower SES groups of 3-8 year olds despite comparable comprehension (Stevens et al., 2009), and differences in theta power between higher and lower SES pre-adolescent children (age 11-14) in attended and unattended streams, despite equivalent reaction times and accuracy (D'Angiulli et al., 2008). Finally, an 8 week intervention to improve selective attention on preschool children of lower SES (Neville et al., 2013) successfully improved attention to the attended stream.

The first SES study detailed above (Hampton Wray et al., 2017) and similar studies showing neural differences in groups of children of different socioeconomic status are especially relevant to the themes of this thesis for two reasons. First, they explicitly attribute differences in neural processing to environmental variables and

thus support theories of neuroplasticity. This is analogous to bilingualism, another such environmental variable, which affects neural structure and processing, as summarised in Chapter 1 (Mechelli et al., 2004; Garbin et al., 2010; Burgaleta et al., 2016; Hayakawa & Marian, 2019; García-Pentón et al., 2014). The second is the finding of neural differences between the groups despite equivalent behaviour. One of the conclusions from the researchers into socioeconomic status and selective attention is that children of different SES backgrounds recruit different neural processes to achieve the same result (D'Angiulli et al., 2008). This again is similar to studies of bilingualism (in adult samples) which have found comparable behavioural performance but distinct neural patterns across a variety of methodologies: MEG (Bialystok, Craik, et al., 2005), fMRI (Luk et al., 2010; Chee et al., 2004; Román et al., 2015), EEG (Olguin et al., 2019) and ERP (Kousaie & Phillips, 2012; Kousaie & Phillips, 2017). These parallel results suggest that not only is the brain plastic, but a degenerate organ which, in spite of structurally diverse components, can generate equivalent outputs (see Chapter 1, also Navarro-Torres et al., 2021; Edelman & Gally, 2001; Green et al., 2006; Mason et al., 2015).

Not only is SES similar to bilingualism in being an environmental variable that influences attention; it is also a significant covariate (Naeem et al., 2018), particularly in childhood. As mentioned in the Introduction, a number of studies in the early to mid twentieth century claimed that bilingual children were at a cognitive disadvantage after they performed significantly worse than their monolingual peers in I.Q. tests (W. R. Jones & Stewart, 1951; Lewis, 1959; Wang, 1926); arithmetic and reading (Macnamara, 1966; Manuel, 1935) and verbal intelligence (Darcy, 1953), but a fundamental criticism of these studies was that they did not control for SES. Since those early studies, SES has been recognised and well-documented as an influence on cognitive development (Hoff, 2003; Walker et al., 1994; Andrews Espy et al., 2001; Mezzacappa, 2004), and more recent studies on the impact of bilingualism on cognitive performance in childhood, which have controlled explicitly for SES, have claimed a bilingual advantage as a separate effect from SES in tests of children from different backgrounds (Calvo & Bialystok, 2014; Carlson & Meltzoff, 2008). A positive effect of bilingualism has also been found in a sample comparing

monolinguals and bilinguals in a lower SES group (Engel de Abreu et al., 2012), suggesting that bilingualism may mitigate any effects of lower SES on children's cognitive processing on tasks requiring selective attention (Blom et al., 2014).

2.4 Neuroplasticity and bilingualism

As discussed above, neural differences can exist despite behavioural equivalence, due to neuroplastic adaptations to environmental factors such as SES (as above) and bilingualism. The case for neuroplastic change as a consequence of bilingualism is well-documented (see Li et al., 2014 for review), with claims that bilingualism confers a "neural signature" (Kovelman et al., 2007; Jasińska & Petitto, 2014) even at the initial stages of learning a second language (Yang et al., 2015), and showing different activation in multiple regions from monolingual processing, especially in tasks of selective attention (Luk et al., 2010). There are findings that bilingualism influences the processing of the most basic linguistic stimuli. For example, bilingual adults have been shown to have stronger subcortical representation of fundamental frequency (F0) and more consistent neural responses to attended syllables (/ba) compared to monolinguals (Krizman et al., 2012; Krizman et al., 2014), even when they have been exposed to different combinations of languages (Skoe et al., 2017), indicating that bilingual experience has an impact on underlying neural processing of linguistic input regardless of the languages of exposure.

However, it has also been found that neural adaptation as a result of bilingualism can itself vary according to factors including, but not limited to: a) age of acquisition, affecting volume of cortical areas (Wei et al., 2015), distribution of white matter (Nichols & Joanisse, 2016) cortical thickness (Klein et al., 2014) or the processing of syllables (Zinszer et al., 2015); b) proficiency, affecting grey matter in the temporal pole (Abutalebi et al., 2013) and prefrontal activity (Videsott et al., 2010); c) language similarity, affecting activation when reading in the second language (Kim et al., 2016); or d) age, increasing activation (McNealy et al., 2011; Jasińska & Petitto, 2014).

Changes in the developing brain as a result of bilingualism have also been of great interest to researchers. Infants as young as 4-6 months have been found to have different phonetic processing as a result of bilingualism, using fNIRS (Petitto et al., 2012). Monolingual and bilingual children aged 8-11 were found to have structural differences in white matter tracts when scanned using magnetic resonance diffusion tensor imaging (MR-DTI) (Mohades et al., 2012). Neural activation was increased in left-hemisphere classic language areas for 7-10 year old bilinguals relative to monolinguals in an fNIRS study (Jasińska & Petitto, 2013). A neuroanatomical study of two groups of bilingual children aged 6-13 who differed in frequency of use of their L2 (Archila-Suerte et al., 2018) found significantly different neuroanatomical profiles according to language experience: children with "balanced bilingualism" (equal use of both L1 and L2) had thinner cortices of the left superior temporal gyrus (STG), left inferior frontal gyrus (IFG), left middle frontal gyrus (MFG) and a larger bilateral putamen, whereas "unbalanced bilinguals" (those who mainly used L1, and less frequent use of L2) showed thicker cortices of the same regions and a smaller putamen.

Together, the results from the above studies indicate that a "neural signature" (Kovelman et al., 2007) is not a single pattern or presents exclusively in languagespecific regions. Rather, the effects of bilingualism have multiple ways of demonstrating neural change (grey matter, cortical thickness etc.) in different brain regions, and, furthermore, such manifestations are also subject to variation due to language experience (Kuhl et al., 2016).

Behavioural studies of children indicate that these neural modifications, as a result of bilingualism, at the very least correlate with behavioural measures of linguistic processing. These are apparent from earliest development. Newborn bilingual babies are already able to differentiate between the two languages that were spoken while they were in the womb (Byers-Heinlein et al., 2010). Bilingual infants are reported to have enhanced ability to discriminate between languages when watching silent videos in different languages (Sebastián-Gallés et al., 2012). Bilingual

children aged 6-9 showed enhanced performance when learning the phonological patterns in a new language (Kuo & Anderson, 2012). Groups of bilingual children aged 7 and 9 showed enhanced phonological awareness and meta-linguistic skills despite lower vocabulary scores than their monolingual peers (Eviatar & Ibrahim, 2000). [Vocabulary scores for bilingual children have been shown to be lower in one language but equivalent across languages to a monolingual's single language (Pearson et al., 1993; Hoff et al., 2012)].

These differences in processing linguistic input can affect performance on tests of auditory selective attention in children. In one study (Filippi et al., 2015), researchers tested whole sentence comprehension by bilinguals and monolinguals under conditions of auditory interference. They compared bilingual and monolingual children's comprehension of simple (canonical) and complex (non-canonical) sentences with and without whole-sentence interference. The L2 of the bilingual children varied across nine languages, but all spoke English fluently. The target sentence was always spoken in English and the interference was presented in either English or Greek, and different combinations of target and distractor sentences in English and Greek were counterbalanced. Bilinguals performed better than monolinguals at comprehension of complex sentences when the interference was in Greek, the unknown language. This supported similar finding in adults (Filippi et al., 2012), in which the bilingual (Italian-English) group of adults significantly outperformed their Italian monolingual peers, especially in the comprehension of non-canonical sentences, in the condition of auditory interference (in either language, but particularly in their native Italian). There are several parallels between the above studies (Filippi et al., 2015; Filippi et al., 2012), and the studies in the experimental chapters that follow. Not only do these studies use an auditory task with different levels of interference, the later iteration (Filippi et al., 2015) directly compares children and adults performing a similar task. Equivalent patterns in behavioural results between bilingual children and adults suggested that children had already developed similar modifications to performance, to those shown by adults, as a result of bilingualism, within the parameters of this particular task. This

implies that bilingualism had caused similar neural adaptation to auditory selective processing across both age groups, a finding very pertinent to the first and second questions addressed in this thesis, albeit with a differently designed dichotic listening task.

A different line of enquiry exploring bilingual auditory linguistic processing has been investigations of processing speech in noise. Studies have found that bilinguals routinely show poorer recognition of words in noise relative to monolinguals (Bradlow & Alexander, 2007; Mayo et al., 1997; Rogers et al., 2006). Across these studies, bilinguals required either a greater signal-to-noise ratio than monolinguals (Shi, 2010; Mayo et al., 1997) or increased clarity or predictability of the speech signal (Bradlow & Alexander, 2007) than monolinguals in order to perform the task. Age of acquisition and proficiency have been identified as contributing factors (late bilinguals or those with lower proficiency struggle more to identify the speech sounds) but not enough to account fully for a bilingual speech-in-noise disadvantage, as near-native proficiency bilinguals also demonstrated poorer performance in a speech in noise task than monolinguals; despite equivalent performance in a speech in quiet condition (Rogers et al., 2006). The proposed explanation for these findings is that the greater demands of managing two or more languages create greater processing demands for bilingual speakers, particularly when accessing lexical information within a sentence, similar to the hypothesis about constrained attentional capacity in bilinguals introduced in Chapter 1.

One study (Krizman et al., 2017) further explored the proposed bilingual disadvantage at recognizing speech in noise, by presenting five different conditions to bilingual and monolingual participants, with varying amounts of speech and non-speech in noise. They explicitly tested whether this specific processing disadvantage was language-dependent by including conditions of tones (non-linguistic input) as well as speech (linguistic) in noise. They found that performance differences between bilinguals and monolingual adolescents (aged 14-15) varied by condition and its linguistic processing load: bilingual adolescents (early, high-proficiency)

performed worse than monolinguals when perceiving sentences. There was comparable performance between the two groups when perceiving individual words. Bilinguals performed better than monolinguals, however, at perceiving tones in noise. The researchers concluded that this pattern of differences indicates that any bilingual disadvantage when perceiving stimulus in noise is due to bottlenecks in language-dependent processing, while non-linguistic input is free of such constraints. These findings have direct relevance to several of the questions addressed in this thesis. In reference to determining the effects of bilingualism on attentional capacity, in the above study a bilingual "bottleneck" for linguistic processing was identified, consistent with the concept introduced in Chapter 1 of an attentional bottleneck (Pashler, 1994) caused by increasing competition for the limited capacity resource, which leads to a deterioration in performance. In the context of my investigations of bilingual processing of linguistic and non-linguistic interference in a dichotic listening task, the study described above showed different patterns of bilingual performance when processing linguistic and non-linguistic stimuli in noise, indicating not only differences by group (relative to monolingual performance) but by condition (sentence, word, tone). This will be compared and contrasted with my own results in the Discussion in Chapter 7.

Additional investigations of monolingual and bilingual performance in both linguistic and non-linguistic tasks have revealed contrasting results to the findings described above (Krizman et al., 2017). A study which compared the performance of younger adult bilinguals and monolinguals during both a linguistic processing task and a nonverbal task (Blumenfeld & Marian, 2011), found no difference in performance in either task between the groups. The linguistic task consisted of listening to words and then identifying them by selecting their visual representations from a group of four pictures. This group of pictures also contained a linguistic competitor, e.g., if the target (spoken) word was "hamper", a picture of a hamper was accompanied by a picture of a hammer, as well as pictures of two other random words. Accuracy in the linguistic task was measured by pressing the correct button on a key pad to identify the target word. Simultaneous eye tracking measured the activation of

competitor words and control words during word recognition, by recording the amount of time the participant looked at each (word) picture from the four options. In addition, the word recognition task was followed by one of three types of priming probe, which presented a grey asterisk in the same place as one of three variations: the target, competitor or one of the control (irrelevant) stimuli from the previous trial. Eye tracking measured how long attention rested on the grey asterisk quadrant of each priming trial. Attention on a control probe, which was designed to elicit minimal attention from the previous round, was considered the baseline. Response latencies (eye gaze time) for the grey asterisk when the competitor picture had been on the previous trial were considered to reflect inhibition and were expected to be longer than control probe trials. Finally, response latencies for target probes (where the grey asterisk was in the same quadrant as the target picture in the previous trial) were expected to elicit the shortest gaze relative to both the control and competitor probes, representing location facilitation as the previous trial had garnered maximum attention. The non-verbal task was a Stroop task, with speed and accuracy measured by responses on a key pad to congruent and incongruent arrows on a screen. Despite overall equivalent performance (accuracy) in both tasks between the groups, results showed different patterns for both inhibition and correlations between tasks (bilingual inhibition mechanisms between the linguistic and non-verbal tasks were correlated; monolinguals recruited different resources for each task). A follow up study tested older and younger adult bilinguals and monolinguals during the same linguistic processing and non-verbal tasks (Blumenfeld et al., 2016). Once again, there was no difference between monolingual and bilingual performance in either age group. There was, however, an age-related difference, which varied in magnitude between the groups. Bilinguals suffered less performance decline as they aged. Once again, results for both age groups revealed different patterns suggesting that, in both tasks, bilinguals recruited similar inhibition mechanisms; whereas monolinguals showed different mechanisms for performing the linguistic and non-linguistic tasks. The finding from both studies are directly relevant to my later comparisons of bilinguals in different age groups performing the same behavioural task (Blumenfeld et al., 2016); and the discussion

about monolingual and bilingual performance in a nonverbal task (Blumenfeld & Marian, 2011; Blumenfeld et al., 2016).

The above study found equivalent behavioural performance between monolingual and bilingual participants, in two age groups, for both the linguistic and non-verbal tasks. These findings contrast with other studies that have found differences between monolinguals and bilinguals in age groups of younger and older adults. One such study tested both age and language groups in a non-verbal switching task (Gold et al., 2013), in which bilingual participants in both age groups outperformed their monolingual peers. Furthermore, concurrent fMRI data revealed that bilingual participants in both age groups showed decreased activation in the left lateral frontal cortex and cingulate cortex. The researchers concluded that lifelong bilingualism mitigates age-related declines in neural processing for cognitive control tasks (see also Costa & Sebastián-Gallés, 2014, for a review of how bilingualism affects processing across the lifespan).

2.5 Summary

In sum, the literature reviewed above synthesises findings on the topics of bilingualism, selective attention (visual and auditory), cognitive development, neuroplasticity and interactions of the above, supported by both behavioural and neural evidence. First I examined the neural basis for the claim that a bilingual's two languages are co-activated, leading to competition between them, which must be resolved. Experience of managing this continual competition has been shown to lead to neuroplasticity, specifically in the areas and processing associated with selective attention.

On the topic of selective attention, I summarised the key theoretical positions for visual selective attention and the neural basis of both visual and auditory selective attention. Auditory selective attention relies on tracking of the speech stream, which, according to the 'selective entrainment hypothesis' (Costa-Faidella et al.,

2017), tracks the attended stream preferentially in conditions of interference or dichotic listening. Furthermore, response patterns in selective entrainment vary according to both the type of interference and due to factors such as bilingualism (Olguin et al., 2019), although to date this has not been tested on children.

On the topic of children and the developing brain, I summarised the development of both cognition and specifically selective attention in both the visual and auditory domains, with both behavioural and neural evidence. I presented evidence that the development of (auditory) selective attention is affected by factors such as age, and environmental factors including bilingualism, socioeconomic status, and combinations of the above.

This led to the topic of bilingualism and neuroplasticity, both generally and in the developing brain. This has been associated with modifications in linguistic performance in childhood, encompassing a linguistic advantage in one comprehension task (Filippi et al., 2015); a linguistic disadvantage, but non-verbal advantage in another (Krizman et al., 2017), and no behavioural difference in a third, but evidence of differences in processing through neurophysiological evidence (in this case, eye gaze) (Blumenfeld et al., 2016).

In this way, I have set the scene for the three questions that will be addressed in this thesis that draw on elements of bilingualism, selective attention, development, neuroplasticity, maturation, language experience and the impact of bilingualism on linguistic vs. non-verbal processing. After presenting my experimental chapters, I will review the findings in the context of some of the studies described above to evaluate where they fit in the landscape of the current literature.

Chapter 3: Methods

3.1 Introduction

The neural and behavioural studies used different but complementary methods to investigate the differences in monolingual and bilingual attentional processing. All the experiments used a dichotic listening task with variations in the attended and unattended conditions, and auditory stimuli, which differed according to the experiment. The dichotic listening task is detailed in Section 1.

In the investigations of neural encoding of attended and unattended speech, the dichotic listening task was combined with the neural data using EEG (electroencephalography), which were analysed using the mTRF toolbox. Section 2 describes the justification for using these methods; how the data were collected and cleaned using a pre-processing pipeline in EEG Toolbox; the calculation of the speech envelope; and the features and computation of the mTRF toolbox.

In the behavioural studies of multimodal attentional processing, a dual task paradigm was devised. Here the dichotic listening task was combined with an adapted version of visual T.O.V.A. (Task of Variables of Attention) to be performed simultaneously. Section 3 describes the dual task paradigm and T.O.V.A. in more detail.

3.2. Dichotic listening task

3.2.1 Overview

The task common to all the studies was a version of the dichotic listening task pioneered by Cherry (1953). His version was named the "cocktail party" task, due to

its novel use of natural speech streams instead of clicks or tones, and similarity to the scenario of a partygoer who must ignore distracting noise in order to focus on a single conversation. In the task, participants are presented with two streams of auditory stimulus simultaneously, in different ears via headphones, and asked to concentrate on one, the target, and ignore the other, unattended stream. In the design used for the experiments in this thesis, target streams were created to be naturalistic speech and unattended streams were varied across conditions.

3.2.2 Dichotic listening and selective attention

The dichotic listening task as described above is a well-established tool for measuring selective attention. Behavioural and neural studies have used versions of it to investigate the locus of processing (Kimura, 1967; Har-shai Yahav & Zion Golumbic, 2021) and compare the strength of cortical response to the target stream relative to the unattended information (Kerlin et al., 2010). It is compatible with a range of neuroscientific methods such as MEG (magnetoencephalography, e.g., Ding & Simon, 2012), ECoG (electrocorticography, e.g., Zion Golumbic et al., 2013) fMRI (functional magnetic resonance imaging, e.g., Wikman et al., 2021) and EEG (electroencephalography, used in this study and others, including Olguin et al., 2019; Horton et al., 2014; Mesgarani & Chang, 2012; Fiedler et al., 2018; Broderick et al., 2018). The stimulus can be tailored to the age range and ability of the participants, and interference manipulated to test responses to a variety of linguistic and non-linguistic streams (Olguin et al., 2018; Har-shai Yahav & Zion Golumbic, 2021). It can use naturalistic stimulus, which produces more accurate and generalisable results (Hamilton & Huth, 2020) than single words, clicks or tones; and is a more engaging experience for the target participant age range in these studies. Therefore this version of the dichotic listening task was extremely suitable for both the behavioural and neural studies to assess auditory selective attention across age groups and language backgrounds, and test responses to various linguistic and nonlinguistic unattended stimuli.

3.2.3 Task design

The dichotic listening task was designed to investigate the effects on selective attention of listening to continuous speech, whilst ignoring different levels of linguistic and non-linguistic interference; and establish whether these patterns differed between monolingual and bilingual children and adults. Each experiment followed a similar procedure for the dichotic listening element; with different stimuli pairs according to the experiment, which are detailed more fully in each experimental chapter. In summary, both groups were presented with two competing continuous streams, which were played simultaneously in different channels of headphones. Participants were instructed to focus on the target narrative (attended stream) in either the left or right ear, and ignore the interference (unattended stream) in the other ear. Attended/unattended streams were counterbalanced between the left and right ears to avoid a dichotic right-ear advantage, especially in children (Berlin et al., 1973). The target stream was always in English, as this was the language common to all participants, and to provide equivalence between the target streams in all conditions. The narratives consisted of sentences concatenated to form continuous speech, which lasted approximately 3.3 minutes per presentation. These uninterrupted narratives in continuous speech enabled an assessment of comprehension in a setting that was both naturalistic and ecologically valid.

Conditions. Each experiment consisted of four auditory conditions: in the first, all participants attended to a target story in English without any interference (Single Talker). In the subsequent three conditions, the target story in English was paired with different levels of interference. The type of interference and length of the conditions differed according to the experiment, and is detailed in the experimental chapters. The EEG experiments consisted of two 3.3 minute blocks per condition; the behavioural experiments presented a single 3.3 minute block per condition. In addition, the EEG experiments used different unattended stimuli (linguistic

distractors in EEG Experiment 1, non-linguistic noise in EEG Experiment 2); the behavioural experiments shared the same unattended stimuli as EEG Experiment 1. These are detailed in the relevant chapters.

Stimuli. The target stories for the attended ear in the EEG experiments were four stories in English specifically aimed at children. Two of these stories were used as the attended stimulus in the behavioural experiments, which were shorter in duration than the EEG experiments, and used half of each target story per condition. The use of children's stories, with simple syntax and vocabulary, ensured that all participants would be able to understand them and that there would be no variance in target content comprehensibility across conditions.

The target streams for the unattended ear varied by experiment, except for the first condition, which was always silence in the unattended ear (Single Talker condition). In the first EEG experiment and the behavioural experiments, the unattended streams in the remaining three conditions were stories in English (English-English condition), an unknown language (English-Latin condition) and an acoustic stream derived from one of the target stories (English-Musical Rain condition). In the second EEG experiment, which tested response to non-linguistic interference, the three unattended streams after Single Talker were a simple and repetitive melody (English-Music condition), a recording of children in a classroom with no distinguishable words (English-Babble condition) and a stream of vehicle noise (English-Traffic condition).

The stories in English were sourced from online resource storynory.com. Care was taken to choose stories that were not universally familiar (e.g., Pinocchio, Cinderella), where participants may have been able to answer comprehension questions about the target story without paying full attention. After selecting more obscure or original stories, some details were further changed, and the stories were transcribed into narratives 120 sentences long, with each sentence lasting approximately 3 seconds in length. The four attended English stories were then recorded sentence by sentence by a native English female speaker. The unattended

linguistic stimuli were recorded by different female speakers. Gender of the speaker was kept the same for all stories (same female voice for all target stories, different female voices for interference), to reduce segregation strategies based on talker's gender (Brungart & Simpson, 2007). Transcripts of the attended English stories, unattended English story and unattended Latin story can be found in Appendix A.

Each target story was then split into 2 blocks and participants attended to the first half in either the left or right ear (randomly assigned in EEG experiments, left ear first in behavioural experiments), with interference in the other, and then swapped ears for Block 2. Each block (half of a story) consisted of 60 sentences, with all 60 sentences concatenated with a 300ms gap between them to create a single block lasting 3.3 minutes. Block 1 was always the first half of the target story and Block 2 the second half.

All stories' volumes were normalised to ensure equivalent average amplitude. The stimulus of Musical Rain used in EEG1 and the behavioural studies was identical in length, root mean squared level and long-term spectrotemporal distribution of energy to its corresponding target story, but did not trigger a speech percept (Uppenkamp et al., 2006). It was generated in MATLAB by extracting temporal envelopes of the target sentences and filling them with 10ms fragments of synthesised vowels jittered in frequency and periodicity.

Instructions were recorded by the same female speaker of the target stories. These were played before each block in the target ear, telling the participant: *"This is your right/left ear. Please listen carefully to the story in this ear, on your right/left side, and ignore the story or sound in the other ear".*

Comprehension. To ensure that participants were attending to the target story, they were instructed at the start of each experiment that they would have to answer ten either/or questions about the target story at the end of each block. Participants were assured that these were straightforward questions – e.g., "This is a story about a KING/QUEEN" and they were asked to choose the correct response each time.

There were ten questions after each block, resulting in behavioural scores out of a total of 80 questions for the participants in the EEG experiments, and 40 comprehension questions in the behavioural studies.

The questions covered the content at the beginning, middle and end of each story and the distribution of correct responses was evenly distributed between the first and second option to each question. The comprehension questions can be found in Appendix B.

Figure 1 is an illustration of the dichotic listening task procedure.

Figure 1

Dichotic listening task experimental procedure



Note : Participants were instructed to attend to one side. The stimuli were presented for 3.3 min, and participants were then asked to complete 10 comprehension questions about the attended story.

3.3 EEG

3.3.1 Overview

EEG (electroencephalography) is a method of recording surface electrical activity, via multi-electrode recordings from the cortex, which represents the neural activity beneath (Pizzagalli, 2007). The output is data showing the voltages of the brain's electrical activity in different states (e.g., resting state, response to stimulus). EEG data can be analysed in different ways: as ERPs (event-related potentials, averaged EEG responses that are time-locked to the processing of stimuli); or ongoing EEG signals, for example neuronal oscillations (signals correlating with, but not precisely time locked with, cognitive events). An additional advantage of EEG is the multidimensional nature of the data, which describe time, space, frequency and power of the signal and thus allow a breadth of analysis (Cohen, 2014). EEG data provide coarse spatial but high temporal resolution, which makes EEG suitable for tracking overall response to stimuli where temporal precision is prioritised over spatial localisation. Indeed, the high temporal resolution of EEG, reflecting direct activation of neurons in response to stimulus, is especially suited to a study investigating neural encoding of speech.

EEG is particularly suitable for use with children (Bell & Cuevas, 2012). It is noninvasive and relatively simple to administer; the caps/nets can accommodate different head sizes; the recording session can be varied according to the participants' attention span; the equipment is not particularly cumbersome and its application does not distress children or cause claustrophobia; and the data are of sufficient quality for meaningful analysis after pre-processing, even allowing for agerelated interference. Consequently, it has been used successfully in a range of studies with children including infants (Kalashnikova et al., 2018), younger children (Stevens et al., 2009; Coch et al., 2005; Sanders et al., 2006; Bartgis et al., 2003) and pre-adolescents (Power et al., 2013; Gomes et al., 2007). It is thus an established technique for neural data collection of the target age group of children aged 7-12 for these studies.

3.3.2 EEG in speech and attention tasks

One of the established ways of assessing the brain's processing of speech is by recording the neuronal oscillations mentioned above, and assessing how well they track (or encode) properties of the speech, such as the speech envelope, or semantic information. The neural encoding of speech envelopes is a well-established method for investigating attentional processing, following the 'selective entrainment hypothesis' (Zion Golumbic et al., 2013; Giraud & Poeppel, 2012b), which proposes that attention causes low-frequency neural oscillations to entrain to the temporal envelope of speech. As described in Chapter 2, there is a large body of evidence confirming robust correlation between attended speech envelopes and neural activity (Aiken & Picton, 2008; Di Liberto et al., 2018; Mesgarani & Chang, 2012; Olguin et al., 2018) and showing that the neural encoding of speech envelopes plays an important role for speech intelligibility in both adults (Obleser & Kayser, 2019) and children (Power et al., 2016; Ríos-López et al., 2020).

When this is applied to the cocktail party task, it has been shown that as listeners attend to one speech stream in the presence of competing interference, auditory neurons differentially track changes in the speech envelope of the attended stream (O'Sullivan et al., 2015). Furthermore, the pattern of this tracking differs according to both the type of interference (Olguin et al., 2018), and the language group of participants (Olguin et al., 2019).

Neural data and speech properties can be analysed in a number of ways when testing for selective entrainment. Methods have included calculating crosscorrelations across different channels and time points (Olguin et al., 2019; Olguin et al., 2018); computing the cross-correlation at different frequencies of the signal (Kubanek et al., 2013), and using the mTRF toolbox (Crosse et al., 2016) to integrate the inputs of neural and speech data and compute either a forward predictive correlation (TRF or temporal response function) or backward correlation

(reconstruction score), both of which are widely accepted as robust indices of neural tracking (see Jessen et al., 2021 for a comparison of the methods).

In the neural studies presented here, I employed EEG to capture the neural encoding of attended and unattended speech envelopes in the groups of monolingual and bilingual children. I then used the mTRF toolbox (Crosse et al., 2016) to model the relationship between the speech envelopes and the neural data, and applied it in a backward direction to assess how well the attended and unattended speech envelopes could be reconstructed from the responses of the neuronal populations that encode them. The workings of the mTRF toolbox are explained in more detail in section 3.3.6.

3.3.3 The use of EEG in this research

3.3.4 Procedure

For the neural experiments, EEG was recorded using a child-sized 64-channel electrode net (Electrical Geodesics Inc., Eugene, OR, USA), connected to Netstation software. Conductive gel was applied to the electrodes to maximise connectivity and reduce interference. Impedances were checked during the gelling process and the experiment only proceeded when all impedances were under 100 Ω . The stimuli were played through foam-tipped earphones in the pre-allocated part-randomised order. During the presentation of the stimuli, voltages were recorded at a sampling rate of 500 Hz. The raw data were then pre-processed in MATLAB: EEGLAB Toolbox (Delorme & Makeig, 2004) before using them as an input to the mTRF toolbox.

3.3.5 Preprocessing

The EEG data were imported, cleaned and prepared for the toolbox using the following 10 step pre-processing pipeline:

- Raw data was imported to EEGLAB; channels 61-64 (located in muscular/facial areas) were removed, leaving data from 60 channels for processing. Data was filtered between 1-100Hz using zero-phase bandpass Hamming windowed FIR filters (transition band widths of 1Hz with cutoff frequencies at –6 dB) and down-sampled to 250 Hz.
- 2) Markers were inserted into the EEG data for each sentence onset in attended and unattended conditions. These were identified and inserted using the .mat file generated per participant, which listed the part-randomised order of story presentation; and the signal information from the story recordings, with the exact length of each sentence. These markers could not be added during acquisition, as the stories were played as continuous streams. Markers were inserted for individual sentences denoting status (A = Attended, U = Unattended), Block Number (1 or 2), Condition (A-D for Conditions 1-4) and sentence number (1:60). For example, the 30th sentence in the Block 2 of the Latin story of Condition 3 was marked U2C30 – Unattended/Block 2/Condition 3/Sentence 30.
- Bad channels were identified via probability and kurtosis if they were 5 SD away from the mean kurtosis and 3 SD from the mean power spectrum and marked in red.
- 4) Independent Component Analysis (ICA) algorithm (EEGLAB) was run to identify components corresponding to artefacts (e.g. eye blinks) which contaminate data. The participants' age in these studies made the data especially vulnerable to artefacts from non-brain activity. The components identified by ICA were visually inspected, by reviewing topographies, spectral traits and time course, and bad components were removed manually from the data.
- 5) After ICA, epochs were extracted, starting at 200ms pre-onset of the sentence and ending at 2800ms post onset, resulting in 480 attended epochs and 360 unattended epochs per participant. This length of epoch was chosen so that, after allowing for epoch rejection, there would be a minimum threshold of five minutes of data per condition for input to the mTRF toolbox (as recommended by Jessen et al., 2019).

- 6) The bad channels identified in step 3) were interpolated, via spherical interpolation. This replaced missing data from faulty electrodes with estimates based on the activity and locations of surrounding electrodes.
- 7) After the bad channels were interpolated, sub-standard epochs were rejected with the *pop_autorej* function (EEGLAB), removing epochs with values outside a 3 SD of the probability and kurtosis thresholds.
- 8) The epoch rejection rate was calculated for both attended and unattended datasets in each condition and by group. In Experiment 1, this resulted in an overall epoch rejection of 16.33% for all participants' data (18.31% for monolinguals and 14.35% for bilinguals). In EEG Experiment 2 this was 15.96% overall (14.36% for monolinguals and 17.47% for bilinguals). This is consistent with epoch rejection rates for children of a similar age range (Barry et al., 2004; Picton et al., 2000).
- 9) Next, the EEG data, which had been recorded using the vertex (Cz) electrode as reference, were re-referenced to the average of all channels using the *pop_reref* function (EEGLAB). This process calculates the average of the signal across all electrodes and then subtracts it from the EEG signal at individual electrodes and timepoints. This ensures that the voltage at each electrode reflects what is unique to that location relative to all the other electrodes, independently of reference location (Nunez et al., 2006).
- 10) Finally, data were resampled to 100 Hz to reduce the computational load and match the speech envelope for processing in the mTRF toolbox.

Following this process, the EEG data from all participants were entered into the subsequent mTRF analysis to be cross-validated with the corresponding speech envelopes.

3.3.6 mTRF

Neural tracking of the stimulus envelopes was computed using multivariate temporal response functions, as implemented in the mTRF toolbox (Crosse et al., 2016).

TRF can be considered as a filter that describes the linear transformation of the ongoing stimulus to the ongoing neural response. This approach uses linear regression to model the relationship between each speech input and the signal at each EEG channel (Di Liberto et al., 2015). The inputs to the calculation of the TRF model in the toolbox are the stimulus (in this case the normalised speech envelope), response (normalised EEG data), minimum and maximum time lags, sampling rate and a series of ridge parameters or smoothing constants (as detailed in Crosse, Di Liberto, Bednar, & Lalor, 2016).

The model is then used to either "predict" the ongoing brain response from the stimulus (the forward or encoding model); or "reconstruct" features of the stimulus from the EEG data (backward or decoding model) and this reconstruction is compared to the original stimulus, resulting in a reconstruction accuracy score (Pearson's correlation, r). The forward model is analogous to regularised linear regression; the backward model (reconstruction) has the advantage of mapping all available neural data simultaneously across all channels and is therefore particularly suited to multi-channel systems such as EEG. The prediction and reconstruction accuracy scores computed by the forward and backward models are widely accepted as measures of encoding accuracy, both generally (Power et al., 2012), for children (Power et al., 2016; Jessen et al., 2019; Kalashnikova et al., 2018), and across modalities (O'Sullivan et al., 2017), and are consistent with other computations of neural tracking. The mTRF technique is especially appropriate for natural speech (Broderick et al., 2018; Ding & Simon, 2014), a stimulus which is more ecologically valid than artificial, contrived stimuli (Hamilton & Huth, 2020; Huk et al., 2018; Matusz et al., 2018).

For the purpose of these experiments, I chose to measure stimulus reconstruction accuracy (backward model). The backward model is arguably preferable to both the forward model and traditional cross-correlation approaches (Crosse et al., 2016) as it uses information from all EEG channels simultaneously to reconstruct the speech envelope, calibrating the relative influence of different channels so that informative channels receive greater weights than those which provide less data, and dividing

out any autocovariance between channels. This way, even stimulus features that are not explicitly encoded in the neural response in a one-to-one mapping may be inferred from correlated input features that are encoded, which would not be the case using direct correlation to the raw signal. The inputs to the calculation of the TRF models were the stimulus (normalised speech envelope), response (normalised EEG data), minimum and maximum time lags (0 and 250ms), sampling rate (100 Hz) and a series of ridge regression parameters (12 λ values: 1×10⁻³:1×10⁸).

To calculate the models, I created matrices of EEG data and matching stimuli for each attended and unattended epoch per condition per participant per group. The size of the matrices corresponded to the number of viable epochs per condition (minimum 100 for a single condition in each participant). Decoder weights over time lags from 0 to 250ms were calculated using the cross validation (mTRFcrossval) function. The cross validation uses a 'leave-one-out' computation which first fits individual models to every trial for each specified λ , then excludes one trial at a time ('test set') while averaging the others across models ('training set'). The averaged model from the training set is then convolved with the test set to generate a stimulus reconstruction. In each model, this was done in rotation with each trial serving once as the 'test set', repeated across all 12 λ values. Figure 2 illustrates the computation of the mTRF model.

Figure 2

Stimulus reconstruction using the mTRF toolbox



Note: mTRF stimulus reconstruction: A backwards mTRF decoding model was fit separately to the speech envelope of each of trials for each participant, using a leave-one-out cross-validation procedure. This generated a reconstruction of each speech envelope that was validated against the original stimulus envelope.

Each reconstruction was then validated against the original stimulus, resulting in 12 reconstruction accuracy scores (Pearson's r) per stimulus, with the *r* value at the optimal λ (identified as that which yields the highest overall *r*-value across epochs) taken. This optimal lambda value selection mitigated against the potential overfitting of the TRF model.

Figure 3 illustrates the outcome of reconstruction for a sample sentence '*This cat* was getting skinnier and skinnier'.

Figure 3

Reconstruction from a sample sentence



Note: The blue line shows the speech envelope from one trial of the original stimulus. The orange line is the estimate of the envelope reconstructed by the decoder. The reconstruction accuracy score (r) is a measure of the correlation between the original (blue) and reconstructed speech envelope (orange). Resulting r values per sentence per condition per participant were used in statistical analyses.

The reconstruction accuracy scores for each cell of the matrices (corresponding epochs and speech envelopes) were then compared across groups, attention status and condition using linear mixed-effect models (Baayen et al., 2008) as implemented in the Ime4 R package (Bates et al., 2014). The step function in the ImerTest package (Kuznetsova et al., 2017) was used to arrive at the best-fitting model. The Satterthwaite approximation (Satterthwaite, 1946) was used for degrees of freedom. Significant p-values are reported in the results section at p<.05. All post-hoc tests were FDR corrected for multiple comparisons.

3.3.7 Speech envelopes

Speech envelopes were calculated from the .wav files of the recorded transcripts, using the Hilbert2 function in EEGLAB, downsampled to 100 Hz to match the data

and normalised using nt_normcol (Noisetools

<u>http://audition.ens.fr/adc/NoiseTools/</u>). They were then used to cross-validate with the EEG data in the mTRF toolbox.

3.4 Dual task

3.4.1 Overview

The dual task paradigm extends the selective attention demands made in one modality from a single task, and puts extra pressure on overall attentional capacity by making simultaneous demands in another, different, task. Research into the impact of dual tasks on performance can reveal fundamental aspects on the mechanisms and limitations of attentional processing, such as the consistent psychological refractory period (PRP), the delay generated when performing two tasks simultaneously. This delay is attributed to the need to execute both tasks' demands with a limited-capacity processor (Kahnemann, 1973), which can only process a restricted amount of information at any given point (Broadbent, 1965; Clark & Dukas, 2003). In a dual-task design, the tasks presented can be in the same or different modalities and often include perceptual-motor tasks, which have the advantage of precise measurement of reaction times per experimental trial, showing differences in milliseconds as the processing load increases. Typically, either two RT tasks are presented simultaneously, or a motor response task combined with a separate demand on cognitive processing, for example a working memory task. In all versions of a dual task paradigm, the two tasks are defined as independent, with distinct goals and separate measurements.

In the dual task paradigm for the behavioural experiments used here, participants were required to perform tasks which competed for their attention in two modalities: visual and auditory. The primary task was an edited version of the dichotic listening task used in the previous EEG experiment. In contrast to previous experiments however, the auditory task, which retained its varying levels of

interference, was now presented in conjunction with a visual distractor task, creating a hybrid paradigm, with increased cognitive demands, and whose dual modality was consistent with a theory of central processing (Koch et al., 2018).

The visual distractor task was a continuous stimulus-response task, adapted from the T.O.V.A. (Test of Variables of Attention) task, a computerised fixed-interval continuous performance test that assesses inhibition and vigilance (Greenberg & Waldmant, 1993). Participants were asked to perform the task in front of a computer screen with a keyboard, whilst wearing headphones connected to the computer audio. The visual task demanded either a positive response (pressing the space bar), or inhibition of response (not pressing the space bar), to a stimulus on a screen. All behavioural experiments consisted of five conditions. The first condition (control) only tested the visual modality, without the concurrent auditory task. The next condition placed an additional load on the participant by requiring them to concentrate on a story in English streamed into one side of their headphones, whilst still performing the visual task, and answer comprehension questions on the story after the trial. The three subsequent conditions further increased the attentional load by requiring the participant to ignore a distractor story or stream, whilst listening to a target story in English, and simultaneously performing the visual task (T.O.V.A.). The details of the dichotic listening task are in section 3.2.

3.4.2 The visual task (T.O.V.A.)

The visual task was prepared and presented in Psytoolkit (Stoet, 2010; 'Stoet, 2017). The task required pressing the SPACE bar as quickly as possible when the cartoon picture of a dog appeared at the top of the screen. If the dog appeared at the bottom of the screen, the participant was instructed to ignore and wait for the next presentation. Each dog image was on the screen for a maximum of 2000ms. If the SPACE bar was pressed correctly, the reaction time was recorded and the next trial began immediately. If the participant failed to press the SPACE bar when the dog appeared at the top of the screen (error of omission), the trial was recorded as a

target error. If the participant incorrectly pressed the SPACE bar when the dog was at the bottom of the screen (error of commission), the reaction time was recorded, the trial was coded as a distractor error, and the next stimulus appeared immediately. The duration of the visual task was designed to align with that of the audio task.

The ratio of target:non-target stimulus was coded to be 1:3 with the 24 targets and 72 non-targets appearing in a random order in each round. These 96 visual stimuli, with correct responses, and a 500ms break between presentations, aligned with the 3.3 min length of the auditory stimulus per condition. If, however, a participant completed the visual task before the end of the auditory stream, the experiment randomly presented extra stimuli in the same 1:3 ratio until the story and condition ended.

In the first condition, the participant only performed the visual task ('Control'). There was a brief explanation of the task and practice session (which could be repeated, if the participant needed), then the first condition began while there was silence in the headphones. From the second condition onwards, the visual task was repeated in exactly the same way, but in a random order, while the auditory task was also presented.

Figure 4 illustrates the T.O.V.A. procedure for this study and how it was combined with the auditory task.

Figure 4



T.O.V.A. Experimental Procedure

Note: Stimuli were presented randomly in a target:non-target ratio of 1:3 for a maximum of 2000ms. A blank screen appeared for 500ms in between presentations. A picture of the dog at the top of the screen was the target requiring a response within 2000ms. A presentation of the dog at the bottom of the screen was the distractor requiring an inhibition of response for 2000ms. Condition 1 was the control condition, in which the participant only performed the visual task. From Condition 2 onwards, participants were instructed to listen to a story in one ear (which alternated between rounds), whilst also responding to the visual task. In Conditions 3-5 the participant was instructed to attend to the target story in one ear and ignore a distractor stream in the other ear, whilst also responding to the visual task. After the auditory and visual tasks ended, the participant answered 10 comprehension questions about the target story.

Chapter 4: Experiment 1

4.1 Introduction

One of the key concepts presented in Introduction chapter was the importance of selective attention in the development of cognition and educational outcomes. Selective attention has been proposed as one of the key foundational skills for children's academic success (Stevens & Bavelier, 2012; Hampton Wray et al., 2017), due to its influence on factors such as inhibitory control (Reck & Hund, 2011), working memory (Veer et al., 2017), speech (L. B. Astheimer & Sanders, 2012), metalinguistic skills (L. Astheimer et al., 2014) and arithmetic (Moll et al., 2015). Thus any influences on this fundamental developmental process have the potential to be highly significant.

Selective attention, like many other executive functions, is subject to substantial individual differences (Hunt et al., 1989; Ruff et al., 1998). Nevertheless, there is consensus that there are multiple factors that can also influence its development across a group level (Mezzacappa, 2004), ranging from biological characteristics, such as age (Plude et al., 1994), or preterm birth (van de Weijer-Bergsma et al., 2008; Mulder et al., 2009); and environmental factors which are correlated both with a negative impact, such as lower SES (Hampton Wray et al., 2017; Stevens, Lauinger, & Neville, 2009; Isbell, Wray, & Neville, 2016; Karns, Isbell, Giuliano, & Neville, 2015; D'Angiulli, Herdman, Stapells, & Hertzman, 2008); or are regarded as a positive influence, such as musical experience (Benz et al., 2016; Kasuya-Ueba et al., 2020).

Another such factor that has been linked to modifications to selective attention in children is bilingualism. This is due to the fact that learning and using multiple languages is a major processing demand for the cognitive system, with evidence showing that bilingual language use leads to parallel activation and competition between the two languages, requiring the speaker to selectively select one and

inhibit the other (Green, 1998; Costa et al., 2008; Bialystok, 2015; see Chapter 2 for neural evidence substantiating this theory). The bilingual neurocognitive system accommodates these additional processing demands by modifying and adapting the underlying neural and functional architecture, as evidenced by a large number of studies in both children and adults (Filippi et al., 2011; Archila-Suerte et al., 2018; Pliatsikas et al., 2020). Importantly, effects of bilingualism on aspects of neurocognitive processing have been observed from very early on, with data showing differences between monolingual and bilingual infants as young as 4-6 months old (Comishen et al., 2019; Nacar Garcia et al., 2018) as well as in older children (Barac et al., 2016; Arredondo et al., 2017).

However, the nature of these adaptive changes is still not entirely clear and is the subject of vigorous debate. One widely held view is that the increased processing demands arising from bilingual language use lead to enhanced capacity for selective attention, resulting in better performance for bilingual children on selective attention tasks (Bialystok, 2017; Poulin-Dubois et al., 2011). However this view has also been challenged, with a number of reports either not finding evidence for enhanced performance in bilinguals (Antón et al., 2014; Duñabeitia et al., 2014; Paap & Greenberg, 2013) or arguing that they can be accounted for by variables other than bilingual experience, such as working memory (Namazi & Thordardottir, 2010), or are only discernible in specific contexts (Paap et al., 2015).

Furthermore, using primarily behavioural studies to test for the neurocognitive effects of bilingualism is proving to be inadequate. There is ambiguity about whether the tests measure the executive functions they claim: separate studies either use different tests (e.g., Simon, Stroop, ANT and flanker) to measure the same function; or the same task (e.g., Stroop) to measure different functions, with inconsistent results (Paap, Johnson, & Sawi, 2015; Valian, 2015a). This ambivalence has been highlighted by neural studies showing that the same behaviours can arise from multiple cognitive processes. For example, an analysis of fMRI data (Ali et al., 2010) found that different brain areas were activated for the Stroop (left head of caudate) and the Simon task (prefrontal areas and left globus pallidus); both of

which are considered to be tasks of controlled attention. A comparison of auditory and visual Stroop (Roberts & Hall, 2008) showed different areas of brain activation for the management of conflict in each domain. Similarly, when comparing bilingual performance, separate studies using a switching (Garbin et al., 2010) and flanker tasks (Abutalebi et al., 2012) generated activation in different brain areas.

Notably, several neuroimaging studies have identified neural differences between monolinguals and bilinguals even when behavioural outcomes are comparable on the same tasks (Bialystok et al., 2005; Kousaie & Phillips, 2012; Luk et al., 2010; Olguin et al., 2019). These apparently incompatible results suggest that the neuroadaptive changes from bilingualism might develop in order to enable bilingual children and adults to recruit mental resources differently (Kroll & Bialystok, 2013) to achieve comparable performance. This shifts the debate from bilingualism itself creating advantages or disadvantages, to an exploration of bilinguals' different "constellation of skills" (Hoff et al., 2012) which manifest in neuroplastic changes, and are more subtle than what most behavioural tests to date have elicited.

One such exploration was a study investigating neural and behavioural responses of monolinguals and bilinguals in a selective attention task (Olguin et al., 2019). The researchers compared behavioural performance and the neural encoding of attended and unattended spoken narratives in monolingual and bilingual adults. Participants were instructed to listen to a story in their native language, while ignoring different types of linguistic interference presented in the other ear. The results showed that, even though the respondents' comprehension scores were the same, there were significant differences in the pattern of neural encoding of the attended streams between the monolingual and the bilingual group.

With this is mind, a somewhat different interpretation of the mechanisms of neurocognitive adaptations to selective attention in bilingualism might be that they emerge in order to enable bilingual speakers to achieve and maintain optimal behavioural performance under the increased processing demands of bilingualism. Importantly, this compensation for the more complex processing environment is

achieved in the context of a finite selective attention capacity. As discussed in the Introduction chapter, this account acknowledges that attention is ultimately limited such that the cognitive system can only process a restricted amount of information at any given point (Broadbent, 1965; Clark & Dukas, 2003), and that selective attention might also require processing capacity itself (Johnston & Heinz, 1978; Kahneman, 1973). In the bilingual context this could mean that the process of selecting the target language and inhibiting the non-target one will itself utilise some of the existing attentional resources (Wickens, 2007). This would then impact on the remaining attentional resources such that they need to be economised in order to support optimal task completion. This view builds on, and extends, the hypothesis that bilingual control processes themselves adapt to the recurrent processing demands placed upon them, as proposed in the adaptive control hypothesis (Green & Abutalebi, 2013; Abutalebi & Green, 2016), and while it does not preclude the possibility that this may lead to greater flexibility in the usage of the residual capacity, it shifts the focus from the often-inconsistent behavioural comparisons to the patterns of modification and adaptation in the underlying neural and functional architecture. Critically however, this account also gives rise to a different set of predictions about the patterns of these underlying adaptations. In particular, instead of assuming an overall enhancement in neural indices of attentional processing for bilinguals compared to monolinguals, this view predicts no increase - or possibly even a slight reduction - combined with their different distribution as determined by the requirements of the task at hand.

To date, studies exploring the above possibilities have focused primarily on adult samples. It is therefore unclear what the developmental trajectory of these neuroplastic modifications might be, and to what extent they affect the mechanisms of selective attention in bilingual children. One possibility is that the demands of competition and inhibition between co-activated languages reconfigure the patterns of attentional processes right from the onset ("Perceptual Wedge Hypothesis", Petitto et al., 2012), such that the effects can be clearly discerned by the time children can respond to selective attention tasks. Alternatively, these modifications might have a protracted maturation dependent on exposure to the demands of

bilingualism, in which case they would be much more evident in adults than in children.

This chapter's experiment (EEG Experiment 1) therefore aimed to explore how the demands of bilingualism modify the neural mechanisms of selective attention in children. Building on the existing data from Olguin et al. (2019), the aim of EEG Experiment 1 was to establish if bilingual and monolingual children also demonstrate the differential neural encoding of the speech envelope in the face of different distractors; and consider the implications of resultant behavioural and neural indices in the context of theories of enhanced bilingual selective attention (Bialystok, 2017) and constraints on bilingual neural resource (Wickens, 2007).

4.2 Current study

To dissociate between the alternatives presented in the Introduction, EEG Experiment 1 explored how bilingualism modulates attentional processing in children through assessing the neural encoding of attended and unattended speech envelopes in monolingual and bilingual listeners aged 7-12. The neural encoding of speech envelopes is a well-established method for investigating attentional processing. According to the 'selective entrainment hypothesis' (Zion Golumbic et al., 2013; Giraud & Poeppel, 2012b), attention causes low-frequency neural oscillations to entrain to the temporal envelope of speech. Supporting this, there is a large body of evidence confirming robust correlation between attended speech envelopes and neural activity (Aiken & Picton, 2008; Di Liberto et al., 2018; Mesgarani & Chang, 2012; Olguin et al., 2018) and showing that the neural encoding of speech envelopes plays an important role for speech intelligibility in both adults (Obleser & Kayser, 2019) and children (Power et al., 2016; Ríos-López et al., 2020). The current study thus employed EEG to capture the neural encoding of attended speech envelopes in monolingual and bilingual children. Linear regression as implemented in the mTRF toolbox (Crosse et al., 2016) was used to model the relationship between the speech signal and the neural data, and applied it in a
backward direction to assess how well the attended and unattended speech envelopes could be reconstructed from the responses of the neuronal populations that encode them (see Methods chapter for more details). The accuracy of speech envelope reconstructions from the EEG data was assessed by comparing the reconstructions to the original speech envelopes, resulting in reconstruction accuracy scores (Pearson's r) - where the higher r value signifies that more stimulusrelevant information was encoded in the EEG signal and the better model could be created, leading to a better reconstruction. Reconstruction scores calculated this way are widely accepted as measures of neural encoding in children (Power et al., 2016; Kalashnikova et al., 2018) and are consistent with other computations of cortical tracking (Sassenhagen, 2019). Another feature of the reconstruction method is that it maps all available neural data simultaneously and is therefore specifically suited to multi-channel systems such as EEG. The mTRF technique has also been shown to be particularly suitable for natural speech (Di Liberto et al., 2018; Broderick et al., 2018), and thus highly appropriate for the streams of spoken narratives used in the design of this experiment.

Another important consideration for investigation into the ways bilingualism shapes selective attention in children is the trajectory of selective attention development. As described in the Introduction chapter, although auditory selective attention is sufficiently developed by age 4 to respond to auditory dichotic tasks (Stevens & Bavelier, 2012; Hampton Wray et al., 2017), an older age range is recommended due to younger children's high variability in performance (Sanders et al., 2006; Takio et al., 2009). Auditory selective attention typically stabilises from age 7 upwards (Gomes et al., 2007; Jones et al., 2015). Given these considerations, participants in the age range of 7-12 were recruited for this experiment, as this age range not only represents a developmental plateau for selective attention in childhood, but is also likely to generate relatively stable effects whilst ensuring that children can reliably perform the selective attention task element.

To investigate whether and how bilingualism modifies the neural mechanisms of selective attention in children, the current experiment used a dichotic listening task,

the origins of which (Cherry, 1953) are explained in the Introduction chapter. Following the design used previously for adults (Olguin et al., 2019), children were presented with two competing narratives simultaneously and instructed to attend to one while ignoring the other. The nature of the competing stream was manipulated across four different conditions to create perceptual or linguistic interference. The first condition was 'Single Talker', a control condition where children attended to a narrative presented in one ear, with no interference presented in the other ear. This allowed us to establish the extent of attentional encoding in monolingual and bilingual listeners at baseline (i.e., without any interference present). In the second condition, children attended to a narrative in English presented in the attended ear while ignoring another English story presented in the unattended ear (English-English condition). In the third condition, children attended to a narrative in English while ignoring a narrative in Latin, a language unknown to them (English-Latin condition). These two conditions therefore tested attentional encoding in the context of linguistic interference, where the known language distractor (English) could be expected to interfere more strongly with the attended stream than the language that children cannot process for meaning (Latin). In the fourth condition, the interfering stream was Musical Rain (MuR), a stream derived from speech that is closely matched to the acoustic properties of speech, but does not trigger speech percept and is therefore expected to only engage low-level acoustic processing (English–MuR condition). Another key feature of this design was that participants were instructed to listen to the attended stream for comprehension, a task that children in this age group were expected to able to do without difficultly. Based on the existing adult data (Olguin et al., 2019) it was also expected that there would be no significant difference between the ability of monolingual and bilingual listeners to perform the task. By equating on behavioural performance, this approach enabled a focus on the patterns of modification of the mechanisms that underpin selective attention, rather than performance per se.

The set of conditions described above allowed the study to investigate whether bilingualism modifies the neural correlates of selective attention in children, and to assess directly the predictions of the two hypothesised mechanisms of this

modification which were presented earlier. Following the existing evidence (Olguin et al., 2019; Di Liberto et al., 2018; Mesgarani & Chang, 2012; Olguin et al., 2018), it was assumed that attention would modulate the neural encoding of speech envelope in both monolingual and bilingual children, with the type of distractor probably further influencing the strength of the encoding of the attended stream. Critically however, it was also assumed that the way different distractors influence attentional encoding might differ between the groups. According to the hypothesis that bilingual experience leads to general enhancement of attentional processing, results would be expected to show an overall increase in reconstruction accuracy scores for bilingual compared to monolingual children. Specifically, while the overall pattern of effects might be similar in the two groups – with distractors narrated in a language likely causing stronger interference than the non-linguistic distractor and the Single Talker condition - all these markers of attentional encoding would be expected to be enhanced in the bilingual group. On the other hand however, there might be no discernible increase, or even a decrease in the indices of attentional encoding in bilingual children, reflecting the hypothesis that language selection and inhibition themselves might draw on the existing attentional capacity, restricting the resources available to track the speech envelope. In addition and more importantly, this could lead to a modification of the encoding patterns across conditions in bilinguals, suggesting that the remaining attentional capacity would have been distributed to maximise this finite resource and meet the task requirements in the context of increased processing demands of bilingualism.

4.3 Materials and method

4.3.1 Participants

48 typically-developing children aged 7-12 were tested, comparable to the sample size of similar EEG studies on children (Di Liberto et al., 2018; Power et al., 2016; Kalashnikova et al., 2018; Stevens et al., 2009). They were split into two categories: bilingual (n=24, sixteen males, age M=9.3 yr, SD=1.83) and monolingual (n=24, thirteen males, age M=9.6 yr, SD=1.48), which were matched groupwise on mean and distribution of age (t=.54, p=.59). All participants were healthy with no history of hearing problems or neurological disorder. 43 were right-handed, with four of the left-handed children being monolingual and one bilingual. All participants' parents completed a language history questionnaire, which provided an overview of children's exposure to languages. As confirmed by the questionnaire, all monolingual participants were native speakers of English, with no significant exposure to other languages. The participants in the bilingual group all had a similar profile: the language they first learnt was not English, and they used this language at home on a daily basis. They were however fluent and highly proficient in English, following English-speaking curriculum at school, and with native-like English conversation skills comparable to their monolingual peers. The second languages spoken were Afrikaans, French, Finnish, Greek, Hindi, Hungarian, Igbo, Japanese, Lithuanian, Mandarin, Polish and Turkish. Additionally, two children spoke a third language proficiently (French and Spanish), and one spoke a total of four languages other than English proficiently (Arabic, French, Hebrew and Spanish). Children were recruited via posters, social media, and word of mouth. Parental education information was collected as an indication of SES, a well-documented influence on selective attention in children (Stevens et al., 2009) and modulator of the cognitive effects of bilingualism (Naeem et al., 2018). The majority of participants' parents (87.2%) were educated to degree level or higher, and the groups were not significantly different on this approximation of SES (bilinguals M=2.56, SD=.52; monolinguals M=2.35, SD=.79; Mann-Whitney U=259, p =.53).

The study was approved by the Cambridge Psychology Research Ethics Committee, and performed in accordance with relevant guidelines and regulations. Prior to the testing session, parents and children were given detailed information on the aims of the project and what to expect from the session. Upon arrival, informed consent was given by parents signing a consent form and the children an assent form. They were told they could withdraw from the study at any time.

1.3.2 Design

The experiment consisted of four conditions (Table 1). In each condition, children were attending to a story in English in one ear. Condition 1 had no interference in the other ear ('Single Talker'). In the other three conditions children were also presented with a distractor in the other ear, which they were instructed to ignore. The nature of the distractor varied, from a different story in English ('English-English'), to a story in a language unknown to children ('English-Latin') and non-linguistic acoustic interference ('English-Musical Rain').

Table 1

Experimental Conditions

Condition	Attended	Interference				
condition	stream					
<u>1. Single Talker</u>	English story 1	No interference				
2. English-English	English story 2	English story 5				
3. English-Latin	English story 3	Latin story				
<u>4. English-MuR</u>	English story 4	Musical Rain (acoustic stream)				

4.3.3 Target streams

The target stories for the attended ear were four children's stories in English specifically aimed at this age group, taken from online resource storynory.com. All stories were transcribed into 120 sentences each, with each sentence lasting approximately 3 seconds in length. Each target story was then split into 2 blocks and children attended to the first half in either the left or right ear (randomly assigned), with interference in the other, and then swapped ears for Block 2. Each block (half of a story) consisted of 60 sentences, with all 60 sentences concatenated with a 300ms gap between them to create a single block lasting 3.3 minutes. Block 1 was always the first half of the target story and Block 2 the second half. Gender of the speaker was kept the same for all stories (same female voice for all target stories, different female voice for interference), to reduce segregation strategies based on talker's gender (Brungart & Simpson, 2007). All stories' volumes were normalised to ensure equivalent average amplitude.

4.3.4 Distractor streams

There was no interference in Condition 1, to provide a baseline against which to compare different levels of distractor. The interference in Condition 2 was a fifth children's story in English sourced from storynory.com, adapted in the same way as the target stories and recorded by a second female native English speaker (i.e., different from the female narrator of the target stories). For the unknown language interference in Condition 3, Latin was chosen as a non-artificial language, which would almost certainly be unknown to the participants. The narrative was sourced from teaching resource *Ritchie's Fabulae Faciles* and was recorded by a native English female who was a postgraduate Classics student at the University of Cambridge.

The acoustic interference of Musical Rain used in Condition 4 was identical in length, root mean squared level and long-term spectrotemporal distribution of energy to the target story in Condition 4, but did not trigger a speech percept (Uppenkamp et al., 2006). It was generated in MATLAB by extracting temporal envelopes of the target sentences and filling them with 10ms fragments of synthesised vowels jittered in frequency and periodicity. The resulting stream was described by participants as "the sound of a jug pouring water".

Instructions were recorded by the same female speaker of the target stories. These were played before each block in the target ear, telling the child: *"This is your right/left ear. Please listen carefully to the story in this ear, on your right/left side, and ignore the story or sound in the other ear".*

4.3.5 Procedure

The participants had a practice session of listening to an English story in both the left and the right ear while ignoring a distracting English story in the other ear, in order to familiarise themselves with the dichotic listening paradigm. After practice, they were asked to summarise the target story to check they could hear correctly and understood the instructions to attend to one ear at a time. The task itself took 45-60 minutes across the four conditions. Children first heard Block 1 of Condition 1 (Single Talker) followed by 10 comprehension questions. They then listened to Block 2 of Condition 1 (Single Talker), again followed by 10 comprehension questions. Each block was preceded by the recorded instructions in the relevant (target) ear. This procedure was repeated for the other three conditions, which were presented in a random order. Children were instructed to stay as still as possible while the stories were playing and were allowed to stretch, yawn etc. during the comprehension breaks.

Comprehension questions consisted of simple sentences to check understanding of each story (for example: *'This story is about a funny old MAN/WOMAN'*), and children pointed or verbally confirmed which option they thought was correct. The children did not receive feedback on their responses. At the end of the experiment children were presented with a certificate of completion and compensation for their time. Figure 5 is an illustration of the procedure of the dichotic listening task in Condition 3.

Figure 5

Dichotic listening task procedure for EEG Experiment 1



4.3.6 Data collection and preprocessing

EEG was recorded using a 64-channel electrode net (Electrical Geodesics Inc., Eugene, OR, USA), connected to Netstation software. The stimuli were played through foam-tipped earphones in the pre-allocated part-randomised order. All data were pre-processed in MATLAB: EEGLAB Toolbox (Delorme & Makeig, 2004). Channels 61-64 (located in muscular/facial areas) were removed, leaving data from 60 channels for processing. Data was filtered between 1-100Hz using zero-phase bandpass Hamming windowed FIR filters (transition band widths of 1Hz with cutoff frequencies at –6 dB) and down-sampled to 250 Hz. Bad channels were identified via probability and kurtosis and were interpolated (via spherical interpolation) if they were 5 SD away from the mean kurtosis and 3 SD from the mean power spectrum. Independent Component Analysis (ICA) algorithm (EEGLAB) was conducted to identify components corresponding to artefacts (e.g. eye blinks). These were visually inspected and bad components removed from the data. After ICA, epochs were extracted, starting at 200ms pre-onset of the sentence and ending at 2800ms post onset. This length of epoch was chosen so that, after allowing for epoch rejection, there would be a minimum threshold of five minutes of data per condition for input to the mTRF toolbox (Jessen et al., 2019). After the bad channels were interpolated, bad epochs were rejected with the *pop_autorej* function (EEGLab), removing epochs with values outside a 3SD of the probability and kurtosis thresholds. This resulted in an overall epoch rejection of 16.33% for all participants' data (18.31% for monolinguals and 14.35% for bilinguals). By condition, epoch rejection was 14.67% in Single Talker, 19.68% in English-English, 12.77% in English-Latin and 17.36% in English-MuR datasets. Next, data were re-referenced to the average of all channels and finally resampled to 100 Hz to reduce the computational load. Following this process, the EEG data from all 48 participants were entered into the subsequent mTRF analysis.

4.3.7 Speech envelopes

Speech envelopes were calculated using the Hilbert2 function in EEGLAB, downsampled to 100HZ to match the data and normalised using *nt_normcol* (Noisetools http://audition.ens.fr/adc/NoiseTools/).

4.3.8 Analyses

Neural tracking of the stimulus envelopes was computed using multivariate temporal response functions, as implemented in the mTRF toolbox (Crosse et al., 2016). As explained in the Methods chapter, TRF uses linear regression to model the relationship between speech input and signal at each EEG channel. The backward model (reconstruction) was used, which has the advantage of mapping all available neural data simultaneously across all channels, calibrating their relative influence so

that informative channels receive greater weights than those which provide less data, and dividing out any autocovariance between channels. This way, even stimulus features that are not explicitly encoded in the neural response in a one-toone mapping may be inferred from correlated input features that are encoded, which would not be the case using direct correlation to the raw signal. The inputs to the calculation of the TRF models were the stimulus (normalised speech envelope), response (normalised EEG data), minimum and maximum time lags, sampling rate and a series of ridge regression parameters (λ). To calculate the models, I created matrices of EEG data and matching stimuli for each attended and unattended condition per participant per group. The size of the matrices corresponded to the number of viable epochs per condition (minimum 100 for a single condition in each participant). Decoder weights over time lags from 0 to 250ms were calculated using the cross validation (mTRFcrossval) function. The cross validation uses a 'leave-oneout' computation which first fits individual models to every trial for each specified λ , then excludes one trial at a time ('test set') while averaging the others across models ('training set'). The averaged model from the training set is then convolved with the test set to generate a stimulus reconstruction. In each model, this was done in rotation with each trial serving once as the 'test set', repeated across all λ values (12 λ values, 1×10^{-3} : 1×10^{8}). Each reconstruction was then validated against the original stimulus, resulting in 12 reconstruction accuracy scores (Pearson's r) per stimulus, with the r value at the optimal λ (identified as that which yields the highest overall rvalue across epochs) taken. This optimal lambda value selection mitigated against the potential overfitting of the TRF model. The reconstruction accuracy scores were then compared across groups, attention status and condition using linear mixedeffect models (Baayen et al., 2008) as implemented in the Ime4 R package (Bates et al., 2014). To arrive at the best-fitting model, the step function in the ImerTest package (Kuznetsova et al., 2017) was used. The Satterthwaite approximation (Satterthwaite, 1946) was used for degrees of freedom. Significant p-values are reported at p<.05. All post-hoc tests were FDR corrected for multiple comparisons. Figures 2 and 3 in the Methods chapter illustrate the procedure of mTRF model computation, and the outcome of reconstruction for a sample sentence.

4.4 Results

4.4.1 Behavioural comprehension scores

Children from both groups performed the task equally well, with overall comprehension scores of 98.1% in the monolingual group, and 98.7% in the bilingual group. A summary of comprehension scores and standard deviations are shown in Table 2 and Figure 6.

Table 2

Condition	Monolinguals	Bilinguals
Single Talker	99.6 (1.41)	99.2 (2.41)
English-English	95.8 (6.02)	96.7 (6.7)
English-Latin	98.8 (2.66)	99.6 (1.41)
English-MuR	98.3 (3.18)	99.4 (2.24)
Overall across conditions	98.1% (3.92)	98.7% (3.92)

Comprehension scores and standard deviation by condition and group

Figure 6



Comprehension scores and standard deviation by condition and group

To test for any differences between the groups' comprehension scores, the raw scores were first converted to binary correct/incorrect and then a glmer model from the binomial family in R was run, with binary scores as the dependent variable, and factors of group (two levels: monolingual, bilingual), condition (four levels: Single Talker, English-English, English-Latin and English-MuR) and their interaction, in addition to participant age, socio-economic status (SES) plus subjects as a random effect. The results showed that the only significant factor was condition [X^2 (1, N = 48) = 17.3, p = .00061] and there was no difference in comprehension by group nor condition:group interaction. Pairwise tests on participants' scores per condition, using t tests, revealed that the only condition across groups that differed from the others was Condition 2, English-English, as per Table 3 below:

Table 3

Condition A	Condition B	t	р	d	
1. Single Talker	2. English-English	3.27	<.01	.67	
1. Single Talker	3. English-Latin	.27	.79	.056	
1. Single Talker	4. English-MuR	1.06	.44	.22	
2. English-English	3. English-Latin	-3.19	<.01	65	
2. English-English	4. English-MuR	-2.62	<.05	53	
3. English-Latin	4. English-MuR	.88	.46	.18	

T tests for comprehension scores between conditions in EEG Experiment 1

In addition to the frequentist approach, comprehension scores were also analysed using a Bayesian independent sample t-test and repeated measures ANOVA in JASP (version 0.13), where the null was defined as a group effect of 0, with a Cauchy prior width set to 0.707. The Bayes factor from the t test of overall comprehension scores indicated anecdotal evidence for the null hypothesis (BF10=.378), suggesting comparable performance between the groups. A Bayesian repeated measures ANOVA with default priors and the factors of group and condition, with covariate of age, identified extreme evidence for an effect of condition (BF10=2248). In a subsequent best model assuming the effect of condition, there was extreme evidence for the null hypothesis of the factor of group (BF10=.000124). This indicates that, once the participants' variation by condition is taken into account, the Bayes factor provides extreme support for the claim that the groups performed equivalently in the comprehension task.

These results show comparable performance of monolingual and bilingual children in the behavioural comprehension task, with the performance in both groups suffering in the English-English condition relative to the other conditions. The comprehension scores for both groups in the English-English condition were lower than in any other part of the task, indicating most interference in this condition. This substantiates reports by participants in both groups that they found the interference in English the most difficult.

4.4.2 EEG Data

4.4.3 The effects of attention on speech reconstruction accuracy

In the analysis of the neural data, datapoints more than 1.5 interguartile ranges above the upper quartile or below the lower quartile were removed as outliers, excluding 170 datapoints (0.5% of the total). Visual inspection of residual plots did not reveal any obvious deviations from normality. The first analysis of the neural data aimed to test the robustness of the paradigm, by establishing whether attention modulated speech reconstruction accuracy in children. The Imer model included the three conditions where both attended and unattended narratives were presented to the participants (English-English, English-Latin and English-MuR); thus excluding the condition where there was no interference (Single Talker). The dependent variable was reconstruction accuracy score (r), and the fixed factors were group (two levels, monolingual, bilingual), attention (two levels, attended and unattended) and condition (three levels), and the interactions between them. Participant age and parental SES were also included as predictors, and subjects and items as crossed random effects. Results showed a significant effect of attention $[F(1,712.8)=46.53, p<.001, \eta^2=.06];$ a significant effect of condition $[F(2, 720.6)=59.3, p<.001, \eta^2=.06]$ p<.001, η^2 =.14] and a significant interaction between condition and attention [F(2, 710)=5.4, p<.01, η^2 =.01] as well as between condition and group [F(2, 28096.1)=7.4, p<.001, η^2 =.005]. Pairwise comparisons confirmed that the attended stream EEG data showed on average higher stimulus reconstruction accuracy than the unattended ones, with the difference between them significant overall [r_{attd}=.057, r_{unattd} =.040, t=8.79, p<.001, d=.10] and in each condition separately [English-English r_{attd}=.047, r_{unattd}=.032, t=4.21, p<.001, d=.09; English-Latin r_{attd}=.053, r_{unattd}=.026, t=8.39, p<.001, d=.17; English-MuR r_{attd}=.071, r_{unattd}=.062, t=2.38, p<.05, d=.05]. They confirm that attention improves reconstruction accuracy of spoken narratives in children, replicating similar results in the literature.

The next analysis tested whether the same general pattern holds in monolingual and bilingual groups separately. In *monolinguals*, a model including attention (two levels, attended and unattended), condition (three levels) and their interaction, participant age, parental SES, plus subjects and items as crossed random effects showed significant effects of attention [F(1, 706.03)=29.5, *p*<.001; η^2 =.04] and condition [F(2, 718.01)=55.28, *p*<.001; η^2 =.13]. In *bilinguals*, the equivalent model showed a significant effect of attention [F(1, 731.19)=29.03, *p*<.001, η^2 =.04], condition [F(2, 731.52)=21.44, *p*<.001, η^2 =.06], and their interaction [F(2, 724.01)=4.66, *p*<.01, η^2 =.01]. Pairwise comparisons confirmed that in both groups attended streams were reconstructed more accurately than unattended streams in each condition separately, other than in the English-MuR condition in bilinguals. Table 4 and Figure 7 show attended and unattended reconstruction accuracy scores, and results of t tests between the attended and unattended results, by group and condition.

Table 4

	Monolinguals				Bilinguals					
Condition	Attd	Unattd	t	Р	d	Attd	Unattd	t	р	d
Single Talker	.075					.06				
English–English	.048	.036	2.59	<.05	.08	.045	.029	3.29	<.01	.1
English-Latin	.055	.028	5.99	<.001	.17	.051	.025	5.87	<.001	.16
English-MuR	.085	.073	2.45	<.05	.07	.059	.054	.99	ns	.03
Overall across conditions	.066	.045	6.36	<.001	.11	.054	.036	6.01	<.001	.1

Reconstruction accuracy scores (r) by condition and group

Attd=attended stream, Unattd=unattended stream

Figure 7



Attended and unattended reconstruction accuracy scores (r) by condition and group

Note: Results show robust effects of attention on the reconstruction accuracy of speech envelopes, with higher reconstruction accuracy for the attended than for the unattended envelopes in both groups

4.4.4 Reconstruction accuracy of attended streams in monolinguals and bilinguals

A key question driving this research was to establish whether bilingualism modulates the neural encoding of attended speech envelopes in children; and what pattern does this modulation follow. The next set of analyses therefore asked whether monolingual and bilingual groups differ in reconstruction accuracy of attended streams across conditions. To this end a model was run that included attended condition (four levels, Single Talker, English-English, English-Latin, English-MuR), group (monolingual, bilingual) and their interaction, participant age, parental SES, plus subjects and items as crossed random effects. The results showed that the only significant predictors were condition [F(3, 483.2)=13.63, p<.001, η 2=.08] and group by condition interaction [F(3, 18283.7)=3.59, p<.05, η 2=.005].

To explore what was driving this interaction, the next analysis investigated the patterns of reconstruction across attended conditions in each group separately. In monolinguals, a model with four levels of attended condition, participant age, parental SES, and subjects and items as random effects, showed a significant effect of condition $[F(3,481.02)=15.1, p<.001, n^2=.09]$ only, with the post-hoc tests showing significantly higher encoding in the Single Talker condition than in the English-English and English-Latin conditions [t=5.49, p<.001, d=.16, and t=4.07, p<.001, d=.19 respectively], and significantly higher encoding in the English-MuR than the English-English and English-Latin conditions [t=-7.44, p<.001, d=-.22; and t=-6.04, p<.001, d=-.18 respectively]. There was a trend of stronger encoding in the Single Talker than in the English-MuR condition [t=-2.01, p=.054]; but no difference between the English-English and English-Latin conditions [t=-1.41, p=.16]. The equivalent analysis in bilinguals showed a comparable, but much reduced pattern of differences between conditions, with a significant effect of condition $[F(3,491.09)=4.03, p<.01, n^2=.02]$ reflecting weaker encoding in the English-English condition compared to the Single Talker and English-MuR conditions [t=3.05, p<.05, d=.09; and t=-2.81, p<.05, d=-.08 respectively]. No other differences emerged in the bilingual group, implying that the type of interference significantly modulated attentional encoding in monolinguals but had an attenuated effect in bilinguals, comparable to the results seen in adults (Olguin et al., 2019).

To confirm that monolinguals indeed encoded attended stream envelopes more strongly than bilinguals in some of the conditions, the next analysis directly compared the reconstruction accuracy between the groups in each attended condition separately. These pairwise comparisons for individual conditions showed significantly higher attentional encoding in monolinguals than in bilinguals in the Single Talker and English-MuR conditions [t=3.12, *p*<.01, *d*=.09; and t=5.32, *p*<.001, *d*=.16; respectively], but no difference between the groups in the two language

interference conditions [English-English t=.74, p=.46; English-Latin t=.83, p=.46]. Hence, even if the attentional encoding in the language interference conditions in bilinguals was comparable to that seen in monolinguals, the significantly weaker encoding of the Single Talker and the English-MuR conditions in this group has resulted in the overall much flatter pattern of results across conditions in bilinguals (Figure 8). In other words, the key underlying variable behind the differences between the two groups appears to be the strength of attentional encoding in conditions of weak or no interference. These differences are illustrated in Figure 8.

Figure 8



Between-group differences in reconstruction accuracy scores per condition

Note: Summary of the pattern of results for (a) attended streams, and (b) unattended streams. Error bars represent 95% CI.

4.4.5 Reconstruction accuracy of unattended streams in monolinguals and bilinguals

Following the same approach as used in the analyses of the attended streams, an equivalent model was run with unattended conditions (three levels, English-English, English-Latin, English-MuR), group (monolingual, bilingual) and their interaction, participant age, parental SES, and subjects and items as random effects. The results showed a significant main effect of condition [F(2, 355.68)=46.44, p<.001, n²=.21], but no main effect of group, and no interaction between group and condition. Further analyses confirmed that both groups showed the same pattern on differences across the three unattended conditions, with the unattended acoustic interference (MuR) showing more encoding and higher reconstruction accuracy than the two unattended languages distractors (English, Latin). For monolinguals, the results of the pairwise t-tests were t=7.64, p<.001, d=.22 for English-MuR vs English-English comparison, and t=9.85, p<.001, d=.29 for English-MuR vs English-Latin comparison. For bilinguals they were t=5.55, p<.001, d=.16 and t= 6.62, p<.001, d=.18 respectively. Neither group showed a significant difference in reconstruction accuracy for the unattended envelopes between the English-English and English-Latin conditions. The patterns between groups for unattended reconstruction accuracy scores across conditions were therefore equivalent.

Finally, the between group comparisons showed a significantly higher encoding of unattended envelopes in the English-MuR condition for monolinguals compared to bilinguals [t=4.05, p <.001, d=.11]. These results are summarised in Figure 8 (line chart above).

In sum, the results revealed that monolingual children modulated the accuracy of attended stimulus reconstruction as a function of the type of interference, with natural language distractors (English, Latin) most strongly interfering with the reconstruction of the attended stream. In contrast, bilingual children showed weaker

differentiation in the encoding of attended speech across conditions. The key factor driving these between-group differences appears to be the strength of encoding in conditions of little or no interference (Single Talker, English-MuR), with significantly stronger encoding in monolinguals than in bilinguals here. Monolingual and bilingual children showed comparable patterns of reconstruction accuracy of unattended speech.

4.5 Discussion

Building on the substantial evidence that learning and using multiple languages modulates selective attention in children (Barac et al., 2014), the current experiment investigated the mechanisms that drive this modification. Using a dichotic listening task, I assessed the patterns of responses to different types of interference in monolingual and bilingual children aged 7-12; comparing their behavioural comprehension scores and their cortical tracking of attended and unattended speech envelopes. Despite equivalent behavioural performance, clear differences emerged in the way monolinguals and bilinguals encoded attended speech, confirming that the processing demands of bilingualism shape the supporting neurocognitive architecture (Green & Abutalebi, 2013). Most important however was the observation that, instead of enhanced attentional capacity, these neuroadaptive modifications appear to reflect its redistribution, arguably aimed at economising the available resources to support optimal behavioural performance. I discuss these results in more detail below.

In terms of behavioural comprehension scores, the results clearly showed that all children performed the task equally well, and were able to process the attended stories for meaning. This aligns with the general pattern observed in dichotic listening studies that the information presented to the attended ear can usually be processed with very few errors (Cherry, 1953; Treisman, 1969). Importantly however, data showed no difference in the pattern of comprehension scores between monolingual and bilingual children, with both groups achieving high

comprehension scores across the board, but finding the English-English condition most difficult. Similar to the arguments already made in the literature (Paap et al., 2015), this finding that both groups achieved equivalent high-level performance can be taken to imply that any modification to the underlying neural mechanisms in the bilingual group could be considered as adaptation aimed at supporting such performance, made necessary by the increased processing demands of the bilingual environment.

The analysis of the neural data focused on reconstruction accuracy of attended and unattended speech envelopes from the EEG data as the index of attentional encoding. As reviewed in the Introduction, it has been well-established in both children and adults that cortical activity encodes the temporal envelope of speech, synchronising to its slow amplitude modulations (Poeppel & Assaneo, 2020; Lalor & Foxe, 2010). Selective attention robustly influences these synchronisations, with the results showing preferential tracking of the attended stream over the ignored one (Rimmele et al., 2015; Horton et al., 2014). These synchronisations between the auditory signal and the neural data were typically investigated by assessing their linear relationship using cross-correlation or forward modelling; however here a backward 'stimulus reconstruction' approach was used. This approach has been gaining increased popularity in the recent literature (Power et al., 2016; Kalashnikova et al., 2018; Jessen et al., 2019; Attaheri et al., 2020) as it offers advantages such as providing increased sensitivity to signal differences between highly correlated EEG channels (Crosse et al., 2016).

Consistent with the existing evidence (Rimmele et al., 2015; Horton et al., 2013; Ding & Simon, 2012) the results in this experiment showed a robust effect of attention, with higher reconstruction accuracy scores consistently seen for the attended than for the unattended envelopes in both groups. Given that reconstruction scores reflect how much stimulus-relevant information is encoded in the EEG signal and how well this can be modelled, these results imply that attended streams were encoded more strongly than the unattended streams. Also consistent with the existing data (Olguin et al., 2018; Brungart & Simpson, 2007; Hawley et al., 2004),

there was evidence that the type of interference influenced attentional processing; with language distractors (English and Latin) reducing reconstruction accuracy of the attended envelopes more strongly than the less interfering distractors (Single Talker and English-MuR conditions). This is arguably because attentional selection between competing streams of information can be achieved either on the basis of lower-level sensory differences between them, or based on higher-level syntactic and semantic information - with the latter argued to occur later and require more processing capacity (Johnston & Heinz, 1978; Bronkhorst, 2015). The separation between the two streams in the language distractor conditions is more likely to require this latter type of processing; more robustly impacting on the attentional capacity available for the processing of attended stream in these conditions. Alternatively, this pattern of results might be explained in terms of increased difficulty of auditory object formation and selection in the language distractor conditions (Shinn-Cunningham, 2008); where the similarity between the attended and the unattended streams might cause them to be perceived as a unified auditory object, thus resulting in poorer sensitivity to the content of the attended target stream.

In relation to the account of early vs. late selective processing, these results support the multimode theory of attention (Johnston & Heinz, 1978), an approach that integrates the early and late accounts and incorporates capacity constraints. This theory proposes that target information is selected at the level of sensory analysis by early modes of attention but not until after semantic analysis by late modes. Thus along the continuum of different attention modes from early mode to late mode, attended information and unattended information can be differentiated at different depths, consistent with the results in this thesis. Experiment 1 indicated that the weakest encoding of the attended stream occurred in the English-English condition. In this condition, the children were presented with two simultaneous stories that could not be dissociated based on intelligibility or some low-level sensory differences (as was the case in the other conditions). The low reconstruction accuracy scores indicated that in this condition both attended and unattended inputs were processed equivalently and arguably only dissociated after they had undergone

some degree of semantic encoding and analysis. Therefore, in the English-English condition the unattended semantic information was dissociated from the attended information further up the processing stream compared to the conditions with nonlinguistic interference, which used up more attentional capacity and left fewer resources for encoding the attended stream. This is consistent with the multimode theory of selective attention, which states that semantic processing happens later up the selective attention stream and requires more effort (Johnston & Heinz, 1978). The English-English condition therefore also created the greatest interference, consistent with reports that the participants found it the most difficult condition.

The key finding of this study, however, was that the attentional encoding across conditions differed between the monolingual and the bilingual children. In the monolingual group, results revealed a prominent contrast between the conditions with low or no interference and the language interference conditions; yet this effect was markedly attenuated in the bilingual group (Figure 8). The differential patterns of encoding in monolingual and bilingual listeners observed here replicates the results found in adults (Olguin et al., 2019), adding further support to the hypothesis that bilingualism modifies the neural mechanisms of selective attention across the lifespan (Comishen et al., 2019; Krizman et al., 2014; Garbin et al., 2010). At the start of this chapter, two accounts were presented that might explain the possible mechanisms of this modification. The first was that the need for constant management and inhibition of competing languages in bilinguals enhances their capacity for selective attention, resulting in better performance and increased attentional control (Bialystok, 2015). The second was that these demands of selection and inhibition will themselves utilise some of the existing attentional resources, which might impact on the available attentional capacity and require that the remaining resources are optimised in order to achieve full task performance. The results in this experiment showed no evidence for the enhanced attentional capacity, behaviourally or neurally, in the bilingual group. In contrast there was statistically significant weaker neural encoding in bilinguals overall for both attended conditions (r_{attd}=.054 for bilinguals vs r_{attd}=.066 for monolinguals; t(18660)=-4.81,

p<.001,d=-.070) and unattended conditions (r_{unattd} =.036 for bilinguals vs r_{unattd} =.045 for monolinguals; t(14784)=3.43, p<.001, d=-.056), and significantly weaker reconstruction in attended conditions of low or no interference in bilingual compared to monolingual children, lending support to the second proposition.

The observed indication of reduced cortical encoding overall in bilinguals is not without a precedent, with examples of reduced neural activity during selective attention tasks most commonly seen in the cortical areas associated with conflict processing. For instance, functional imaging during a Flanker task performed by bilinguals and monolinguals (Abutalebi et al., 2012) revealed significantly lower patterns of activation in the anterior cingulate cortex (ACC) for bilinguals, leading the authors to conclude that 'bilinguals...resolve cognitive conflicts with less neural resource'. A similar fMRI study of a Stroop-like switching task (Garbin et al., 2010) also found that monolinguals activated the ACC during the task, whereas bilinguals did not. An ERP study tracking bilingual and monolinguals' neural responses during a variety of selective attention tasks (Kousaie & Phillips, 2012), predicted superior performance (greater accuracy and faster reaction times) and larger N2 amplitudes for bilinguals relative to monolinguals. On the contrary, behaviour was equivalent between the two groups; and the monolingual group exhibited larger N2 amplitudes than the bilingual group during the Stroop task. The Simon task also elicited the 'unexpected and surprising' result that monolinguals demonstrated larger P3 amplitudes than bilinguals. Furthermore, higher ERN amplitudes for bilinguals than monolinguals in the final Flanker task, which would usually be interpreted as evidence of enhanced cognitive control, were due to a longer tail for incongruent trials, indicating a prolonged post-response conflict and slower recovery for bilinguals in these trials. Taken together, this evidence supports the hypothesis that different configurations of the underlying neurofunctional architecture can support equivalent behavioural performance, with these different configurations reflecting different processing demands presented to the system over time. This functional plasticity, also known as degeneracy in the scientific literature (Green et al., 2006; Mason et al., 2015; Navarro-Torres et al., 2021) is a common feature in biological systems, allowing flexible adaptation to changing environments. Hence, while the

findings reveal that the management of competing languages draws on attentional resources in bilingual children, they do not show any adverse effects on performance – the outcome is primarily indicative of the modifications to the underlying processing networks that are aimed at supporting performance. In fact, as mentioned in the Introduction, these results could be interpreted as showing increased flexibility in the usage of the available resources in bilingual children, enabling them to do 'more with less'.

Turning to the more specific pattern of reduced cortical tracking of the attended speech envelope in bilinguals observed, where this was most prominent in the Single Talker and English–MuR conditions – the two conditions with weakest interference, and thus requiring least effort to comprehend the attended steam. Figure 9 demonstrates this by combining the low-interference (Single Talker and MuR) conditions and high-interference (English and Latin) and then plotting the groups side by side. The monolingual and bilingual participants are comparable in the highinterference conditions, but there is a clear difference between the groups in the low-interference conditions, especially when the monolinguals' low-interference is plotted (the dashed line) as a baseline reference.

Figure 9



Differences in combined low-interference and high-interference conditions

One hypothesis is that this directly results from the need to economise the available attentional capacity in order to support optimal behavioural performance. To understand this, it is again necessary to recall that behavioural comprehension scores were equivalent between the groups for all conditions. Yet, achieving optimal behavioural performance is not equally demanding across different conditions, and can arguably be more easily accomplished with reduced attentional resources in the conditions that are less taxing for the processing system. It can therefore be assumed that this reduction in cortical tracking in the conditions of weak or no interference in bilinguals arises because it can be most easily accommodated while still retaining full behavioural performance. In contrast, reductions of attentional encoding in conditions with stronger interference (English-English and English-Latin) would likely lead to diminished performance compared to the monolingual group.

Whilst tentative, this interpretation aligns with evidence from research into the mechanisms of adaptive neural plasticity, which suggest that 'experiences contributing to mastery over environmental challenges modulate neural responses in ways that enhance optimal performance' (Lambert et al., 2019).

The final set of findings to address concerns the pattern of reconstruction accuracy scores seen for the unattended streams. Here results showed that, in both groups, the unattended MuR stream was significantly better reconstructed than the unattended Latin and English stories. In addition, the MuR encoding was stronger in the monolingual than in the bilingual group. Both of these findings might be explained by the same mechanisms discussed above, with the selection between competing streams being less demanding for the MuR distractor and for monolinguals, thus impacting least on processing capacity available for encoding. However, it is more likely that the strong MuR encoding reflects the fact that the unattended MuR envelopes used in the experiment were generated from the same narratives that the participants were presented with as target stories in their attended ear. Given that the MuR envelope largely preserves the spatio-temporal features of the source utterance, it is unsurprising that there is a high degree of similarity between the envelope reconstruction scores for attended and unattended steams in the English-MuR condition. Despite this, the results showed that the attended steam was more strongly encoded than the unattended steams (significantly so in the monolingual group), adding further evidence that attention significantly influences the neural encoding of speech envelope (Ding & Simon, 2012).

In sum, EEG Experiment 1 investigated the mechanisms underlying the modification of selective attention in bilingual children. The data showed no evidence for the enhanced attentional capacity in the bilingual group. Instead, equivalent behavioural performance was observed, coupled with a modified pattern of neural encoding that was most prominent in conditions of weak or no interference. These data are interpreted as showing that the available resources are economised to support optimal behavioural performance; potentially suggesting increased flexibility of their

usage in response to the demands of bilingual language processing. Subsequent experiments in the next chapters explore the limitations of this adaptation of attentional resource; and investigate to what extent these findings generalise beyond the language domain by using non-linguistic distractors.

Chapter 5: Experiment 2

5.1 Introduction

The previous chapter described a study in which bilingual children's reconstruction scores, an index of attention, were generally lower and flatter than those of monolingual children. Their behavioural comprehension scores were equivalent, however. This supports three of the suggested consequences of bilingualism on selective attention discussed in the Introduction. First, that bilingualism triggers modifications to the neural correlates of selective attention, evidenced in the differential scores and patterns of reconstruction accuracy between monolinguals and bilinguals performing the dichotic listening task, and consistent with evidence that bilingualism shapes brain structure and processing across the lifespan (in childhood: Jasińska & Petitto, 2013; Archila-Suerte et al., 2018; Pliatsikas et al., 2020; into adulthood: Bialystok et al., 2005; Mechelli et al., 2004; Filippi et al., 2011 and old age: Luk et al., 2011; Abutalebi et al., 2015; Frutos-Lucas et al., 2020).

The second conclusion is that these neural adaptations emerge in order to support optimal performance, as seen in the equivalent behaviour (comprehension scores) in both groups. This result replicates other studies which have observed neural differences between monolinguals and bilinguals even when the groups display equivalent behavioural performance (Bialystok et al., 2005; Kousaie & Phillips, 2012; Olguin et al., 2019), lending support to the hypothesis that bilingualism creates a degenerate system (Green et al., 2006; Mason et al., 2015; Navarro-Torres et al., 2021), in which differences in functional structure can lead to the same output.

Furthermore, the pattern of neural responses in bilinguals in the EEG study specifically revealed lower indices of attention. This suggests that the third conclusion to be drawn is that the processing demands of bilingualism draw on some of the attentional resources (Wickens, 2007) of an attentional processor of finite capacity (Kahneman, 1973; Broadbent, 1965; Clark & Dukas, 2003). This results in lower indices of attention but, as seen in Chapter 4, does not have a negative impact on performance.

This supports the view that typically attentional systems are flexible enough to accommodate parallel processing demands and maintain performance as long as total capacity is not exceeded (Moray, 1967). Thus the increased processing load from an additional language would decrease overall capacity (Dornic, 1980), but, in the context of a primary task, still leave enough bandwidth for optimal performance. This would explain why, as shown in Chapter 4, bilinguals accomplish single behavioural tasks with no apparent loss of performance relative to monolinguals. Indeed, an alternative interpretation of the neurocognitive adaptations of selective attention in bilingualism might be that they emerge precisely in order to enable bilinguals to overcome the consumption of attentional resources and achieve and maintain optimal (equivalent) behavioural performance in the context of standard selective attention tasks.

This raises the following question: what are the limitations of this bandwidth in the context of bilingualism and selective attention tasks? One way of investigating this is to increase the attentional load in a dual-task paradigm. Research shows that any attentional system, despite being flexible enough to accommodate parallel processing when attentional demands are not excessive (Hunt et al., 1989; Moray, 1967), has limitations on the processing of simultaneous tasks (Gopher, 1986). Multiple tasks make demands on different mechanisms, each with its own capacity (Navon & Gopher, 1979), leading to a processing pipeline of prioritization of competing demands (Janssen & Brumby, 2010) and allocation of attentional resources to overcome the resultant attentional bottleneck (Pashler, 1994; Schubert, 1999), as described in the multiple resource theory framework (Wickens, 2008).

Studies into the impact of dual tasks on performance can reveal fundamental aspects on the mechanisms and limitations of attentional processing. Contrary to

many examples of "multitasking" in daily life (such as driving and maintaining a conversation) being achieved without noticeable deterioration of performance, researchers have found that the dual task paradigm in an experimental setting consistently results in a psychological refractory period (PRP), the delay generated when performing two tasks simultaneously (Telford, 1931; Vince, 1949; Welford, 1952; Pashler & Johnston, 1998; Osman & Moore, 1993; Schubert, 2008). This delay is again attributed to the need to execute both tasks' demands with a limited-capacity processor (Kahnemann, 1973), which can only process a restricted amount of information at any given point (Broadbent, 1965; Clark & Dukas, 2003). Parallels have been drawn between the allocation of attentional resources in dual tasks and models of information processing, resulting in models such as multiple resource theory (Wickens, 2008), which provide a theoretical framework of the prioritisation of tasks.

Thus the aim of the current study was to test the impact of adding a secondary task to a primary selective attention task, specifically on the performance of bilinguals who were assumed to have more constrained attentional capacity. In spite of the long-standing premise that typical performance is limited under competing task demands (Treisman, 1969; Mowbray, 1954), the interaction between these limitations and bilingualism has been less clear cut. On the one hand, there have been studies which show a protective effect of bilingualism, through reduced deterioration relative to monolinguals in performance, when managing competing tasks (Janic et al., 2020; Telner et al., 2008) compared to baseline performance in a single (control) task. These studies used a visual search (Janic et al., 2020) and driving simulation (Telner et al., 2008) as the primary tasks and counting backwards or verbal tasks as the secondary or distractor task. Another combination of tasks, memory recall and switching (Sörman et al., 2017), has suggested a more nuanced effect of bilingualism. Using this paradigm, a longitudinal study investigating the dual-task costs of encoding and retrieval of a list of words while performing a simultaneous card sorting task found lower costs for bilinguals in one condition only (encoding + retrieval + card-sorting) and this finding was not consistent across age groups.

A series of complementary behavioural studies (Bialystok et al., 2006; Bialystok, 2011) have investigated the effect of dual tasks in different modalities for monolingual and bilingual groups, using semantic categorisation tasks across visual and auditory domains. In the first (Bialystok et al., 2006), two semantic categorisations were required (of letters/numbers or animals/musical instruments) for both visual and auditory stimuli. Both age groups (younger and older adults) showed a bilingual advantage in one section only of the visual task; but not for an alternative version of the visual task, nor in the auditory task. In the second study (Bialystok, 2011), two groups of 8 year olds performed one of the semantic categorisation tasks in two modalities simultaneously and the only difference between the groups was the accuracy, but not response time, in the categorisation of the visual stimuli. The above point to a slight bilingual advantage in the visual domain for dual tasks across modalities.

However, a separate study (Fernandes et al., 2007) of younger and older adults, combining memory recall of spoken words with a visual distraction task using semantically similar or unconnected words to the primary task, found that bilingual participants in both groups performed significantly worse in all the conditions (no distraction and distraction with both connected and unconnected words). The authors attributed this bilingual limitation to the task's dependency on lexical access, which might advantage monolinguals who have a larger vocabulary in their primary (only) language.

Given the inconsistent results of previous studies of the effect of bilingualism on behavioural performance of competing tasks, several precautions were taken when designing the dual task for the study in this chapter. Firstly, the tasks selected for this study were not explicitly dependent on lexical access/semantic retrieval/vocabulary. Secondly, their combined attentional load varied throughout progressive rounds in order to isolate and identify attentional resource constraints.

Thirdly, tasks that were used were across different modalities, in order to minimise perceptual masking and resource similarity for each of the dual tasks (Rollins & Hendricks, 1980). The neural mechanisms for selective attention tasks in different modalities are clearly dissociated in imaging studies (Shomstein & Yantis, 2004), as would be expected. Furthermore, the neural patterns differ between single selective tasks in either modality and the areas recruited to perform in a divided bimodal task (J. A. Johnson & Zatorre, 2006), which, in addition to recruiting the associated sensory cortical areas, also uses an area of the prefrontal cortex. This suggests the recruitment of additional resource to prioritise and juggle the demands of the two competing tasks in different domains, and is consistent with accounts that a limited-capacity system draws on central – as well as task specific - control processes to manage competing demands (Lien et al., 2008).

Finally, to prevent motor response interference, the response mechanisms also differed by task (timed motor response throughout the visual task; untimed multiple choice selection at the end of the auditory task).

The primary task was an edited version of the dichotic listening task used in the EEG experiment, which again featured varying levels of interference (Olguin et al., 2018, Phelps et al, 2022) to be inhibited in order to attend to the target story. This was presented in combination with a simple visual distractor task: a version of the T.O.V.A. (Test of Variables of Attention) task, a computerised fixed-interval continuous performance test that assesses inhibition and vigilance (Greenberg & Waldmant, 1993). Details of the T.O.V.A. are presented in the Methods chapter.

Of primary interest was the prioritisation of the tasks and subsequent performance for different groups of monolingual and bilingual participants. With this in mind, the dual task was deliberately designed so that the listening task would take priority over the visual task in conditions of divided attention. Interest was sustained within each round of the listening task by using stories which were unpredictable and entertaining. Additionally, each round used a different narrative as the target story, and presented different levels of interference in the non-attending ear, adding

variety and a range of difficulty across the whole task. In contrast, the visual (T.O.V.A.) task was repetitive and simple within each round; and the same stimuli were presented in a random order for each of the conditions. Finally, in the dichotic listening task, participants were instructed explicitly to listen carefully in order to answer comprehension questions at the end of each round; in each of the visual tasks there was neither feedback nor performance assessment.

The prediction was that the demands of this dual task would reveal any effects of second language inhibition on overall attentional capacity, by showing the relative performance and prioritisation of the visual and auditory tasks in all groups and across different levels of difficulty/attentional load. The study was replicated across three groups: two age groups (children and adults), and with the adult group split by language experience. The motivation for recruiting different age groups was twofold: firstly, to confirm whether the inconsistent effect across age groups using a dual-task paradigm seen in previous studies would be replicated (Sörman et al., 2017); and secondly to see if the lack of differentiation between monolingual and bilingual adults' performance noted in a single-task paradigm, attributed to ceiling cognitive capabilities (Bialystok et al., 2005; Valian, 2015b), would also emerge in a dual-task design. The variable of language experience on adults' responses was included after noting inconsistencies in the literature on bilinguals' performance in selective attention tasks dependent on language exposure and age of acquisition (Prior & Gollan, 2011; Luk, Sa, et al., 2011; Pelham & Abrams, 2014; Shi, 2010; Mayo et al., 1997; Swain & Cummins, 1979; Linck et al., 2008). Thus, the dual-task paradigm was used on a sample of school-age children and on young adults with different bilingual profiles, and results compared.

5.2 Materials and method

The experiment was conducted on three groups of participants: monolingual and bilingual children aged 7-12 (Study 1) and adults aged 18-45 with differing second language exposure (Studies 2 and 3). All studies were approved by the Cambridge

Psychology Research Ethics Committee, and performed in accordance with relevant guidelines and regulations.

5.2.1 Study 1 Participants

80 typically-developing children aged 7-12 were tested. They were split into two categories: bilingual (n=40, twenty two males, mean age 9.7 yrs, SD 1.39) and monolingual (n=40, twenty five males, mean age 10.4 yrs, SD 1.64). The recruitment process checked that all participants were healthy with no history of hearing problems or neurological disorder. All participants' parents completed a language history questionnaire, which provided an overview of children's exposure to languages. As confirmed by the questionnaire, all monolingual participants were native speakers of English, with no significant exposure to other languages. The participants in the bilingual group all had a similar profile: they all had one or more parents whose first language was not English, and they used this language at home on a daily basis. The average age of acquisition for the language other than English was 0.1 years (age 0 for 38 participants, age 2 for two participants). All children were also fluent in English, being resident in the UK, and following English-speaking curriculum at school. The average age of acquisition for English was 0.3 years (from birth for 32 participants, age 1 for five participants, age 2 for two participants, age 3 for one participant). The languages spoken by the bilinguals in addition to English were Creole, French, German, Greek, Gujarati, Hungarian, Japanese, Italian, Portuguese, Russian, Spanish and Swedish. Additionally, six children spoke a third language proficiently (French, Japanese, Italian and Urdu). Children were recruited via their parents, who responded to posters on social media, and word of mouth. The monolingual children were marginally older on average than the bilingual group [10.4 vs 9.7 yr respectively, t(76)=2.06, p=.043], and age was included as a predictor in all analyses. Parental education information was collected as an indication of SES, a well-documented influence on selective attention in children (Stevens et al., 2009). The majority of participants' parents (79.2%) were educated to degree level or higher, and the groups were not significantly different following a Mann-Whitney U

test on this approximation of SES [bilinguals M=2.34, SD=.66; monolinguals M=2.09, SD=.69; W=608.5, p = .085].

5.2.2 Study 2 Participants

89 adults were recruited via the online platform Prolific (https://www.prolific.co/) and tested, of whom five were rejected after initial quality checks (see details of rejection criteria in Analyses section). This left 84 adults whose data were included in Study 2. They were split into a bilingual group (n=42, six males, mean age 27.4 yrs, SD 5.95) and monolingual (n=42, fifteen males, mean age 28.9, SD 7.29). All participants were neurotypical with no history of hearing problems or neurological disorder. Participants completed a language questionnaire on age of acquisition, proficiency and daily usage of their languages. All monolingual participants were recruited as native speakers of English with no fluency in other languages. All bilingual participants were residents in the UK; fluent and highly proficient English speakers (average English proficiency = 9.3/10, SD .96); and using English on a daily basis (average English use = 73%, SD 24.9), but with a language other than English as their native language. The native languages spoken by the bilingual participants were Afrikaans, Bengali, Danish, Farsi, Filipino, Finnish, French, Galician, Greek, Gujarati, Hindi, Igbo, Indonesian, Italian, Konkani, Latvian, Lithuanian, Malayalam, Persian, Polish, Portuguese, Romanian, Russian, Scots, Spanish, Turkish and Welsh. As well as English as a second language, the majority of participants spoke a third language proficiently (Albanian, Catalan, Dutch, French, German, Hindi, Icelandic, Italian, Japanese, Russian, Russian Sign Language, Spanish, Tamil, Telegu, Urdu and Xhosa), several a fourth (Annang, French, Italian, Polish, Portuguese, Spanish, Swedish) and one a fifth (Japanese). The monolingual and bilingual participants were matched on age [t(78.9)=1.01, p=.32, d=.22]. Highest level of education was collected as an indication of SES, and the groups were comparable on this approximation [monolinguals M=1.5, SD=.59, bilinguals M=1.81, SD=.74; W=685, p=.053].
5.2.3 Study 3 Participants

60 adults were recruited via the online platform Prolific (https://www.prolific.co/) and tested, of whom two (one monolingual and one bilingual) were rejected after initial quality checks (see details of rejection criteria in Analyses section). This left 58 adults whose data were included in Study 3. They were split into a bilingual group (n=29, eighteen males, mean age 28.5 yrs, SD 7.72) and monolingual (n=29, fourteen males, mean age 28.9, SD 7.29), who were matched on age [t(54.9)=1.31, p=.194, d=.345]. Highest level of education was collected as an indication of SES, and the groups were comparable on this approximation [bilinguals M=1.77, SD=.612, monolinguals M=2.04, SD=.649; W=359.5, p=.157]. All participants were neurotypical with no history of hearing problems or neurological disorder. Participants completed a language questionnaire on age of acquisition, proficiency and daily usage of their languages. All monolingual participants were recruited as native speakers of English with no fluency in other languages. All bilingual participants were screened as bilingual and proficient in English as well as another language, but there was no specification that they were to be residents in the UK. As the experiment was an online study that could be completed anywhere in the world, this less stringent set of criteria for the bilingual participants resulted in more variation in language background and daily exposure. English was used less frequently on average for the bilinguals in Study 3 than in the bilingual group recruited for Study 2 (Study 3: selfreported mean English use 33.6%, SD 26.9) but they were all still proficient English speakers (self-reported mean for English proficiency was 8.66/10, SD 1.28). The native languages spoken by the bilingual participants were Estonian, Italian, Polish, Portuguese, Slovenian, Spanish and Swedish. As well as English as a second language, the majority of participants spoke a third language proficiently (French, Galician, German, Italian, Japanese, Polish, Spanish and Turkish), and six spoke a fourth language (Chinese, French and Spanish). Due to the substantial difference in second language use, the bilinguals in adult groups from now on will be referred to as high-exposure for Study 2 and low-exposure for Study 3.

109

5.2.4 Design

In this dual task paradigm, participants were required to perform tasks simultaneously in two modalities: auditory and visual (Figure 10). The visual task demanded pressing the space bar in response to a pre-specified target image on the screen. The auditory task employed a dichotic listening paradigm (an edited version of the one used in Experiment 1), where participants were asked to concentrate on a story in English played in one ear, while ignoring a distractor steam presented in the other ear. The distractor stream was manipulated to create different levels of interference, from purely acoustic to linguistic. This created a total of five conditions, which are summarised in Table 5.

Figure 10

Dual task procedure



Note. Participants were instructed to attend to a story in one ear and ignore a distractor stream in the other ear, whilst also responding to a visual task. The distractor streams were manipulated to create different levels of interference. In the visual task, a picture of a dog at the top of the screen was the target, while a picture of a dog at the bottom of the screen was the distractor. Visual stimuli were presented in a target:non-target ratio of 1:3. After each block of the audio-visual task participants answered 10 comprehension questions about the target story.

In the first condition ('Control'), the participant only performed the visual task. From Condition 2 onwards, the participants were also presented with the dichotic listening task, and asked to attend to a target story in English in one ear. Condition 2 had no interference in the other ear ('Single Talker'). In the remaining three conditions participants also heard a distractor in the other ear, which they were instructed to ignore. In Condition 3 the distractor was a different story in English ('English-English'). The distractor in Condition 4 was a story in a language unknown to them ('English-Latin'). The distractor in Condition 5 was non-linguistic acoustic interference ('English-Musical Rain').

Table 5

Experimental conditions

Condition	Visual task	Auditory task	
		Attended stream	Interference
1. Control	sustained attention	None	None
2. Single Talker	sustained attention	English story 1, block 1	None
3. English-English	sustained attention	English story 1, block 2	Different story in English
4. English-Latin	sustained attention	English story 2, block 1	Story in unknown language (Latin)
5. English-MuR	sustained attention	English story 2, block 2	Acoustic interference (Musical Rain)

The target stories for the attended ear were two children's stories in English, taken from online resource storynory.com. Both stories were transcribed into 120 sentences each, with each sentence lasting approximately 3 seconds in length. The sound files were then uploaded to the experiment on Psytoolkit. Each target story was then split into 2 blocks and participants attended to the first half in the left ear with any interference in the right ear, and then the target and interference in the right and left ears respectively for Block 2. Each block (half of a story) consisted of 60 sentences, with all 60 sentences concatenated with a 300ms gap between them to create a single block lasting 3.3 minutes. Latin was chosen as the interference in Condition 4 as a non-artificial language which would almost certainly be unknown to the participants. Gender of the speaker was kept the same for all stories (same female voice for all target stories, different female voice for English interference), to reduce segregation strategies based on talker's gender (Brungart & Simpson, 2007). All stories' volumes were normalised to ensure equivalent average amplitude. The non-linguistic interference of Musical Rain in Condition 5 was identical in length, root mean squared level and long-term spectrotemporal distribution of energy to another source story, but did not trigger a speech percept (Uppenkamp et al., 2006). It was generated in MATLAB by extracting temporal envelopes of the target sentences and filling them with 10ms fragments of synthesised vowels jittered in frequency and periodicity. Instructions were recorded by the same female speaker of the target stories. These were played before each block in the target ear, telling the participant: *"This is your right/left ear. Please listen carefully to the story in this ear, on your right/left side, and ignore the story or sound in the other ear".*

After each block had finished, a screen appeared with ten comprehension questions in multiple-choice format. Comprehension questions consisted of simple written sentences to check understanding of each story (for example: *'This story is about a BEGGAR/SHEPHERD'*), and participants selected which option they thought was correct by clicking a box next to the response. Participants did not receive feedback on their responses. After completing the comprehension questions, a screen appeared introducing the next condition.

Visual task: The task required pressing the SPACE bar as quickly as possible when the cartoon picture of a dog appeared at the top of the screen. If the dog appeared at the bottom of the screen, the participant was instructed to ignore and wait for the next presentation. Each dog image was on the screen for a maximum of 2000ms. If the SPACE bar was pressed correctly, the reaction time was recorded and the next trial began immediately. If the participant failed to press the SPACE bar when the dog appeared at the top of the screen (error of omission), the trial was recorded as a target error. If the participant incorrectly pressed the SPACE bar when the dog was at the bottom of the screen (error of commission), the reaction time was recorded, the trial was coded as a distractor error, and the next stimulus appeared immediately. The ratio of target:non-target stimulus was coded to be 1:3 with the 24 targets and 72 non-targets appearing in a random order in each round. These 96 visual stimuli, with correct responses, aligned with the 200s length of the auditory stimulus per block. If, however, a participant completed the visual task before the end of the auditory stream, the experiment presented extra stimuli in the same 1:3 ratio until the story and condition ended.

There was a brief explanation of the task and practice session (which could be repeated, if the participant needed), then the first condition began while there was silence in the headphones. From the second condition onwards, the visual task was repeated in exactly the same way, while the auditory task was also presented.

Participants' responses from the visual task and comprehension answers were collated in Psytoolkit and downloaded for analysis.

5.2.5 Study 1 (children): Procedure

Prior to the testing session, parents and children were given detailed information on the aims of the project and what to expect from the session. Once they had agreed to participate, parents were sent a link to the Psytoolkit landing page for the study. This page contained another short explanation of the experiment and a sound check for the headphones to ensure they were connected to the computer audio and at an acceptable volume. Parents then completed a consent form, confirming informed consent. Parents also completed a short background questionnaire on the child, indicating age, language history and approximation of SES (parental highest educational level). Then the child watched a short video explaining the rules of both tasks in child-appropriate language. The visual "looking" task was called "Spot the Dog" and the auditory task was renamed "the listening task". Before starting the experiment, children were reminded that they could withdraw from the study at any time. If they were happy to proceed, the experiment started with the press of a key. The children had a practice session of performing the visual task, with indications of correct and incorrect responses, and the opportunity to repeat the practice as many times as needed to familiarise themselves with the task. They were then invited to start the first condition (visual only) with the press of a key. After each condition, there was a "Well done!" page explaining that the round (condition) was complete, and an instruction to press a key to resume, allowing children to take a break if they so wished between tasks. The key press led to a new set of instructions explaining the "rules" of the next condition. For example, the instruction before Condition 4 stated: "In this fourth round, you will listen to the story in your left ear while continuing the experiment. Keep playing 'Spot the Dog' and listen closely to the story in your LEFT ear. Ignore any distracting sounds in your right ear."

After finishing Condition 5, children were given a completion code to show their parent. The parent emailed the code to the researcher who issued a £5 Amazon e-voucher as compensation. The study duration from the landing page to the end of Condition 5 took approximately 30-45 minutes.

5.2.6 Studies 2 and 3 (adults): Procedure

In both adult studies, eligible participants were given access to an introductory page on Prolific, which explained the aims and design of the study. A link took them to landing page for the study on Psytoolkit. Here, they read more detailed instructions, performed a headphone audio check, and ticked an informed consent form. They then completed the background questionnaire, asking details of sex, age, educational background and details of language proficiency.

Participants had a practice session of performing the visual task, with indications of correct and incorrect answers, and the opportunity to repeat the practice as many times as needed to familiarise themselves with the task. They were then invited to start the first condition (visual only) with the press of a key. After each condition, there was a page explaining that the round (condition) was complete, and an

instruction to press a key to resume, allowing participants to take a break if they so wished between tasks. The key press led to a new set of instructions explaining the "rules" of the next condition. For example, the instruction before Condition 4 stated: "In this fourth round, you will listen to the story in your left ear while continuing the experiment."

After finishing Condition 5, participants were redirected to Prolific to claim compensation of £3.50. The study duration from the landing page to the end of Condition 5 took approximately 25 minutes. The adults' average duration was shorter than the children's because they did not watch an explanatory video or have a pause during the handover from parent to child between the explanatory landing page with consent form and background questionnaire (completed by the parent), and the experiment (performed by the child).

5.3 All studies: Analyses

5.3.1 Dichotic listening task

At the end of each block participants answered 10 comprehension questions about the attended narrative, where they had to choose between true and false options (e.g., 'This story is about a QUEEN/KING'). Correct answers were coded as 1 and errors as 0, and their probability across groups and conditions was modelled using a glmer function from the binomial family in R (Boeck et al., 2011).

5.3.2 Visual task

Reaction times and error rates were recorded. As an initial quality check, all outlier reaction times under 200ms and over 1500ms were eliminated. The remaining correct target reaction times were then log transformed to eliminate skew and datapoints more than 1.5 interquartile ranges above the upper quartile or below the lower quartile were removed as outliers. This resulted in exclusion of 129/8860 datapoints (1.5% of the total) in Study 1 (children), 302/9724 datapoints (3.1% of the total) in Study 2 (high-exposure bilingual adults) and 237/6742 datapoints (3.52% of the total) in Study 3 (low-exposure bilingual adults). Error rates were calculated per participant, including both missing the target (not pressing the space bar when the target was presented) and incorrect detection of targets (pressing the space bar when the target was not presented), thus all 40000 trials were included. The initial quality check of total error rates did not identify any children in Study 1 whose error score exceeded the normal range for attention task in that age group [the maximum error rate identified was 10%; normal range in neurotypical children aged 6-13 performing the task is 0.8%-14% according to Greenberg & Waldmant (1993)], hence the data of all 80 children were included in the analyses. In Study 2 (high-exposure bilingual adults), a check of total error rate identified two monolingual and two bilingual participants whose total error rates significantly exceeded 3SD of the mean; likely indicating that they did not attend to the task as instructed. One further monolingual participant's language questionnaire revealed proficiency in a second language, acquired in adulthood. The data of these five participants were excluded, leaving 84 participants' data for further analysis. In Study 3 (low-exposure bilinguals), a check of the total error rate identified two participants (one monolingual and one bilingual) whose error rates were significantly higher than 3SD of the mean and they were excluded, leaving 58 participants' data for analysis.

The reaction times of correct target responses, total error rates (omission and commission) and comprehension scores were then compared across groups and conditions using linear mixed-effect models (Baayen et al., 2008) as implemented in the Ime4 R package (Bates et al., 2014). To arrive at the best-fitting model, the step function in the ImerTest package (Kuznetsova et al., 2017) was applied. The Satterthwaite approximation (Satterthwaite, 1946) was used for degrees of freedom. Significant p-values are reported at p<.05. All post-hoc tests were FDR corrected for multiple comparisons.

116

5.4 Results

5.4.1 Study 1 (children)

Auditory task: Children from both groups performed the task equally well, with overall comprehension scores of 97% in the monolingual group, and 98.6% in the bilingual group. A model was used to test whether the probability of making an error differed across groups and conditions, with categorical response accuracy as the dependent variable, fixed effects of group (two levels: monolingual, bilingual), condition (four levels: Single Talker, English-Latin, English-English and English-MuR) and their interaction, participant age and parental socio-economic status (SES); and participants as a random intercept. The results showed that age was a significant predictor [X^2 (1, N = 80) = 13, p = .00031] and condition showed significance overall [X^2 (3, N = 80) = 9.46, p = .024] but there was no difference in comprehension by group nor condition: group interaction (all p>.1), confirming that both groups performed the auditory task equally well. Pairwise tests on all participants' scores per condition, using t tests, revealed no significant differences between individual conditions.

In addition to the frequentist approach, comprehension scores were also analysed using a Bayesian independent sample t-test and repeated measures ANOVA in JASP (version 0.13), where the null was defined as a group effect of 0, with a Cauchy prior width set to 0.707. The Bayes factor showed anecdotal evidence for the null hypothesis (no group effect) in an independent samples t-test (BF_{10} =.601). A subsequent Bayesian repeated measures ANOVA with default priors, incorporating group, condition and age as a covariate, suggested there was very strong evidence for the null hypothesis in the factor of group (BF_{10} =.03). This indicates that, once the factors of condition, age and group are taken into account, the Bayes factor provides very strong support for the claim that the groups performed equivalently in the comprehension task.

A summary of comprehension scores and standard deviations for children is shown in Table 6 and Figure 11.

Table 6

Study 1 (Children): Percentage correct comprehension scores by group and condition

Condition	Monolinguals	Bilinguals
1. Control	n/a	n/a
2. Single Talker	98.5 (5.3)	98.8 (4.0)
3. English-English	96.5 (7.7)	97.2 (5.9)
4. English-Latin	98 (7.2)	99.5 (2.2)
5. English-MuR	95 (10.4)	98.8 (4.0)

Figure 11

Study 1 (Children): Percentage correct comprehension scores by group and condition



Results from the visual task:

The first analysis of response times aimed to test if performance varied by group or across different attentional loads of the auditory conditions. The first model included all five conditions (Condition 1: visual only; Condition 2: visual + Single Talker; Condition 3: visual + English–English; Condition 4: visual + English–Latin; Condition 5: visual + English–Musical Rain). The dependent variable was log-transformed reaction times (log RTs), and the fixed factors were group (two levels, monolingual, bilingual), condition (five levels), and the interactions between them. Participant age and parental SES were also included as predictors, and subjects as crossed random effects. Results showed a significant effect of condition [F(4,8645.6)=26.62, *p*<.001, η_p^2 =.01]; age [F(1,77)=24.19, *p*<.001, η_p^2 =.24]; and a significant interaction between condition and group [F(4,8645.6)=5.38, *p*<.001, η_p^2 =.0025].

Pairwise comparisons confirmed that the bilingual group's log response times were slower than the monolingual group's in all conditions except Condition 1 (visual task only). Condition 1 showed equivalent performance between the groups [t=-1.79, p=.37, d=-.084], but there was a significant difference between the groups' performance from Condition 2 onwards, with the difference most significant in Conditions 3 – 5 (visual + auditory interference conditions), which are arguably the conditions that demand most attentional resource. A summary of response times by condition, standard deviations and the log-transformed differences between groups are shown in Table 7 and Figure 12.

Table 7

Study 1 (Children): Reaction times (SD) in ms and t tests between groups' logtransformed scores

Condition	Monolingual RT (ms)	Bilingual RT (ms)	t	p	d
1. Control	499 (182)	509 (179)	-1.79	.37	084
2. Single Talker	527 (192)	548 (185)	-3.10	<.01	148
3. English-English	523 (199)	576 (200)	-6.53	<.001	317
4. English-Latin	523 (194)	552 (186)	-4.20	<.001	202
5. English-MuR	519 (182)	572 (193)	-6.65	<.001	318

Figure 12

Study 1 (Children): Mean RT times by group and condition



The next analysis examined the patterns of RT responses in monolingual and bilingual groups separately. In *monolinguals*, a model, with log RTs as the dependent variable, which included condition (five levels) as a fixed factor, participant age and parental SES as predictors, and subjects as crossed random effects, showed significant effects of condition [F(4, 4416.6)=5.6, *p*<.001, η_p^2 =.005] and age [F(1, 38)=13.47, *p*<.001, η_p^2 =.26]. Post hoc t tests revealed that the only condition significantly different from the rest was Condition 1 (visual only), which had a significantly faster reaction time than each of the auditory conditions. None of the dual task conditions (Conditions 2-5) was significantly different from another. (See Table 8 for summary of post hoc t test results).

Table 8

Condition A	Condition B	t	р	d
1. Control	2. Single Talker	-3.5	<.01	16
1. Control	3. English-English	-2.72	<.05	13
1. Control	4. English-Latin	-2.83	<.05	13
1. Control	5. English-MuR	-2.86	<.05	13
2. Single Talker	3. English-English	.72	.77	.03
2. Single Talker	4. English-Latin	.62	.77	.03
2. Single Talker	5. English-MuR	.69	.77	.03
3. English-English	4. English-Latin	1	.96	005
3. English-English	5. English-MuR	05	.96	002
4. English-Latin	5. English-MuR	.06	.96	.003

Study 1 (Children): Differences in conditions for monolinguals

In *bilinguals*, the equivalent model showed a significant effect of condition [F(4, 4229)=26.1, p<.001, η_p^2 =.02], and age [F(1, 37.9)=9.98, p<.01, η_p^2 =.21]. However, post hoc t tests revealed a more uneven performance across conditions than in the monolingual participants, with reaction times in stronger interference conditions

significantly slower than both Control and Single Talker conditions – see Table 9 for full set of comparisons.

Table 9

Study 1 (Children): Differences in conditions for bilinguals

Condition A	Condition B	t	p	d
1. Control	2. Single Talker	-5.1	<.001	24
1. Control	3. English-English	-7.92	<.001	39
1. Control	4. English-Latin	-5.63	<.001	27
1. Control	5. English-MuR	-7.93	<.001	38
2. Single Talker	3. English-English	-3.01	<.01	15
2. Single Talker	4. English-Latin	55	.64	.12
2. Single Talker	5. English-MuR	-2.84	<.01	15
3. English-English	4. English-Latin	2.47	<.05	.12
3. English-English	5. English-MuR	.27	.79	.01
4. English-Latin	5.English-MuR	-2.28	<.05	11

Errors: The total error rate was 1.9% for monolinguals and 2.07% for bilinguals, which was not statistically significant [t(77.7)=-.36, p=.72]. Because of too few data points, of target and distractor errors across all trials (762/40000), no further error analyses were run.

5.4.2 Study 2 (high-exposure bilingual adults)

Auditory task: Similar to the results from Study 1, adults from both groups performed the task equally well, with overall comprehension scores of 98.4% in the monolingual group, and 98.5% in the bilingual group. A model with categorical response accuracy as the dependent variable; group (monolingual, bilingual), condition (Single Talker, English-English, English-Latin and English-MuR) and their interaction, age and socio-economic status (SES) as fixed effects, and participant as random intercept, indicated that the only significant factor was condition [X^2 (3, N = 84) = 10.9, p = .012]. Post hoc t tests revealed that this overall effect was driven by higher scores in the Single Talker condition than English-English [t(111)=2.4, p<.05, d=.37]; as well as higher in the English-Latin than English-English [t(105)=3.08, p<.05, d=.48); and higher in English-Latin than English-Musical Rain [t(106)=2.39, p<.05, d=.37). However, again there were no effects of group or group by condition interaction. These results indicate comparable performance of monolingual and bilingual adults in the behavioural auditory task, with similar patterns of performance in both groups across all four auditory conditions.

This was further supported by Bayesian analysis, which showed moderate support for the null hypothesis of group in an independent samples t-test (BF_{10} =.232). Additionally, a repeated measures ANOVA identified moderate evidence for an effect of condition (BF_{10} =6.4). In a subsequent best model assuming the effect of condition, there was very strong evidence for the null hypothesis in the factor of group (BF_{10} =.022), i.e., very strong support for the conclusion that the groups performed equivalently.

A summary of comprehension scores, standard deviations and their differences between groups for Study 2 are shown in Table 10 and Figure 13.

Table 10

Study 2 (High-exposure bilinguals): Percentage correct comprehension scores

Condition	Monolinguals	Bilinguals
1. Control	n/a	n/a
2. Single Talker	99.3 (2.6)	99 (3)
3. English-English	96.9 (7.8)	97.6 (5.3)
4. English-Latin	99.5 (3.1)	99.8 (1.5)
5. English-MuR	98.6 (5.2)	97.1 (7.4)

Figure 13



Study 2 (High-exposure bilinguals): Percentage correct comprehension scores

Results from the visual task:

The first analysis of response times aimed to test if performance varied by group or across the increased attentional load of the auditory conditions. The first model included all five conditions (visual only; visual + Single Talker; visual + English– English; visual + English–Latin; and visual + English–Musical Rain). The dependent variable was log-transformed reaction times (log RTs), and the fixed factors were group (two levels, monolingual, bilingual), condition (five levels), and the interactions between them. Participant age and SES were also included as predictors, and subjects as crossed random effects.

Results showed a significant effect of condition $[F(4,9334.5)=23.4, p<.001, \eta_p^2=.0099]$ but no Condition*Group interaction, suggesting that the same pattern of responses across conditions held across both groups. Post hoc t tests showed that the visual only condition (1) was significantly faster than each of the auditory conditions and Condition 2 (Single Talker) was also significantly faster overall than Condition 5. All other conditions were equivalent when averaged across groups.

Table 11

Study 2 (High-exposure bilinguals): Differences in conditions for all participants

Condition A	Condition B	t	р	d
1. Control	2. Single Talker	-4.45	<.001	14
1. Control	3. English-English	-5.62	<.001	18
1. Control	4. English-Latin	-6.16	<.001	2
1. Control	5. English-MuR	-7.02	<.001	23
2. Single Talker	3. English-English	-1.86	.11	
2. Single Talker	4. English-Latin	-1.2	.29	
2. Single Talker	5. English-MuR	-2.59	<.05	08
3. English-English	4. English-Latin	.68	.51	
3. English-English	5. English-MuR	66	.51	
4. English-Latin	5. English-MuR	-1.38	.24	

Despite a numerical trend for slower RTs in bilinguals overall and in each condition separately, Group also did not emerge as a significant predictor of RTs. A summary of response times by condition and standard deviations are shown in Table 12 and Figure 14.

Table 12

Study 2 (High-exposure bilinguals): Reaction times (SD) in ms between groups' log-transformed scores

Condition	Monolingual RT (ms)	Bilingual RT (ms)
1. Control	372 (89.8)	386 (95.1)
2. Single Talker	385 (95.6)	400 (98.7)
3. English-English	391 (98.6)	408 (110)
4. English-Latin	389 (95.6)	404 (106)
5. English-MuR	397 (105)	405 (98.1)

Figure 14

Study 2 (High-exposure bilinguals): Mean RT times by group and condition



Following the analyses strategy used in Study 1 (children), the next analysis examined the patterns of responses in monolingual and bilingual groups separately. In *monolinguals*, a model, with log RTs as the dependent variable, which included condition (five levels) as a fixed factor, participant age and SES as predictors, and subjects as crossed random effects, showed a significant effect of condition [F(4, 4697.2)=14, *p*<.001, η_p^2 =.01].

Post hoc t tests revealed that Condition 1 (visual only) had a significantly faster reaction time than each of the auditory conditions. In addition, Single Talker (Condition 2) was significantly faster than Condition 5 (Eng-MuR).

Table 13

Study 2 (Adults with high-exposure bilinguals): Differences in conditions for monolinguals

Condition A	Condition B	t	р	d
1. Control	2. Single Talker	-3.12	<.01	14
1. Control	3. English-English	-4.29	<.001	2
1. Control	4. English-Latin	-4.19	<.001	19
1. Control	5. English-MuR	-5.38	<.001	25
2. Single Talker	3. English-English	-1.23	.31	057
2. Single Talker	4. English-Latin	-1.06	.32	048
2. Single Talker	5. English-MuR	-2.35	<.05	11
3. English-English	4. English-Latin	.199	.84	.0093
3. English-English	5. English-MuR	-1.1	.32	051
4. English-Latin	5. English-MuR	-1.32	.31	061

In *bilinguals*, the equivalent model showed a significant effect of condition $[F(4,4633)=10.07, p<.001, \eta_p^2=.0086]$. Post hoc t tests revealed that the bilinguals showed a similarly significant difference to the monolingual group between Condition 1 (visual only) and all the other conditions, but not between any of the dual conditions.

Table 14

Condition A	Condition B	t	р	d
1. Control	2. Single Talker	-3.23	<.01	15
1. Control	3. English-English	-4.42	<.001	21
1. Control	4. English-Latin	-3.75	<.001	17
1. Control	5. English-MuR	-4.54	<.001	21
2. Single Talker	3. English-English	-1.33	.35	062
2. Single Talker	4. English-Latin	58	.62	027
2. Single Talker	5. English-MuR	-1.25	.35	058
3. English-English	4. English-Latin	.75	.62	.035
3. English-English	5. English-MuR	.15	.88	.007
4. English-Latin	5. English-MuR	64	.62	029

Study 2 (High-exposure bilinguals): Differences in conditions for bilinguals

Errors: The total error rate was .66% for monolinguals and .59% for bilinguals, which was not statistically significant [t(81.5)=-.51, p=.61]. As a result of too few data points across all trials (250/40000), no further error analyses were run.

5.4.3 Study 3 (low-exposure bilingual adults)

Auditory task: Consistent with the results from Studies 1 and 2, adults from both language groups performed the task equally well, with overall comprehension scores of 99.1% in the monolingual group, and 97.5% in the bilingual group. A model with categorical response accuracy as the dependent variable; group (monolingual, bilingual), condition (Single Talker, English-English, English-Latin and English-MuR) and their interaction, age and socio-economic status (SES) as fixed effects, and participant as random intercept, showed no effects of condition, group or group by condition interaction. These results indicate comparable performance of monolingual and bilingual adults in the behavioural auditory task, with equivalent performance in both groups across all four auditory conditions. This was further supported by Bayesian analysis, which showed anecdotal support for the null hypothesis of group in an independent samples t-test (BF_{10} =.807). A subsequent repeated measures ANOVA including the factors of group, condition and covariate of age also found anecdotal evidence for the null hypothesis in the factor of group (BF_{10} =.65), i.e., anecdotal support for the conclusion that the groups performed equivalently.

A summary of comprehension scores, standard deviations and their differences between groups for Study 3 are shown in Table 15 and Figure 15.

Table 15

Study 3 (Low-exposure bilinguals): Percentage correct comprehension scores

Condition	Monolinguals	Bilinguals
1. Control	n/a	n/a
2. Single Talker	99.7 (1.86)	97.6 (5.77)
3. English-English	98.6 (5.16)	98.3 (7.59)
4. English-Latin	100 (0)	97.6 (5.77)
5. English-MuR	97.9 (5.59)	96.6 (7.69)

Figure 15



Study 3 (Low-exposure bilinguals): Percentage correct comprehension scores

Results from the visual task:

As in the previous two studies, the first analysis of response times aimed to test if performance varied by group or with the addition of the increased attentional load of the auditory conditions. The first model included all five conditions (visual only; visual + Single Talker; visual + English–English; visual + English–Latin; and visual + English–Musical Rain). The dependent variable was log-transformed reaction times (log RTs), and the fixed factors were group (two levels, monolingual, bilingual), condition (five levels), and the interactions between them. Participant age and SES were also included as predictors, and subjects as crossed random effects. Results replicated the pattern that emerged from the equivalent model for children in Study 1: there was a significant effect of condition [F(4,6439.3)=7.04, *p*<.001, η_p^2 =.044]; age [F(1,54.8)=4.16, *p*<.05, η_p^2 =.07]; and a significant interaction between condition and group [F(4,6439.3)=3.32, *p*<.01, η_p^2 =.0021].

Pairwise comparisons confirmed that the bilingual group's log response times were equivalent to the monolingual group's in all conditions except Condition 5 (English-MuR), in which the monolinguals were significantly slower than the bilinguals [t=3.37, p<.01, d=.188]. A summary of response times by condition, standard deviations and the log-transformed differences between groups are shown in Table 16 and Figure 16.

Table 16

Study 3 (Low-exposure bilinguals): Reaction times (SD) in ms and t tests between groups' log-transformed scores

Condition	Monolingual RT (ms)	Bilingual RT (ms)	t	р	d
1. Control	380 (95.1)	387 (93.7)	-1.5	.135	08
2. Single Talker	393 (92.0)	388 (96.5)	1.3	.193	.072
3. English-English	393 (92.6)	388 (95.3)	1.19	.236	.067
4. English-Latin	394 (92.4)	389 (95.4)	1.10	.272	.061
5. English-MuR	408 (102)	390 (94.2)	3.37	<.01	.188

Figure 16



Study 3 (Low-exposure bilinguals): Mean RT times by group and condition

The next analysis examined the patterns of RT responses in monolingual and bilingual groups separately. In *monolinguals*, a model, with log RTs as the dependent variable, which included condition (five levels) as a fixed factor, participant age and SES as predictors, and subjects as crossed random effects, showed a significant effect of condition [F(4, 3226.1)=9.9, *p*<.001, η_p^2 =.01] and age [F(1, 26.8)=5.59, *p*<.05, η_p^2 =.17].

Post hoc t tests revealed that Condition 1 (visual only) had a significantly faster reaction time than each of the auditory conditions. In addition, English-Musical Rain (Condition 5) was significantly slower than all the other conditions.

Table 17

Study 3 (Adults with low-exposure bilinguals): Differences in conditions for	-
monolinguals	

Condition A	Condition B	t	p	d
1. Control	2. Single Talker	-2.99	<.01	17
1. Control	3. English-English	-2.78	<.05	15
1. Control	4. English-Latin	-3.09	<.01	17
1. Control	5. English-MuR	-5.43	<.001	3
2. Single Talker	3. English-English	.15	.9	.0083
2. Single Talker	4. English-Latin	13	.9	0069
2. Single Talker	5. English-MuR	-2.71	<.05	15
3. English-English	4. English-Latin	27	.9	015
3. English-English	5. English-MuR	-2.79	<.05	16
4. English-Latin	5. English-MuR	-2.57	<.05	14

In *bilinguals*, the equivalent model showed no significant effect of condition or age. Post hoc t tests revealed that there were no significant differences between the bilinguals' log RTs in any conditions, even in the Control Condition 1 (visual only).

Errors: The total error rate was .54% for monolinguals and .49% for bilinguals, which was not statistically significant [t(224)=-.35, p=.73]. Because of too few data points (200/40000) no further error analyses were run.

5.5 Comparisons between studies

The final set of analyses combined the visual RT data across the three studies, in order to directly compare the performance of monolingual and bilingual participants across the two age groups, and by language exposure in the two adult cohorts. The first analysis explored trends across age groups for monolinguals and bilinguals separately.

5.5.1 Analysis by age: Comparison of visual data across Studies 1, 2 and 3

Due to the RTs in children being slower than in either cohort of adults, age group (Study) was expected to be a significant predictor of RTs in both monolingual and bilingual groups. Condition was also expected to be a significant predictor in both groups, reflecting the finding that the Control condition was consistently faster than other conditions in all previous analyses. The key question was therefore whether Study and Condition would interact in either monolingual or bilingual groups, suggesting a differential pattern of responses in children and adult participants.

In monolinguals, a model included fixed factors of study (children, high-exposure adults, low-exposure adults), condition (five levels) and their interaction, participant age and SES; and participant as random intercept. Results showed that both study $[F(2,107.8)=37.58, p<.001, \eta_p^2=.41]$ and condition [F(4,12347.6)=22.68, p<.001, n_{p}^{2} =.0073] were significant predictors, but there was no interaction between them, indicating that all monolingual groups responded comparably to all five conditions, and that the two adult groups of monolinguals, while faster than the Study 1 monolinguals (as to be expected), did not vary in their pattern of response to the visual task relative to monolingual children. This is consistent with the previous analysis of patterns in the monolinguals for each study, which found that in all three studies Condition 1 (control, visual only) was significantly faster than every other condition, demonstrating a robust PRP effect (psychological refractory period, or psychological delay incurred when processing a dual task relative to a single task). This result corroborates the literature on dual tasks, which has established that the PRP is a consistent outcome of the cognitive processing of multiple tasks simultaneously (Pashler, 1994; Welford, 1952; Osman & Moore, 1993; Strobach et al., 2015).

134

In bilinguals, the equivalent model also revealed that study [F(2,107.8)=76.75, p<.001, $\eta_p^2=.59$] and condition [F(4,12075.1)=26.83, p<.001, $\eta_p^2=.0088$] were significant predictors, but there was also additional robust interaction between the two [F(8,12075.1)=7.38, p<.001, $\eta_p^2=.0049$]. Previous posthoc tests had already established that there were differences in the patterns of responses by the bilinguals of the different studies. To summarise: 1) in Study 1 (children) the Control Condition 1 was significantly faster than all the other conditions; and additionally Single Talker and English-Latin were both significantly faster than the English-English and English-Musical Rain conditions; 2) in Study 2 (high-exposure bilinguals), Control Condition 1 was significantly faster than all the auditory conditions (2-4), which were equivalent; and 3) in Study 3 (low-exposure bilinguals) there were no significant differences between the bilinguals' log RTs in any conditions. Figure 17 displays a summary of mean RTs for bilingual children were disproportionately affected by interference conditions 3-5 relative to any of the other language or age groups.

Figure 17

Patterns of RTs across conditions for Studies 1, 2 and 3



Additional posthoc tests, comparing the bilingual groups against each other, by study and by condition, showed that not only were the reaction times from Study 1 (children) significantly slower than the bilingual adults in Studies 2 and 3 (all p<.001) with large effect sizes (*d* ranging from .89-1.29) – as to be expected – but there were also significant differences between the bilingual adults in Studies 2 and 3 in every condition except Condition 1 (Control). This brings us to the next comparison: between the two adult groups, to see the effect of differential bilingual language experience.

5.5.2 Analysis of adult language groups: Comparison of data across Studies 2 and 3

As seen above, the adult bilinguals from studies 2 and 3 not only differed from the children (Study 1) but also from each other (details in Table 18 below). This warrants further investigation. The groups of bilinguals also varied slightly in how they compared to their monolingual counterparts (the monolinguals and bilinguals in Study 2 were equivalent in the model; the groups in Study 3 were different and posthoc tests revealed that this was due to the Study 3 monolinguals having significantly slower (higher) RTs in Condition 5 (English-Musical Rain).

Before focusing on the adult bilingual participants in each study, let us first turn to the monolinguals and establish if they are equivalent and therefore a reliable comparator in each study.

Analysis across adult monolinguals in Studies 2 and 3

Demographically, the monolinguals across the two adult studies were equivalent by age (Study 2: mean=28.9 yrs, SD=7.29; Study 3: mean=31.0 yrs, SD=6.69; t(63.5)=-1.27, p=.21, d=-.301); but not by SES, measured by level of education, in which the monolinguals in Study 3 scored higher (Study 2: mean=1.5, SD=.595; Study 3: mean=2.03, SD=.626; W=364.5, p=<.01). A model of comprehension scores of monolinguals across the two studies, with categorical response accuracy as the

dependent variable; study (2, 3), condition (Single Talker, English-English, English-Latin and English-MuR) and their interaction, age and socio-economic status (SES) as fixed effects, and participant as random intercept, showed condition as the only significant factor [X^2 (1, N = 71) = 11.04, p = .012], caused by a slight drop in performance in Condition 3, but none of study or study by condition interaction. These results indicate comparable performance of monolingual adults across studies 2 and 3 in the behavioural auditory task, with equivalent performance in both studies across all four auditory conditions (including slightly lower scores in Condition 3 for both studies equally).

In addition, Bayesian analysis in JASP (version 0.13) showed moderate evidence for the null effect of experimental group in an independent samples t-test of overall comprehension scores (BF_{10} =.32). A subsequent Bayesian repeated measures ANVOA, including the factors of experimental group and condition, and age as a covariate, identified anecdotal evidence for an effect of condition (BF_{10} =1.32) and a best model assuming the effect of condition showed moderate evidence for the null effect of experimental group (BF_{10} =.156). This supports the conclusion that there was no statistical difference in the monolingual adults' comprehension scores across Studies 2 and 3.

Analysis of the two monolingual groups on the visual task with log-transformed reaction times (log RTs) as the dependent variable, and fixed factors of study (two levels, 2, 3), condition (five levels), and the interactions between them, participant age and SES as predictors, and subjects as crossed random effects; found a significant effect of condition [F(4,7927.3)=22.8, *p*<.001, η_p^2 =.01]; but no effect of study nor interaction between study and condition. Despite the difference in the groups' SES scores, there was also no effect of SES in the model. Post hoc t tests found no significant differences between monolinguals in each study in any condition.

137

In fact, a version of Figure 17 shown in Figure 18 below, focusing on Studies 2 and 3 only, shows that the monolingual groups' pattern of scores across conditions is remarkably similar.

Figure 18

Patterns of monolingual and bilingual RTs across conditions in Studies 2 and 3



Analysis across adult bilinguals in Studies 2 and 3

Demographically, the bilinguals across the two adult studies were equivalent by age (Study 2: mean=27.4 yrs, SD=5.95; Study 3: mean=28.5 yrs, SD=7.72; t(50)=-.643, p=.523, d=-.16); and by SES, measured by level of education (Study 2: mean=1.81, SD=.74; Study 3: mean=1.77, SD=.61; W=457, p=.945). Where they did differ, however, was in language background. The language questionnaire captured their age of acquisition (AoA), proficiency in their second language (English), and

percentage of the time the participants estimated they used each of their languages, including their native language and English. The AoA of each of the groups was not significantly different (Study 2: mean=6.21 yrs, SD=6.86; Study 3: mean=6.64 yrs, SD=2.5; t(44.9)=-.33, p=.74, d=-.077), but the bilinguals in the two studies were different on scores of proficiency out of a maximum of 10 (Study 2: mean=9.31, SD=.96; Study 3: mean=8.66, SD=1.28; t(51.1)=2.28, p<.05, d=.59) and their self-assessment of English usage (Study 2: mean=73%, SD=24.9; Study 3: mean=33.6%, SD=26.9; t(57.2)=6.26, p<.001, d=1.53). The difference in English usage was the most evident and Figure 19 shows the different distributions in the two studies.

Figure 19



Percentage of English used regularly by bilinguals in Studies 2 and 3

Note. The question in the language questionnaire was: Please list the approximate percentage of the time you are using each language (please include your native language too). The figure above represents the percentage each group scored for their use of English specifically

It is important to note that even though the self-reported proficiency score for bilinguals in Study 3 was significantly lower than the score for Study 2, this did not negatively impact comprehension scores, which were equivalent for the two groups of bilinguals across studies. A model of comprehension scores of bilinguals across the two studies, with categorical response accuracy as the dependent variable; study (2, 3), condition (Single Talker, English-English, English-Latin and English-MuR) and their interaction, age and socio-economic status (SES) as fixed effects, and participant as random intercept, showed an effect of condition [X^2 (1, N = 71) = 8.59, p = .035] but none of study or study by condition interaction. These results indicate comparable performance of bilingual adults across Studies 2 and 3 in the behavioural auditory task, with no loss of performance due to proficiency.

This was supported by Bayesian analysis, which showed anecdotal evidence for the null effect of group in both an independent samples t-test of overall comprehension scores (BF_{10} =.451), and a repeated measures ANOVA, with factors of group, condition and age as a covariate (BF_{10} =.339).

For the analysis of the visual task, a model of log reaction times of bilinguals across the two studies, with RT as the dependent variable; study (2, 3), condition (Single Talker, English-English, English-Latin and English-MuR) and their interaction, age and socio-economic status (SES) as fixed effects, and participant as random intercept, showed an effect of condition [F(4, 7846.4) = 6, *p*<.001, η_p^2 =.0031]; and a study*condition interaction [F(4, 7846.4) = 3, *p*<.05, η_p^2 =.0015]. The post hoc t tests revealed significant differences in all conditions except Control (Condition 1, visual only).

Table 18

Comparison of bilinguals' scores in Studies 2 (High-exposure bilinguals) and 3 (Lowexposure bilinguals) by condition

Condition	Study 2 (high-	Study 3 (low-	t	p	d
	exposure bilinguals)	exposure bilinguals)			
1. Control	386 (95.1)	387 (93.7)	134	8.93	007
2. Single Talker	400 (98.7)	388 (96.5)	2.56	<.05	.132
3. English-English	408 (110)	388 (95.3)	3.73	<.001	.19
4. English-Latin	404 (106)	389 (95.4)	2.74	<.01	.14
5. English-MuR	405 (98.1)	390 (94.2)	3.23	<.01	.16

These results from the visual task indicate more differences between the bilinguals in Studies 2 and 3 with different language exposure, than between either of them and their monolingual counterparts within the studies. All these results are discussed below.

5.6 Discussion

The results of the previous study (EEG Experiment 1, described in Chapter 4), with a lower and flatter pattern of neural attentional indices in bilinguals despite equivalent behavioural scores, support the proposition that the inhibition of the non-target language in bilinguals does not affect performance on primary tasks, but does lead to a reduction of overall capacity (Dornic, 1980). The purpose of the Studies described above was to explore the consequences of increasing the processing demands, to investigate if bilinguals' optimised performance in spite of constrained capacity (or the ability to "do more with less"), has limitations in the context of dual tasks. Therefore, increasing the processing demands in a controlled way in a dual task, would establish if there were any indicators of additional interference in the bilingual group as well as the expected PRP (psychological refractory period, or well-

documented delay as a result of simultaneous task processing) for all participants as a result of the "central bottleneck" (Pashler, 1994).

The new behavioural task was designed by adding a secondary task to the dichotic listening task from EEG study 1, which additionally varied the attentional load during the study. The prediction was that bilinguals, with reduced attentional capacity due to managing their additional language processing demands, would need to prioritise and economise their responses to competing task demands, relative to monolingual participants, to maintain task performance and that this would affect performance in the secondary task.

Performance was measured on two tasks: a dichotic listening task with comprehension scores as an performance indicator; and a visual T.O.V.A. task in which both accuracy, through errors of commission and omission, and speed of response, through response times were tracked. The study was conducted on three groups spanning age ranges and language experience, with one group of school-age children (Study 1, age range 7-12), and two groups of adults (Studies 2 and 3, age range 18-45), who were differentiated by exposure to English. Study 2 recruited a high-exposure bilingual group and Study 3 a low-exposure bilingual group, as well as monolingual adults in the same age range.

The results showed equivalent performance in the auditory (primary) task between monolingual and bilingual participants, across all three studies. For the visual (T.O.V.A.) task, accuracy, measured by error rates, was extremely high and equivalent between monolingual and bilingual participants, in all studies.

Reaction times in the visual (secondary) task, however, showed key differences between conditions, age groups and language groups. These are summarised and discussed below.

Firstly, there was an expected PRP effect from the addition of the auditory task to the Control condition of the visual task only. Condition 1 (Control, visual only) was

142

significantly faster for all groups of participants across all three studies, confirming a robust psychological refractory period (PRP), the delay generated when performing two tasks simultaneously, consistent with the literature (Pashler, 1994; Telford, 1931; Vince, 1949; Welford, 1952; Pashler & Johnston, 1998; Osman & Moore, 1993; Schubert, 2008).

5.6.1 Differences between monolingual and bilingual children

The results from Study 1 revealed that bilingual children's reaction times were significantly slower relative to monolingual children's in all the dual-interference conditions (Conditions 2-5). The only exception was the Control Condition 1, which comprised the visual task only, in which the groups were equivalent. This result suggests that, as the attentional load increased through multiple tasks, bilingual children's performance was not affected uniformly across tasks and conditions. Rather, the processing system in bilingual children distributed resource in response to the tasks' demands, maintaining optimal performance in the task of higher priority, at the expense of resource to perform the secondary element. This account is consistent with the multimode theory of attention. In their dual task incorporating an auditory task and visual task measuring reaction times, Johnston & Heinz (1978) found that conditions of greater semantic processing in the auditory task generated slower reaction times in the visual task, thus demonstrating the limitation of attentional capacity and its effect of the processing of multiple tasks depending on auditory conditions. The Johnston & Heinz (1978) experiment did not investigate the additional impact of bilingualism, but their conclusion of longer reaction times equating to constraints on capacity is consistent with the results shown by bilingual children.

Thus for bilingual children a hierarchy was revealed in which comprehension of the auditory task and accuracy in the visual task were prioritised and optimal performance maintained. This budgeting of attentional resource left a deficit, relative to monolinguals, in speed in the visual task. I will discuss next whether this

143

order of priority was consistent across age groups and language exposure from the results of the adult studies.

5.6.2 Differences between adult bilinguals in Studies 2 and 3

In spite of both adult groups of bilinguals showing comparable speed (or faster in the English-MuR condition in Study 3) to monolinguals in the conditions of dual interference, there still appeared to be significant differences between the bilinguals' reaction times across the adult studies. This was established after ensuring that the monolinguals across Studies were in fact equivalent. This difference held across every single dual interference condition, in which lowexposure bilinguals were significantly faster than high-exposure bilinguals. One possible interpretation here would be that the high-exposure bilinguals, resident in the UK and using their second language (English) on a daily basis, have greater parallel activation (Green, 1998) and hence processing load due to the second language (Wickens, 2007). Indeed, according to the adaptive control hypothesis (Green & Abutalebi, 2013), the language context in which processing demands such as interference suppression and conflict monitoring – are highest is the duallanguage context, i.e. the profile of the bilinguals in Study 2. The results from comparing the two groups of adult bilinguals suggest that the effect of maturation on selective attention proposed in the section above is enough to close any gap between monolingual and bilingual speakers; but that there is still some residual processing cost that, although it does not manifest when compared to monolinguals, is exposed in speed of response in relation to other bilinguals who do not face the same daily processing demands.

5.6.3 Difference between monolinguals and bilinguals across studies

The initial analysis for each study testing the effect of group, condition, age and SES on RTs, found a significant group:condition interaction for two of the studies: Study 1 (children) and Study 3 (low-exposure bilinguals). Post hoc tests revealed that in
Study 1 bilingual children had significantly slower reaction times in all conditions except the control (Condition 1). In Study 3 (low-exposure bilinguals), the only significantly different condition between the language groups was English-Musical Rain, in which bilinguals were significantly faster. Study 2 (high-exposure bilinguals) revealed no group:condition interaction in the model.

This arguably suggests a developmental trajectory for bilinguals revealed by a dualtask paradigm; in which apparent limitations of attentional resource, while still not impacting on the primary task, are shown in scores of the secondary task in conditions of dual interference in children; but that these effects on response times dissipate with maturation. This adds nuance to the previous discussion of degeneracy, indicating that, while behavioural performance on a primary task is indeed equivalent from childhood (as seen in Study 1 here and EEG Experiment 1 described in Chapter 4); attentional systems impacted by bilingualism may not reach parity under the increased demands of dual tasks, which extend the bandwidth of normal processing, until adulthood and thus maturation of the prefrontal cortex associated with selective attention processing (Giedd et al., 1999; Gogtay et al., 2004; Sowell et al., 2001; Tsujimoto, 2008).

5.6.4 Implications of the prioritisation of the tasks

The prioritisation of attentional resource in response to the auditory vs visual task, manifested by equivalence in all the groups and studies in the auditory comprehension scores and variability across age/language groups in the response times (but not accuracy) to the T.O.V.A., is plausibly a result of the paradigm used in this study. One explanation for the apparent hierarchy of response is that the auditory task was presented explicitly as the primary task. Another explanation is that the auditory task was also more varied and entertaining than the visual task, which was simple and repetitive, thus capturing more attention through content and presentation rather than through a directive. For the visual task, it is important to note that accuracy was not emphasised over speed in the instructions or practice. [The manipulation of specific task instructions has been established as an influence on the outcome of the accuracy-speed tradeoff (Katsimpokis et al., 2020)]. Furthermore, any reaction time longer than 2000ms was deemed a "time out" and recorded as an error of omission, so the slower reaction times as a result of analysis were in the context of correct responses and within a time-limited window. Indeed, slower speeds for bilingual participants – especially younger ones – in the visual task are consistent with neural explanations of how a task using the speed-accuracy tradeoff is processed: the processing network progresses through a series of stages including noise filtering, integration and amplification of evidence, and choice selection (Standage et al., 2011), as well as motor control. In the case of a constrained processing capacity, it is plausible that these processing steps would take fractionally longer – but not be any less accurate.

5.6.5 The findings in the context of other studies

The results of a bilingual limitation in one element of the dual task in order to maintain optimal performance in the primary elements are at odds with some of the previous research investigating dual tasks and bilingualism, notably the studies combining a visual search with counting backwards (Janic et al., 2020) and driving simulation and verbal tasks (Telner et al., 2008). However, in those studies not only were different tasks used and assessed, but the computation of performance was based on the decrements of participants' scores, compared to their baseline score recorded in a control condition, rather than the absolute values analysed above. In order to compare like for like and following the method employed in these papers, decrement scores for each participant were calculated by subtracting each condition's normalised response times from the participant's mean response time in the Control condition (visual only, Condition 1). Comparing the decrements scores of monolinguals and bilingual participants in each study revealed a significant difference between the groups in the children only, consistent with the original results detailed above. Contrary to the previous studies using decrements (Janic et al., 2020; Telner et al., 2008), these showed that bilinguals' performance was more

significantly affected than that of monolinguals in each of the dual conditions that required the inhibition of auditory interference. See Table 19 for results of decrement analysis in all three studies.

Table 19

Decrement t test results for Studies 1, 2 and 3 between monolinguals and bilinguals

Condition	Study 1 (Children)		Study 2 (High-exposure			Study 3 (Low-exposure			
			bilinguals)			bilinguals)			
	t	р	d	t	р	d	t	р	d
2. Single Talker	1.59	.11	.076	.17	.87	.0078	-2.17	<.05	12
3. English-English	4.16	<.001	.2	.13	.9	.0059	-2.22	<.05	13
4. English-Latin	3.06	<.01	15	34	.74	016	-1.75	.08	097
5. English-MuR	4.53	<.001	.22	-1.47	.14	068	-4.56	<.001	25

As seen in the above table, for the adult high exposure bilinguals study (Study 2), comparisons of decrement scores relative to the control task revealed no significant difference between monolinguals and bilinguals.

Finally, results of the decrement analysis for the adult study on low-exposure bilinguals (Study 3) showed not only a significant result for Condition 5 (English-MuR), consistent with the original results, but also significant difference between monolinguals and bilinguals in Single Talker (Condition 2) and English-English (Condition 3). These, like the Single Talker condition, indicated that bilinguals suffered less effect of PRP compared to the Control condition than monolinguals, in direct contrast to the results from children.

In sum, the use of decrement analysis as employed in other studies (Janic et al., 2020; Telner et al., 2008) did not change the overall conclusions that performance differed for the bilinguals in Study 1 (Children) relative to monolinguals; and this effect had been overcome or reversed by adulthood (Studies 2 and 3).

5.6.6 Conclusions

This set of studies investigated the effect of increased processing demands on bilingual performance across three groups. The data from Study 1 show that there is a process of economisation and prioritisation of elements of tasks in bilingual children, in this case concentrating the resources on comprehension of the auditory task and accuracy of the visual task, at the expense of speed in the visual task.

This results in limitations in speed in bilingual children's performance in the visual, but not accuracy of auditory, element of a dual task. This task, tapping two competing domains, deliberately increases attentional load beyond what can normally be accommodated in a flexible attentional processing system. Importantly, these speed limitations in the secondary task appear to recede, relative to monolinguals, by adulthood. Furthermore, analysis of adults with differing second language exposure has revealed differences in performance which are not discernible when comparing either group to monolingual peers.

In the context of the effects of bilingualism, this suggests that maturation of the selective attention system enables adaptation to accommodate the combined load of second language processing and competing task demands; resulting in optimal selective attention processing and performance in "extreme" tasks. It has already been established that behavioural performance is equivalent for children in a primary task, even when neural indices of attention are lower for bilingual than monolingual children (as in Chapter 4) supporting the view that modifications to the neurofunctional architecture as a consequence of bilingualism (Hayakawa & Marian, 2019) are aimed at supporting performance within the boundaries of typical demands on selective attention. The results here still indicate that behavioural performance is maintained for the primary element of a dual task (in this case, auditory), but differs for bilingual children when performing the secondary task element (in this case, visual). The assumption therefore is that the additional processing load of another language leads to constrained capacity, which triggers the

need to economise and prioritise the demands of a task which extends beyond the bandwidth of a typical attentional load.

Critically, however, any bilingual difference relative to monolinguals in performance of the secondary task was no longer discernible in either of the adult groups. This suggests that, as the attentional system matures, bilinguals demonstrate degenerate behaviour (relative to monolinguals) not only in primary tasks but in secondary tasks as well. This, however, does not preclude some residual processing constraints, which although not large enough to differentiate bilinguals significantly from monolingual peers, emerge in indices of attentional processing (reaction times) when groups of bilinguals with different levels of language exposure are compared.

As an aside, one of these differences was that low-exposure bilinguals, as demonstrated in Figures 17 and 18, showed the overall fastest reaction times of all the language groups across all the studies. Perhaps this could be used as an incentive to promote learning a second language in an educational setting, to a reasonable degree of proficiency, which appears to yield benefits in selective attention tasks with multiple demands, without using the additional language every day.

Chapter 6: Experiment 3

6.1 Introduction

The final question was to what extent a difference observed in auditory selective attention patterns between monolinguals and bilinguals could be discerned in a nonlinguistic domain. To this end, I replicated EEG Experiment 1 with the same demographics, and target stimuli but replaced the unattended streams with different levels of non-linguistic interference.

As discussed in Chapter 4, selective attention is a critical component of children's executive functioning, with strong evidence pointing to its influence on outcomes such as academic success (Stevens & Bavelier, 2012) and mitigation of environmental barriers such as lower SES (Hampton Wray et al., 2017; Isbell et al., 2016). Bilingualism, as a consequence of the habitual processing demands associated with managing an additional language (Green, 1998), has been shown to trigger neuroplastic adaptations not only in areas associated with language processing (Perani, 2005); but also in regions associated with more generalised selective attention, such as the anterior cingulate cortex (Abutalebi et al., 2012). This is consistent with the hypothesis that bilingualism causes adaptation to control processes associated with these regions. Furthermore, some researchers claim that these modifications not only affect behaviour in verbal selective attention tasks (Kaushanskaya et al., 2014), but also generalise to nonverbal selective attention tasks (Carlson & Meltzoff, 2008). This has motivated theoretical frameworks such as the Adaptive Control Hypothesis (Green & Abutalebi, 2013), which details the attentional control processes affected, and specifically proposes a framework for precise predictions about the performance of bilinguals on non-verbal tasks.

However, such proposals of adaptation to control processes, leading to modified behaviour in non-verbal tasks, are not universally accepted, with many studies finding no difference between the performance of bilinguals and monolinguals in non-verbal attention tasks (Hilchey & Klein, 2011; Morton & Harper, 2007). Importantly, this proposed – and contested – modification to non-verbal as well as verbal selective attention processing occurs while children's selective attention processes are still undergoing development in a prolonged period of maturation (see Lane & Pearson, 1982, for review).

Furthermore, even non-verbal selective attention tasks must be performed in the context of finite attentional capacity (Kahneman, 1973), in which the management of a second language has been proposed to be a constraint (Wickens, 2007). Therefore, it is plausible to assume that there is a similar economisation of residual capacity in order to maintain optimal behavioural performance, comparable to that observed for attentional processing in the verbal domain. This would also have implications for frameworks such as the Adaptive Control Hypothesis (Green & Abutalebi, 2013) by placing predictions of bilingual performance on non-verbal tasks in the context of a narrower bandwidth of attentional capacity.

As before, this does not preclude the possibility of a resulting greater flexibility, i.e., doing "more with less" in bilinguals, but shifts the focus from inconsistent behavioural comparisons on non-verbal tasks (von Bastian et al., 2016) to patterns of adaptation and economisation in the neural processing of such tasks. Indeed, one prediction would be that, as in EEG Experiment 1, neural indices would not point to an overall enhancement, but might show some reduction and reconfiguration, in the bilingual group's attentional processing of the task, using non-verbal unattended stimuli.

6.2 Current study

To establish if the patterns that emerged in EEG Experiment 1 also generalise to the non-linguistic domain, the current study again investigated the neural encoding of attended and unattended speech envelopes in monolingual and bilingual listeners

aged 7-12. I used the same attended narratives (and their speech envelopes) as in the previous experiment, but substituted the unattended stimuli with non-linguistic streams. As established in the previous EEG experiment, neural encoding of speech envelopes is a proven method for investigating attentional processing. Evidence that there is a robust correlation between attended speech envelopes and neural activity (Aiken & Picton, 2008; Di Liberto et al., 2018; Mesgarani & Chang, 2012) underpins the selective entrainment hypothesis (Zion Golumbic et al., 2013) which proposes that this neural activity, or encoding, is essential for speech intelligibility in adults (Obleser & Kayser, 2019; Olguin et al., 2018) and children (Power et al., 2016; Ríos-López et al., 2020).

Following the design of EEG Experiment 1 described in Chapter 4, the current study employed EEG to capture the neural encoding of attended and unattended speech envelopes in monolingual and bilingual children. As before, linear regression as implemented in the mTRF toolbox (Crosse et al., 2016) was used to model the relationship between the speech signal and the neural data, and applied in a backward direction to assess how well the attended and unattended speech envelopes could be reconstructed from the responses of the neuronal populations that encode them (see Methods chapter for more details). The measure of correlation between the speech envelope and EEG signal was the reconstruction accuracy score computed by the mTRF toolbox for each matching sentence/EEG epoch per condition per participant, using all neural data simultaneously in a "leaveone-out" computation. This method of assessing correlation is not only suitable for multi-channel data such as EEG (O'Sullivan et al., 2015), but an established method for quantifying neural tracking in children (Power et al., 2016).

Although the mTRF technique has been shown to be particularly suitable for natural speech (Wong et al., 2018; Sassenhagen, 2019; Pasley et al., 2012) and differentiation of speech streams in the cocktail party design (Rimmele et al., 2015; Riecke et al., 2018; O'Sullivan et al., 2015), it is also suitable for sensory signals in general (Crosse et al., 2016), including non-speech streams (Crosse et al., 2021). Examples of studies tracking non-speech auditory stimuli include streams derived

from linguistic stimuli such as unintelligible vocoded speech (Di Liberto et al., 2018), or individual syllables (Power et al., 2016) ; to auditory stimuli devoid of any linguistic input such as birdsong (Di Liberto et al., 2018), animal communication calls (Machens, 2004) and classical music (Di Liberto et al., 2020). The mTRF toolbox has even been used in conjunction with non-auditory stimuli, for example to track the response to motion and luminance of visual streams (Jessen et al., 2019).

As in the first EEG experiment, children were presented with two competing auditory stimuli simultaneously, and instructed to attend to one while ignoring the other. The nature of the competing stream was manipulated across four different conditions to create different types of non-linguistic interference. As in EEG Experiment 1, the first condition was 'Single Talker', a control condition where children attended to a narrative presented in one ear, with no interference presented in the other ear. This established a baseline measure for attentional encoding in monolingual and bilingual listeners, which could also be compared to the previous EEG Experiment 1 to assess equivalence of baseline responses.

The non-linguistic interference streams for the other three conditions were selected with the aim of providing naturalistic nonverbal distractors for the target sample of school-age children (age range 7-12) that they would plausibly hear in an everyday scenario. Thus in the second condition, children attended to a narrative in English presented in the attended ear while ignoring unintelligible multi-talker babble, recorded from children in a classroom, presented in the unattended ear (English– Babble condition). In the third condition, children attended to a narrative in English while ignoring a stream of typical road traffic noise (English–Traffic condition). In the fourth condition, the interfering stream was a simple melodic backing track (English– Music condition), of the genre used in television adverts, but with no lyrics, to avoid triggering a response to linguistic content.

As before, participants were instructed to listen to the attended stream for comprehension, a task that children in this age group had already performed without difficultly in EEG Experiment 1. Based on the previous experiment's

comprehension results, it was also expected that there would be no significant difference between the ability of monolingual and bilingual listeners to perform the task. As before, by maintaining consistency in target (comprehension) narratives and behavioural performance, the focus was on the neural response patterns under new conditions of interference, rather than performance alone.

The set of conditions described above allowed an assessment of whether the patterns discerned in conditions of linguistic interference in EEG Experiment 1, also generalised to a study using non-linguistic unattended stimuli. To summarise the findings from EEG Experiment 1, results showed that attention modulated the neural encoding of speech envelope in both monolingual and bilingual children, with the type of distractor further influencing the strength of the encoding of the attended stream. Critically however, the way different distractors influenced attentional encoding differed between the groups. Bilingual reconstruction scores tended to be lower and flatter. Monolinguals showed substantial variability between conditions, with significantly higher attended reconstruction scores in conditions of reduced (Musical Rain) or no interference (Single Talker) relative to their reconstruction scores were also significantly higher than the bilinguals' reconstruction scores in the same (no or low interference) conditions.

This tendency for lower neural indices of selective attention in bilingual children – despite equivalent behavioural performance – supports the hypothesis that language selection and inhibition themselves draw on the existing attentional capacity (Wickens, 2007), restricting the resources available to track the speech envelope. Results from EEG Experiment 1 further indicate that the subsequent economisation of this restricted resource leads to a modification of the encoding patterns across conditions in bilinguals, leading to a redistribution of the residual attentional capacity to optimise the usage of this constrained resource and perform the task in the context of the increased processing demands of bilingualism.

4.3 Materials and method

4.3.1 Participants

45 typically-developing children aged 7-12 were tested. One child (bilingual) was excluded from analysis after achieving an outlier comprehension score more than seven standard deviations lower than the sample mean, after which there remained 44 participants across two groups: 22 bilingual children (seven males, age M=9.67 yr, SD=1.56) and 22 monolingual children (fourteen males, age M=10.1 yr, SD=1.56), which were matched groupwise on mean and distribution of age (t=.83, p=.41). All participants were healthy with no history of hearing problems or neurological disorder. 41 were right-handed, with two of the left-handed children being bilingual and one monolingual. All participants' parents completed a language history questionnaire, which provided an overview of children's exposure to languages. As confirmed by the questionnaire, all monolingual participants were native speakers of English, with no significant exposure to other languages. The participants in the bilingual group all had a similar profile: the language they first learnt was not English, and they used this language at home on a daily basis. They were however fluent and highly proficient in English, following English-speaking curriculum at school, and with native-like English conversation skills comparable to their monolingual peers. The second languages spoken were Afrikaans, Cantonese, French, Hebrew, Korean, Mandarin, Polish, Russian and Spanish. Children were recruited via posters, social media, and word of mouth. Parental education information was collected as an indication of SES. The majority of participants' parents (92%) were educated to degree level or higher, and the groups were not significantly different on this approximation of SES (bilinguals M=2.68, SD=.48; monolinguals M=2.46, SD=.58; Mann-Whitney Z=-1.43, p=.16).

The study was approved by the Cambridge Psychology Research Ethics Committee, and performed in accordance with relevant guidelines and regulations. Prior to the testing session, parents and children were given detailed information on the aims of the project and what to expect from the session. Upon arrival, informed consent

was given by parents signing a consent form and the children an assent form. They were told they could withdraw from the study at any time.

Due to the covid pandemic, these data were acquired over a prolonged period from December 2019 to November 2021, with twelve participants tested in December 2019 – March 2020 and the remaining 33 in July 2021-November 2021 after lockdown restrictions were lifted.

4.3.2 Design

The experiment consisted of four conditions (Table 20). In each condition, children were attending to a story in English in one ear. Condition 1 had no interference in the other ear ('Single Talker'). In the other three conditions children were also presented with a non-verbal distractor in the other ear, which they were instructed to ignore. The nature of the distractor varied, from a stream of babble with no intelligible words ('English-Babble'), to a continuous stream of traffic noise ('English-Traffic') and a stream of simple and highly rhythmic melodic interference ('English-Music').

Table 20

Condition	Attended	Interference		
conunion	stream	interjerence		
<u>1. Single Talker</u>	English story 1	No interference		
2. English-Babble	English story 2	Multi-talker babble		
3. English-Traffic	English story 3	Traffic noise		
<u>4. English-Music</u>	English story 4	Music (no words)		

Experimental conditions

4.3.3 Stimuli

The target stories for the attended ear were the same as those used in EEG Experiment 1, i.e., four children's stories in English specifically aimed at this age group, taken from online resource storynory.com. These stories had been transcribed into 120 sentences each, with each sentence lasting approximately 3 seconds in length. Each target story was then split into 2 blocks and children attended to the first half in either the left or right ear (randomly assigned), with interference in the other, and then swapped ears for Block 2. Each block (half of a story) consisted of 60 sentences, with all 60 sentences concatenated with a 300ms gap between them to create a single block lasting 3.3 minutes. Block 1 was always the first half of the target story and Block 2 the second half. Gender of the speaker was kept the same for all target stories to reduce segregation strategies based on talker's gender (Brungart & Simpson, 2007). All stories' volumes were normalised to ensure equivalent average amplitude.

The non-linguistic interference of classroom babble and traffic noise were sourced from the BBC sound effects archive (https://sound-effects.bbcrewind.co.uk) which permits usage for non-commercial projects. The music interference stream was sourced from the online music for video resource Bensound (https://www.bensound.com/free-music-for-videos), which offers royalty-free downloads. As the distractor streams were all continuous, and the target stories concatenated with 300ms gaps of silence, it was necessary to ensure that attention was balanced impartially between the two streams in the "gaps". Accordingly, code in Matlab code was used to create a ramp effect where the distractor stream faded down and then faded back up at 3s intervals, synchronising with the silences in the target narratives. Finally, all distractor streams' volumes were also normalised to ensure equivalent average amplitude.

The same instructions were used as in EEG Experiment 1. These had been recorded by the female speaker who narrated the target stories, and were played before each

block in the target ear, telling the child: "This is your right/left ear. Please listen carefully to the story in this ear, on your right/left side, and ignore the sound in the other ear".

4.3.4 Procedure

The participants had a practice session of listening to an English story in both the left and the right ear while ignoring a distracting stream in the other ear, in order to familiarise themselves with the dichotic listening paradigm. After practice, they were asked to summarise the target story to check they could hear correctly and had understood the instructions to attend to one ear at a time. The task itself took 45-60 minutes. Children first heard Block 1 of Condition 1 (Single Talker) followed by 10 comprehension questions. They then listened to Block 2 of Condition 1 (Single Talker), again followed by 10 comprehension questions. Each block was preceded by the recorded instructions in the relevant (target) ear. This procedure was repeated for the other three conditions, which were presented in a random order. Children were instructed to stay as still as possible while the stories were playing and were allowed to stretch, yawn etc. during the comprehension breaks. An example sequence of a block is illustrated in Figure 1 in the Methods chapter. Comprehension questions consisted of simple sentences to check understanding of each story (for example: 'This story is about a boy named PETER/TOM'), and children pointed or verbally confirmed which option they thought was correct. The children did not receive feedback on their responses. At the end of the experiment children were presented with a certificate of completion and a £20 Amazon gift voucher as compensation for their time.

4.3.5 Data collection and preprocessing

As described in the Methods chapter, EEG was recorded using a child-sized 64channel electrode net while stimuli were played through foam-tipped earphones in the pre-allocated part-randomised order. All data were pre-processed in MATLAB: EEGLAB Toolbox (Delorme & Makeig, 2004) following the 10 step process described in the Methods chapter. After the data had been cleaned following these steps, including filtering, ICA and removal of noisy components and interpolation of bad channels; epochs of 3s length were extracted, to enable a minimum of five minutes of EEG data per participant per condition as an input for the mTRF toolbox (Jessen et al., 2019), allowing for epoch rejection of outlier epochs. These were defined as epochs outside a *3SD* of the probability and kurtosis thresholds. This resulted in an overall epoch rejection of 16.15% overall (14.37% for monolinguals and 17.94% for bilinguals). By condition, epoch rejection was 15.9% in Single Talker, 19.26% in English-Babble, 14.74% in English-Traffic and 14.56% in the English-Music datasets. Finally, after re-referencing and resampling to 100Hz, the EEG data from all 44 participants were entered into the subsequent mTRF analysis.

4.3.6 Speech envelopes

Speech envelopes for attended speech streams and unattended non-linguistic stimuli were calculated using the Hilbert2 function in EEGLAB, downsampled to 100Hz to match the data and normalised using *nt_normcol* (Noisetools http://audition.ens.fr/adc/NoiseTools/).

4.3.7 Analyses

Neural tracking of the stimulus envelopes was computed in the same way as EEG Experiment 1, using multivariate temporal response functions in the mTRF toolbox (Crosse et al., 2016). As described in the Methods chapter, the backward model (reconstruction) was used to model the encoding of the speech as a function of the sound input, for all neural data simultaneously. The model was created using inputs of stimulus (normalised envelopes), signal (normalised EEG data), time lag limits, sampling rate and 12 regression parameters (12λ values, 1×10^{-3} : 1×10^{8}). The outputs were 12 reconstruction scores per epoch, of which the optimal (highest) r score was used in analysis. The model generated these scores using the "leave one

out" computation described in the Methods chapter. See the Methods chapter for figures 2 and 3 which illustrate the procedure of mTRF model computation, and the outcome of reconstruction for a sample sentence.

The reconstruction accuracy scores were then compared across groups, attention status and condition using linear mixed-effect models (Baayen et al., 2008) as implemented in the Ime4 R package (Bates et al., 2014). The step function in the ImerTest package (Kuznetsova et al., 2017) was used to arrive at the best-fitting model. The Satterthwaite approximation (Satterthwaite, 1946) was used for degrees of freedom. Significant p-values are reported at p<.05. All post-hoc tests were FDR corrected for multiple comparisons.

4.4 Results

4.4.1 Behavioural comprehension scores

Children from both groups performed the task equally well, with overall comprehension scores of 99.5% (SD= .62) in the monolingual group, and 98.9% (SD=2.6) in the bilingual group. To test for any differences between them, responses were coded as binary correct/incorrect and then analysed in a logistic regression model with binary scores as the dependent variable, and factors of group (two levels: monolingual, bilingual), condition (four levels: Single Talker, English-Babble, English-Traffic and English-Music) and their interaction, in addition to participant age, socio-economic status (SES) plus subjects as a random effect. The results showed that the only significant factor was age [X^2 (1, N = 44) = 9.3, p = .0023] and there was no difference in comprehension either by group or by condition.

In addition, Bayesian analysis showed anecdotal evidence (BF_{10} =.532) for the null hypothesis, defined as group effect of 0, in an independent sample t test for overall comprehension scores. A Bayesian repeated measures ANOVA in JASP (version 0.13), with default priors, including the factors of group and condition, and age as a

covariate, identified moderate evidence for an effect of age ($BF_{10}=7.8$). In a subsequent best model assuming the effect of age, there was strong evidence for the null hypothesis for group ($BF_{10}=.051$). This offers strong support to the conclusion that the groups performed equivalently in the comprehension task.

A summary of comprehension scores and standard deviations are shown in Table 21 and Figure 20.

Table 21

Comprehension scores and standard deviation by condition and group

Condition	Monolinguals	Bilinguals
Single Talker	99.3 (1.76)	99.5 (2.13)
English-Babble	99.3 (1.76)	99.1 (1.97)
English-Traffic	99.8 (1.07)	98.9 (3.43)
English-Music	99.8 (1.07)	98 (6.67)
Overall across conditions	99.5% (.62)	98.9% (2.6)

Figure 20





4.4.2 EEG Data

4.4.3 The effects of attention on speech reconstruction accuracy

In the analysis of the neural data, datapoints more than 1.5 interquartile ranges above the upper quartile or below the lower quartile were removed as outliers, excluding 157 datapoints (0.5% of the total). Visual inspection of residual plots did not reveal any obvious deviations from normality.

The first analysis of the neural data aimed to establish whether attention modulated speech reconstruction accuracy in children when presented with non-linguistic interference. It included the three conditions where both attended and unattended narratives were presented to the participants (English-Babble, English-Traffic and English-Music); thus excluding the condition where there was no interference (Single Talker). The dependent variable was reconstruction accuracy score, and the fixed

factors were group (two levels, monolingual, bilingual), attention (two levels, attended and unattended) and condition (three levels), and the interactions between them. Also included in the model were participant age and parental SES as predictors, and subjects and items as crossed random effects. Contrary to the findings in EEG Experiment 1, results from the model showed no overall effect of attention; however there was a significant effect of condition [F(2, 724.3)=37.1, p<.001, η^2 =.73], and significant interactions between condition and attention [F(2, 712.5)=9.3, p<.001, η_p^2 =.03] as well as between condition and group [F(2, 25610.1)=3.2, p<.05, η^2 =.063].

Post hoc tests investigating the significant effect of condition in the model above found that, for the groups combined, there were significant differences between multiple conditions in the attended and unattended conditions. In the attended conditions, the mean reconstruction accuracy scores for Single Talker and Eng-Traffic were significantly higher than the other conditions; in the unattended streams reconstruction accuracy for English-Traffic was on average higher than either English-Music or English-Babble.

4.4.4 Effect of condition in monolingual and bilingual scores separately

Following the approach taken in analysing EEG1 experiment, to investigate the condition by group interaction, and test how responses might differ between the groups, I next analysed the distribution of reconstruction scores in monolingual and bilingual groups separately. In *monolinguals*, a model including attention (two levels, attended and unattended), condition (three levels) and their interaction, participant age, parental SES, plus subjects and items as crossed random effects showed significant effects of condition [F(2, 721.94)=22.9, *p*<.001; η^2 =.78] and a condition:attention interaction [F(2, 710.94)=6.39, *p*<.01; η_p^2 =.02]. In *bilinguals*, the equivalent model showed a significant effect of condition only [F(2, 727.88)=24.52, *p*<.001, η_p^2 =.06]. Table 22 summarises the scores in language group by condition.

Table 22

	Mono	olinguals	Bilinguals		
Condition	Attended	Unattended	Attended	Unattended	
Single Talker	.085 (.17)		.073 (.16)		
Eng - Babble	.034 (.16)	.051 (.2)	.032 (.16)	.047 (.19)	
Eng - Traffic	.074 (.17)	.067 (.19)	.07 (.17)	.064 (.18)	
Eng - Music	.065 (.15)	.054 (.2)	.05 (.16)	.039 (.2)	
Overall	.064 (.17)	.057 (.2)	.057 (.16)	.05 (.19)	

Reconstruction accuracy scores for monolinguals and bilinguals by condition

Figure 21

Attended and unattended reconstruction accuracy scores (r) by condition and group



4.4.5 Reconstruction accuracy of attended streams in monolinguals and bilinguals

A key result emerging from EEG Experiment 1 was the difference in patterns of monolinguals and bilingual groups' reconstruction accuracy scores of attended streams across conditions. This indicated that bilingualism does indeed modulate the neural encoding of attended speech envelopes in children, when ignoring linguistic distractors. To see if this is also true of conditions using non-linguistic stimuli as the distracting stream, I investigated the patterns of reconstruction across attended conditions in each group separately. In monolinguals, a model with four levels of attended condition, participant age, parental SES, and subjects and items as random effects, showed a significant effect of condition [F(3,485.45)=31.1, p<.001, η_p^2 =.16] only, with the post-hoc t tests showing differences in encoding between all attended conditions (the Single Talker condition showing significantly higher encoding than those of English-Traffic [t=2.04, p<.05, d=.062], English-Music [t=4.18, p<.001, d=.13] and English-Babble [t=10.1, p<.001, d=.31]; the English-Traffic condition showing significantly stronger encoding relative to the conditions of English-Music [t=2.07, p<.05, d=.06] and English-Babble [t=8.19, p<.001, d=.24]; and the English-Music condition having significantly higher reconstruction accuracy scores than the English-Babble condition [t=6.56, p<.001, d=.19]).

The equivalent analysis in *bilinguals* showed a comparable, but slightly reduced pattern of differences between conditions, with a significant effect of condition $[F(3,486.24)=21.44, p<.001, \eta_p^2=.12]$, reflecting differences between all attended conditions with the exception of Single Talker and English-Traffic. Hence the Single Talker condition showed significantly higher encoding than those of English-Music [t=4.83, p<.001, d=.15] and English-Babble [t=8.35, p<.001, d=.26]; the English-Traffic condition showed significantly stronger encoding relative to the conditions of English-Music [t=4.08, p<.001, d=.13] and English-Babble [t=7.52, p<.001, d=.23]; and the English-Music condition showed significantly higher reconstruction accuracy scores than the English-Babble condition [t=3.66, p<.001, d=.11]).

4.4.6 Reconstruction accuracy of unattended streams in monolinguals and bilinguals

In both groups, the English-Traffic condition differed from English-Babble and English-Music; with no other differences between conditions emerging. Hence for monolinguals, the English-Traffic condition was significantly higher than both English-Babble ([t=2.78, p<.05, d=.084] and English-Music ([t=2.26, p<.05, d=.067]; and the same pattern emerged for bilinguals with English-Traffic showing significantly higher scores than the conditions of English-Babble ([t=3.03, p<.01, d=.092] and English-Music ([t=4.35, p<.001, d=.13].

4.4.7 Comparison of scores by condition between groups

One of the key findings from EEG Experiment 1 was significantly higher indices of encoding in conditions of low or no interference in monolinguals relative to bilinguals. To confirm whether comparable effects may be seen in conditions of nonverbal interference, the next analysis directly compared the reconstruction accuracy between the groups in each attended condition separately. These pairwise comparisons for individual conditions showed significantly higher attentional encoding in monolinguals than in bilinguals in the Single Talker and English-Music conditions [t(4359)=2.37, p<.05, d=.072; and t(4426)=3.17, p<.01, d=.095; respectively], but no difference between the groups in the other two non-linguistic interference conditions [English-Babble t(4201)=.4, p=.69; English-Latin t(4413)=.87, p=.51].

This directly replicates the results of EEG Experiment 1, indicating that, despite the similar patterns within groups, monolinguals show higher levels of attentional encoding in Single Talker and English-Music conditions relative to bilinguals. A similar exercise for the unattended streams revealed significantly higher encoding of

unattended envelopes in the English-Music condition for monolinguals compared to bilinguals [t(4441)=2.5, p < .05, d=.075]. The results of differences between the groups for attended and unattended conditions are summarised in Figure 22.

Figure 22

Differences between the groups for attended and unattended conditions



In sum, the results from this study showed that behavioural comprehension scores were equivalent between groups and conditions, replicating the findings in EEG Experiment 1. Regarding the neural data, patterns in the attended reconstruction scores differed slightly between the groups; with monolinguals showing significant variation between all four conditions. Critically however, a direct comparison between attended conditions in the two groups showed significantly lower indices of neural encoding in bilinguals in two of the attended conditions (Single Talker and English-Music) replicating the pattern found in EEG Experiment 1.

4.5 Discussion

Building on the evidence in EEG Experiment 1 that bilingualism modifies the neural mechanisms of selective attention in school-age children performing a dichotic listening task, the current experiment investigated the extent to which this might generalise to conditions of nonverbal interference. Using a variation of the auditory task, I assessed the patterns of responses to different types of non-linguistic interference in monolingual and bilingual children aged 7-12; comparing their behavioural comprehension scores and their cortical tracking of attended and unattended speech envelopes via reconstruction accuracy scores computed by the mTRF toolbox. The participants in both groups displayed equivalent behavioural performance, replicating the results seen in EEG Experiment 1. While the overall effects of attention on neural encoding, evident in Experiment 1, were not confirmed in this study; results nevertheless showed differences in the way monolinguals and bilinguals encoded attended speech, supporting the hypothesis from EEG Experiment 1 that the processing demands of bilingualism shape the supporting neurocognitive architecture (Green & Abutalebi, 2013), and extending the scope of findings from linguistic interference to include non-linguistic interference. More specifically, results again revealed significantly lower indices of selective attention for bilinguals in conditions similar to those observed in EEG Experiment 1 (Single Talker and English-Music). These results present further evidence that instead of enhanced attentional capacity, these neuroadaptive modifications appear to reflect its redistribution, arguably aimed at economising the available resources to support optimal behavioural performance. I discuss these results in more detail below.

In terms of behavioural comprehension scores, the results clearly showed that all children performed the task equally well, with equivalent ceiling scores across groups and stories. None of the distractors interfered with comprehension in the target story, in contrast to the English-English linguistic condition in Experiment 1, which resulted in both the lowest attended reconstruction scores and the lowest comprehension scores. In this study, the lowest attended reconstruction scores were in the English-Babble condition, but this did not impact comprehension, which was comparable to all the other conditions, for both groups.

This is again consistent with the consensus from the literature that the information presented to the attended ear is typically processed accurately (Cherry, 1953;Treisman, 1969). It also aligns with studies that have found equivalent behavioural performance between bilingual and monolingual children performing a non-linguistic selective attention task (Morton & Harper, 2007; Gathercole et al., 2014; Antón et al., 2014). Furthermore, equivalent ceiling performance adds further weight to the view that any observed neural adaptation – caused as a result of the habitual processing demands of an additional language - serves the purpose of facilitating optimal performance. Hence this is consistent with the proposal that any observed bilingual adaptation is in service of a degenerate system, in which structurally diverse components perform the same function or achieve equivalent outcomes.

The analysis of the neural data focused on reconstruction accuracy of speech envelopes in different attended and unattended conditions from the EEG signal, an established index of neural tracking. As reviewed in the Methods chapter, both children and adults typically encode speech through synchronisation of cortical activity to the temporal oscillations of the speech envelope (Obleser & Kayser, 2019; Lalor & Foxe, 2010). This relationship between signal and speech envelope has been calculated previously using linear cross-correlation (Olguin et al., 2018); however, consistent with EEG Experiment 1, I used the stimulus reconstruction approach implemented in the mTRF toolbox (Crosse et al., 2016). This method has been established in the literature (Power et al., 2016; Kalashnikova et al., 2018; ; Attaheri et al., 2020, Phelps et al. 2022), as a reliable index of neural tracking which offers enhanced sensitivity to all neural input simultaneously.

4.5.1 The role of attention

The synchronisation between cortical activity and speech envelopes is widely assumed to be influenced by selective attention, with numerous studies (including EEG Experiment 1) showing preferential tracking of the attended stream over the ignored one (Rimmele et al., 2015; Horton et al., 2014). In contrast to the literature and EEG Experiment 1, the results for this study showed a weak effect of attention, and even one condition (English-Babble) in which the unattended reconstruction accuracy scores were significantly higher than the ones shown for the target narrative in English. One possible explanation of these anomalous results is that preferential encoding, linked to intelligibility (Millman et al., 2014; Peelle et al., 2013), is necessary in conditions of linguistic interference, such as a classic cocktail party design, but may not be a prerequisite of processing the attended stream when the stream to be ignored is non-linguistic. Indeed, there was no reduction in comprehension scores in the Babble condition despite the apparent preferential tracking of the unattended stream. What is more, the Babble stimulus, although selected as an example of nonverbal noise, could be reasonably characterised as an exception as it contained the (unintelligible) speech of multiple talkers, which is not only derived from linguistic sources, but also widely recognised as "one of the most challenging noise interference for all speech systems" (Krishnamurthy & Hansen, 2009), a challenge which increases with the number of speakers featured (McAuley et al., 2020; Pollack & Pickett, 1958) and is arguably extremely difficult for school-age children to ignore irrespective of cognitive ability (Nagaraj et al., 2020). This makes it in some respects analogous to the English-English condition in EEG Experiment 1, which also generated the lowest reconstruction accuracy scores in the attended stream for both groups. However, comprehension suffered in EEG Experiment 1 in the English-English condition relative to the other conditions, whereas in this study comprehension in all conditions was equivalent, including the English-Babble condition.

However, another possible explanation for these inconsistent results, which do not conform to the usual effect of attention, might be to do with the issues with the acquisition of data for this second EEG experiment. Due to the restrictions enforced during the Covid-19 pandemic, data acquisition was severely disrupted with a substantial gap in the recruitment and testing of participants between March 2020 and July 2021. Furthermore, once testing resumed in 2021, it was conducted in a different EEG suite, with alternative equipment (albeit of similar specification). These disruptions may have impacted the uniformity of the EEG data collected, making it inherently noisy.

4.5.2 Encoding across conditions

In further comparisons with EEG Experiment 1, analysis of the patterns across conditions again showed that attentional encoding across conditions differed between the monolingual and the bilingual children. In the monolingual group in Experiment 1, a prominent contrast between all four attended conditions was observed; yet this effect was somewhat attenuated in the bilingual group (Chapter 4, Figure 8). The differential patterns of encoding in monolingual and bilingual listeners observed in the current study replicates not only the results from EEG Experiment 1 (Phelps et al., 2022), but also the results found in adults (Olguin et al., 2019), adding further support to the hypothesis that bilingualism modifies the neural mechanisms of selective attention across the lifespan (Comishen et al., 2019; Krizman et al., 2014; Garbin et al., 2010). In the Introduction, I presented two accounts that might explain the possible mechanisms of this modification. The first was that the need for constant management and inhibition of competing languages in bilinguals enhances their capacity for selective attention, resulting in better performance and increased attentional control (Bialystok, 2015). The second was that these demands of selection and inhibition will themselves utilise some of the existing attentional resources, which might impact on the available attentional capacity and require that the remaining resources are optimised in order to achieve full task performance. The results of the current study again showed no evidence for the enhanced attentional

capacity, behaviourally or neurally, in the bilingual group. In contrast, results showed weaker neural encoding in bilinguals overall for both attended and unattended streams in Experiment 2 (r_{attd}=.057 for bilinguals vs r_{attd}=.064 for monolinguals; and r_{unattd}=.0499 for bilinguals vs r_{unattd}=.057 for monolinguals), differences which were statistically significant [overall attended scores between groups: t(17577)=3.07, p<.01, d=.046; overall unattended scores between groups: t(13226) = 2.23, p<.05, d=.039]. This is consistent with the findings in EEG Experiment 1 for significantly weaker bilingual scores, relative to monolinguals, in both attended and unattended conditions (see Chapter 4 for detailed analysis). As noted before, the observed indication of reduced cortical encoding overall in bilinguals is not without a precedent, with examples of reduced neural activity during nonverbal selective attention tasks most commonly seen in the cortical areas associated with conflict processing. Lower patterns of activation have been observed in bilinguals performing a Flanker task (Abutalebi et al., 2012), Stroop-like switching task (Garbin et al., 2010), Simon task, and both of the tasks above with the addition of the Simon task (Kousaie & Phillips, 2012). These observations led the authors to conclude explicitly that the bilinguals were resolving the demands of the tasks with less neural resource (Abutalebi et al., 2012).

Of additional note, in EEG Experiment 1 comparison of the groups' pattern of response showed that significant differences between their reconstruction accuracy scores of attended streams occurred in the conditions of no interference (Single Talker) or low interference (English-Musical Rain). In this study, a similar trend was observed for significantly different reconstruction accuracy scores between the groups in the conditions of Single Talker (no interference) and English-Music. In Experiment 1, the significant differences between the groups in the Single Talker and English-Musical Rain conditions were interpreted as an adaptation in the conditions of lowest interference, and hence easiest comprehension; in which the bilingual's more limited capacity processor could economise attention without compromising comprehension; whereas the monolingual attentional system, with greater bandwidth, would have no such need for economisation.

Clearly the Single Talker condition in EEG Experiment 2 can also be defined as a low interference condition, being exactly the same as the condition presented in EEG Experiment 1. However, it is also plausible that Music represented a lowinterference stream, making it analogous to the Musical Rain condition in EEG Experiment 1. First, the melody selected for the Music condition was highly predictable and rhythmic. Not only is there evidence that neural oscillations synchronise with the pulse-rate of music in addition to the rhythms and stress-rate of speech (Haegens & Zion Golumbic, 2018), even when these rhythms are below the perceptual threshold (Oever et al., 2017); but that altered background rhythm in an interference stream has been found to facilitate entrainment and improve comprehension of a target speaker stream (McAuley et al., 2020). Second, the melody was in a major key and a 'feel-good' tune. There is evidence from fMRI (Fernandez et al., 2019) that background music which is happy and high-arousing enhances attention through greater activation of fronto-parietal regions and improved performance in attention tasks, whereas sad and low-valence music elicits slower responses and greater occipital recruitment. Both of these factors support a proposition that the music selected for this task would contribute towards an enhanced attentional capability in the target stream notwithstanding any other constraints. This would explain why monolinguals showed elevated indices of neural attention in both the attended and unattended streams of this condition, relative to bilinguals.

However, EEG Experiment 1 also posited that the linguistic conditions of English-English and English-Latin generate equivalent reconstruction scores between the groups precisely because they cause stronger interference. The two conditions in this study (English-Traffic and English-Babble) with a similar effect on scores cannot be said to be analogous to linguistic interference spoken by a single voice. It could be argued that Babble is derived from linguistic sources and is therefore the most similar interference stream to the linguistic distractor streams in EEG Experiment 1 and thus similarly generates the lowest reconstruction accuracy scores for the attended stream.

Another interpretation, however, is based on the inherent content and structure of the Babble and Traffic streams, which are continuous environmental sounds devoid of rhythm or predictability. As discussed above, rhythm enhances entrainment; it is plausible that a non-rhythmic and unpredictable stream of interference actively diminishes it. This explanation, however, only works in the context of the differences between the groups' scores and does not resolve the finding that reconstruction accuracy scores for the attended stream in the English-Traffic condition are higher in each of the groups than the scores for the target stream in the English-Music condition. One reason for this unexpected finding could be that children are instructed to be on high alert for traffic noise from an early age, but this explanation is as yet lacking in evidence and may not be entirely convincing.

The final set of findings concerned the pattern of reconstruction accuracy scores seen for the unattended streams. These results showed that, in both groups, the unattended Traffic stream was significantly better reconstructed than the unattended Babble and Music streams, similar to the pattern found in the attended streams. In addition, the Music stream encoding was significantly stronger in the monolingual than in the bilingual group. This second result again replicates the finding in EEG Experiment 1 where there was a significant difference in the groups' unattended scores for the English-Musical Rain condition.

In sum, the current study extended the investigation in EEG Experiment 1 - which explored the mechanisms underlying the modification of selective attention in bilingual children – by adding nonverbal interference to the unattended stimuli that participants were asked to ignore. Again, equivalent behavioural performance was observed and there was no evidence for enhanced attentional capacity in the bilingual group. Results in EEG Experiment 1 revealed a modified pattern of neural encoding that showed significant differences between the groups in conditions of weak or no interference. This study also showed significant differences in the condition of no interference, replicating the results from EEG1, and the condition of rhythmic interference, which arguably enhances encoding of the target stream (McAuley et al., 2020), and represents the nonverbal interference most similar to the

stimulus of Musical Rain used in EEG Experiment 1. Therefore these data are interpreted as supporting the conclusion of EEG Experiment 1 that the available resources are economised to support optimal behavioural performance; potentially suggesting increased flexibility of their usage in response to the demands of bilingual language processing. This account is consistent with theories of degeneracy (Navarro-Torres et al., 2021), in which different patterns of neural processing still lead to equivalent behavioural outcomes.

Furthermore, this study showed that the effects observed in EEG Experiment 1 – of equivalent comprehension, lower indices of attentional resource in bilinguals, and different patterns of tracking in the attended streams – are not only applicable in dichotic listening tasks using speech and linguistic interference; but that similar, but attenuated, patterns can be observed when the interference streams are entirely non-linguistic. This aligns with other studies, which have found evidence of modification to the bilingual attentional system when performing nonverbal tasks (Luk et al., 2010; Kousaie & Phillips, 2012; Bialystok et al., 2005; Abutalebi et al., 2012). In this study, however, the distractors used were naturalistic environmental sounds that children of this age group encounter in everyday life, making the study ecologically valid and differentiating it from the designs featuring manufactured nonverbal tasks (Simon, Flanker, ANT etc.) which have typically been used. Hence this study confirms, using naturalistic non-linguistic stimuli, that modification to neural processing in bilinguals as a result of linguistic demands, generalises beyond the linguistic domain and can be observed in nonverbal tests of selective attention.

Chapter 7: Summary and Conclusions

The research presented in this thesis aimed to address three distinct but interrelated research questions:

- 1. Establish whether neural and behavioural modifications triggered by bilingualism are discernible in childhood.
- Determine if the increased processing demands of bilingualism enhance attentional capacity, or lead to an economisation of limited attentional resource, in order to optimise performance; and examine any influences exerted by maturation and language experience.
- 3. Investigate the effect of these modifications on behaviour beyond the language domain.

In this concluding chapter I review the results of all the experiments and draw overall conclusions on the effects of bilingualism on selective attention and influence of maturation and language exposure. I also compare the findings to some of the studies presented in Chapter 2 to assess where this research fits with the broader literature. Finally, I suggest theoretical implications and future directions.

7.1 Neural and behavioural modifications to selective attention in bilinguals– summary of findings and conclusions

The aim of the research in this thesis was to investigate the effects of bilingual modifications to selective attention, in a series of behavioural and neural experiments, with children and adults, using linguistic and nonverbal stimuli, in auditory and visual domains. Each of these variations was based on the same core assumptions. The first was that the additional processing demands of bilingualism (due to parallel activation, Green, 1998), lead to neuroplastic change, which has been documented across the lifespan (Bialystok, 2017; Kovács & Mehler, 2009; Poulin-Dubois et al., 2011; Costa et al., 2008). The second was that bilingualism

specifically influences the neurocognitive architecture of selective attention, with an effect on attentional processing (Olguin et al., 2019; Garbin et al., 2010) across tasks ranging in complexity, but with conflicting evidence on the behavioural consequences of these neural adaptations. The third is that these attentional modifications take place in the context of finite selective attention capacity, i.e., a limited pool of attentional resources (Kahneman, 1973), which can only process a restricted amount of information at any given point (Broadbent, 1965; Clark & Dukas, 2003).

7.1.1 Experiment 1

The first experiment, addressing research question 1), and described in Chapter 4, investigated neural tracking of natural speech under different conditions of linguistic and non-linguistic interference, for monolingual and bilingual children. The accuracy of neural tracking was computed by correlating the EEG response with the attended and unattended speech envelopes, using the mTRF toolbox (Crosse et al., 2016), resulting in reconstruction accuracy scores per participants, per sentence in each condition. In each condition, the target stream was a narrative in English. Distractors ranged in intelligibility (no stream; competing narrative in English; competing narrative in unknown language (Latin); and acoustic stream (Musical Rain). Behavioural performance was also measured through comprehension scores of each target narrative. Behavioural performance was equivalent between the groups. Neural results substantiated the 'selective entrainment hypothesis' (Schroeder & Lakatos, 2009; Zion Golumbic et al., 2013; Giraud & Poeppel, 2012b) with robustly higher scores for attended than unattended streams in all conditions across both groups. Additionally, responses varied according to the type of interference presented, also consistent with the existing literature (Olguin et al., 2018; Har-shai Yahav & Zion Golumbic, 2021).

Most importantly for the questions in this thesis, monolingual and bilingual participants showed different patterns of neural tracking of the attended stream, with monolinguals displaying substantially more variability between attended conditions, and higher scores overall. The bilingual pattern of reconstruction accuracy scores for attended conditions, by comparison, was lower and flatter. These results suggest three conclusions: first, that modifications as a result of bilingualism affect auditory selective processing in the developing brain before continuing into adulthood (Olguin et al., 2019 found similar patterns in adults). Secondly, the lower and flatter scores for bilinguals arguably represent a reduced measure of selective attention, raising the possibility that the management of another language limits overall capacity (Wickens, 2007; Dornic, 1980). Crucially, this did not affect comprehension scores, which were equivalent between the groups. Therefore the third conclusion is that any modifications to neural processing occur in order to mitigate the limitations in attentional capacity that occur as a result of bilingualism, and support optimal performance. This equivalent performance, despite different neural patterns, is an example of degeneracy, in which structurally different components generate comparable outcomes (Navarro-Torres et al., 2021).

7.1.2 Experiment 2

The second experiment, addressing research question 2), and described in Chapter 5, built on the conclusions of Experiment 1, by testing how the bilingual processing system adapts when the demands of a task increase, and seeing if performance across tasks reveals any more about the economisation and prioritisation of attentional resource.

The study used a modified version of the listening task from Experiment 1 (with the same conditions of interference), coupled with a simple visual task (T.O.V.A.), in a dual-task paradigm, designed to stretch attentional load beyond the boundaries of a typical primary task. The intention was that, by increasing the processing demands

in a controlled way, results would reveal if there are any indicators of additional interference in the bilingual group, as well as the expected PRP (psychological refractory period, a well-documented delay as a result of simultaneous task processing) for all participants as a result of the "central bottleneck" (Pashler, 1994). The prediction was that performance in the primary (auditory) and secondary (visual) task would reveal any prioritization of attention to competing task demands. Such additional demands might disproportionately affect bilinguals, with reduced attentional capacity due to managing their additional language processing demands. This reduction in attentional capacity might cause bilinguals to prioritise and economise their responses to competing task demands, to maintain task performance.

In order to assess effects across age groups and language experience, I recruited three groups spanning age ranges and L2 exposure, with one group of school-age children (Study 1, age range 7-12), and two groups of adults (Studies 2 and 3, age range 18-45), who were differentiated by exposure to L2 (English). Study 2 recruited a high-exposure bilingual group and Study 3 a low-exposure bilingual group, as well as monolingual adults matched by age in both studies.

Consistent with Experiment 1 (Chapter 4), the results showed equivalent performance in the auditory (primary) task between monolingual and bilingual participants, across all three studies. In addition, accuracy in the visual task, measured by error rates, was extremely high and equivalent between monolingual and bilingual participants, in all studies.

Reaction times from the visual task, however, revealed significant differences in speed (but not accuracy) between conditions, age groups and language groups. As expected, all groups showed a psychological refractory period (PRP), the delay produced as a result of processing competing task demands, when the auditory task

was added to the visual task. Thus the Control condition (visual only) was significantly faster for all groups of participants across all three studies, confirming a robust psychological refractory period (PRP), consistent with the literature (Pashler, 1994).

In addition, bilingual children's reaction times were significantly slower relative to monolingual children's in all the dual-interference conditions (reaction times for children were equivalent in the Control condition, which was the visual task alone). This result, however, was not replicated in either of the adult groups, with both the high-exposure and low-exposure bilinguals showing statistically equivalent performance to their monolingual counterparts in both the primary (auditory) and secondary (visual) tasks, with one exception (Musical Rain interference), in which low-exposure bilinguals were faster than their monolingual peers in the visual task. When the high and low exposure bilingual adults were compared directly, however, differences emerged between them, revealing that across every single dual interference condition, low-exposure bilinguals were significantly faster than highexposure bilinguals.

These results suggest the following conclusions: In children, the constraints on attention as a result of bilingualism appear to cause an economisation and prioritisation of attentional resource to meet a hierarchy of demands - in this case comprehension in the auditory task, and accuracy in the visual task - at the expense of speed in the visual task. In adults, however, any such limitations are overcome with maturation, resulting in equivalent performance between monolingual and bilinguals adults across all tasks, even ones that exceed 'typical' processing demands (Moray, 1967; Gopher, 1986). Nevertheless, comparisons of the two groups of adult bilinguals with differing language experience suggest that the effect of maturation on selective attention proposed above is enough to close any gap between monolingual and bilingual speakers; but that there is still some residual processing cost for high-exposure bilinguals that, although not apparent relative to
monolinguals, is exposed in a statistically significant reduction in speed of response when compared to low-exposure bilinguals who do not face the same daily processing demands.

These findings add more nuance to the conclusions from Experiment 1 by suggesting a developmental trajectory for bilinguals in a dual-task paradigm. Results suggest that apparent limitations of attentional resource, while still not impacting on the primary task, are shown in scores of the secondary task, in conditions of dual interference in children. Importantly, however, these effects on response times withdraw as the attentional system fully matures. In addition, this also adds more nuance to the conclusion in Experiment 1 that bilingualism creates a degenerate attentional system. Results from Experiment 2 indicate that there is a difference between performance in a primary task and dual task in childhood. Thus, findings indicate behavioural performance on a primary task is indeed equivalent from childhood (as concluded from Experiment 1 and the sample of children in the dual task experiment). However, the results from Experiment 2 also suggest that attentional systems impacted by bilingualism may not reach equivalence under the increased demands of *dual tasks*, which extend beyond a normal processing load, until adulthood. Importantly, this occurs when the prefrontal cortex associated with selective attention processing is fully mature (Giedd et al., 1999; Gogtay et al., 2004; Sowell et al., 2001; Tsujimoto, 2008).

Additionally, however, comparison between bilinguals with different language exposure reveals some residual processing constraints (shown in reaction times in the secondary task), which although not large enough to differentiate bilinguals significantly from monolingual peers, are large enough to differentiate high and low exposure groups of bilinguals. In fact, low-exposure bilinguals showed the overall fastest reaction times of all the language groups across all the cohorts in Experiment 2. This result of low-exposure L2 speakers outperforming high-exposure bilinguals has precedent in the literature, for example in a study comparing monolinguals, fully

immersed language learners and classroom learners, the classroom learners outperformed both the monolinguals and immersed language learners in tasks of inhibitory control, such as the Simon task (Linck et al., 2008). Additionally, greater L2 proficiency was associated with worse inhibitory control, confounding the anticipated outcome that greater proficiency and immersion would lead to better inhibitory control. In Experiment 2 of my thesis, the high-exposure bilinguals could also be categorised as immersed bilinguals, as they were all recruited as UK residents but with L1 in a language other than English. Hence they had daily exposure (full immersion) to their L2 (English). In the conclusions for Experiment 2 at the end of Chapter 5, I suggested that the fact that low-proficiency bilinguals outperformed their monolingual and high-proficiency peers might be used as an incentive in educational settings to promote learning a second language to a reasonable degree of proficiency, which will yield benefits in selective attention tasks with multiple demands even when the L2 will not subsequently be used on a frequent basis. Similarly, in the study of immersion and proficiency of L2 described above (Linck et al., 2008), the authors conclude that "even L2 learners with a late age of acquisition may benefit cognitively from learning the L2".

7.1.3 Experiment 3

The third experiment, addressing research question 3), and described in Chapter 6, replicated the design and target sample of Experiment 1, but changed the interference conditions to exclusively non-linguistic streams, in order to assess whether the same patterns of attentional processing emerge for monolingual and bilingual children in a non-verbal context. Only one of the conditions was the same as in the first study (Single Talker) and the others were changed to naturalistic environmental sounds that would be encountered by a typical school-aged child (traffic, classroom babble and music). Once again there was no difference in comprehension by language group. However, there was no confirmation of the 'selective entrainment hypothesis', with scores for attended and unattended streams equivalent except in the English-Babble condition in which children showed

significantly higher scores for the *unattended* stream (classroom babble) than for the attended narrative in English. Critically, this did not affect comprehension, which was as accurate for both groups in the English-Babble condition as for the narratives in all other conditions. One interpretation of this finding is that children find classroom chatter extremely distracting (as teachers can attest, but also confirmed by Krishnamurthy & Hansen, 2009 and Nagaraj et al., 2020), but have developed cognitive strategies to overcome this specific interference and maintain optimal comprehension.

In similar patterns to Experiment 1, bilinguals showed less variation across the conditions relative to monolinguals, and significantly lower scores for both attended and unattended conditions overall. When examining attended conditions specifically, bilinguals had significantly lower scores than monolinguals in two conditions (Single Talker and English-Music) which were either the same or similar to the attended conditions with largest differences between the groups in Experiment 1 (Single Talker and English-Musical Rain). This significant disparity between the groups is interpreted in both EEG Experiment 1 and behavioural Experiment 2 as bilinguals' economising attentional resource in the conditions of lowest interference (and hence easiest comprehension); with such redistribution unnecessary for monolinguals whose attentional system is not constrained by the demands of an additional language. In this context, music was explained as a lower interference than traffic or babble due to its inherent rhythmic and predictable qualities.

Notwithstanding the above, in the conditions of interference, the highest attended and unattended scores for both groups were in the English-Traffic condition. This was a puzzling result, with a tentative explanation that children might be primed to pay extra attention to all stimuli when encountering traffic in real life. The results from Experiment 3 suggest the following conclusions: Bilingual children display consistently lower indices of attentional encoding in non-verbal as well as linguistic conditions of interference, adding further evidence to the hypothesis that attention is constrained and economised in bilinguals, but redistribution enables optimal (equivalent) behavioural performance. Furthermore, equivalent performance despite disparities in neural indices further corroborates the account that bilingualism creates a degenerate system capable of supporting optimal behaviour from its neural modifications. Finally, the similar patterns of results in EEG Experiments 1 and 3 indicate that neural modifications are not only apparent for cases of linguistic processing but are also applicable in (this design, naturalistic) nonverbal contexts. There was also the anomalous finding of a lack of preferential tracking of the attended stream (and even preferential tracking of the unattended stream in the English-Babble condition). One possible interpretation of this finding is that the 'selective entrainment hypothesis' may only be applicable to competing speech streams as it was not corroborated when the streams were non-linguistic. However, it must also be noted that a key limitation of this study was the extensive interruption when collecting data, due the national lockdown during the Covid-19 pandemic, which not only affected the timeline of this study but the location and equipment used. These disruptions may have contributed to some of the inconsistent results.

Overall, however, Experiment 3 corroborated the key conclusions from Experiment 1 and Experiment 2. Therefore, the overall conclusions across all three experiments can be summarised as: bilingual children and adults display equivalent behavioural performance in a primary task across both linguistic and non-verbal contexts. Bilingual children, meanwhile, show lower and flatter patterns of neural activation, indicating attentional constraints as a result of managing an additional language. These neural disparities do not affect optimal performance, indicating a degenerate system. However, when the processing load is increased, bilingual children show performance disparities, relative to monolinguals, in the elements of the task with the lowest priority. These disparities disappear by adulthood, although some differences remain between bilinguals with different language exposure, indicating differential adaptation of attention in bilinguals, according to maturation and language experience.

7.2 Conclusions in the context of the literature

The conclusions above are relevant to the current discourse in the field of bilingualism and selective attention, with many aspects either being consistent with, or building on, findings from some of the studies in the literature reviewed in Chapter 2. I will take each of the conclusions and in turn, and compare or contrast to relevant findings in the literature:

1. Bilingual children and adults display equivalent behavioural performance in a primary selective attention task across both linguistic and non-verbal contexts

Despite the assertion that bilingualism creates a 'bilingual advantage' in both linguistic (Kaushanskaya & Marian, 2009; Antoniou et al., 2015; Canbay, 2011; Filippi et al., 2015) and non-linguistic tasks (Kovács & Mehler, 2009; Poulin-Dubois et al., 2011; Costa et al., 2008; Bialystok et al., 2012), there is also ample evidence of equivalent behavioural performance across the lifespan (Filippi et al., 2020; von Bastian et al., 2016; Samuel et al., 2018; Jones et al., 2021; Nichols et al., 2020).

The equivalent comprehension scores in the dichotic listening task used as the primary task in this thesis were replicated across five studies (two EEG experiments and three behavioural dual task studies), four of which used linguistic interference, and one non-verbal. The results contrast with studies featured in the Literature Review such as the sentence comprehension test (Filippi et al., 2015) and tones in noise study (Krizman et al., 2017), which both found behavioural differences between monolinguals and bilinguals under different conditions of interference in a listening task (albeit different designs). However, other dichotic listening tasks have also revealed no behavioural differences between monolinguals and bilinguals

(Oliveira et al., 1997) in different age groups (Desjardins et al., 2020). A related design combining auditory comprehension and visual interference, accuracy in a listening and picture naming task (Blumenfeld & Marian, 2011), revealed no difference between the language groups when listening to words and ignoring the interference of visual cues. These inconsistencies may reflect that, as presented in the Literature Review in Chapter 2, factors such as language exposure, age of acquisition, proficiency and language similarity have influenced studies' results. All of the children recruited across all three experiments for this thesis had a similar language profile: all were highly proficient early bilinguals who spoke English at school and a different language at home. The adults, however, differed by language profile and still maintained the same comprehension. One additional consideration is that the dichotic listening task was designed to be engaging and comprehensible, using children's stories, as opposed to designs using tones in noise (Krizman et al., 2017) or syntactically challenging sentences (Filippi et al., 2015). Thus the participants in these studies equated on behaviour in a design more akin to natural speech.

One consequence of using a highly naturalistic and engaging story with simple comprehension questions, was that behavioural performance was consistently extremely high and averaged 98.42% for monolinguals and 98.44% for bilinguals across the five studies. The auditory behavioural scores therefore are not normally distributed and represent behavioural performance at ceiling. My method of analysis of comprehension scores took this into account, by using a glmer model analysis in R, with binary scores as the dependent variable and factors of group, condition, age SES and subjects as a random effect, in order to provide detailed insight of any variance within this upper range.

One finding that emerged from these analyses was significantly lower comprehension scores in the English-English condition in Experiment 1, demonstrating the relative difficulty of ignoring the distractor in this condition. What is more, this mean decline in comprehension scores was experienced equally across both language groups, indicating that the processing of competing known languages was equivalently effortful for both monolingual and bilingual speakers. Importantly, the variability of scores in this condition did not correlate with age (r=.15; t=1.06; p=.29).

There is clearly a possibility that a more difficult comprehension task would have revealed more variance in scores across conditions. However, a more difficult comprehension test would have arguably also made more demands on working memory or lexical retrieval than on comprehension. The development of working memory is highly correlated with age for children until adolescence (S. E. Gathercole et al., 2004). Given that in all studies the task was administered to children as young as 7, it is plausible that any increase in task difficulty would reveal a performance variable with age, rather than provide insights into processing differences due to language background. Similarly, a potentially poorer performance by bilinguals that might have arisen in a more difficult task could have emerged due to problems with lexical retrieval and limitations in bilingual vocabulary, or the "distributed characteristic" (Oller et al., 2007; Pearson et al., 1993) rather than differences in processing per se. As it would have been difficult to isolate comprehension from working memory and lexical retrieval on a more difficult task, especially in the 7-12 age range, the originally designed task has arguably proved to be valid and informative for the questions I aimed to assess.

2. Bilingual children show lower and flatter patterns of neural encoding, indicating attentional constraints as a result of managing an additional language.

The results in Experiment 1 and Experiment 3 of different patterns of response in attended conditions for bilinguals and monolinguals are consistent with an associated study with an adult sample (Olguin et al., 2019). This study, based on a dichotic listening task with different levels of interference, also found greater variability in monolinguals' correlation to the attended stream, while bilinguals showed a much flatter pattern with comparable levels of neural tracking in all attended conditions. In other study designs, lower measures of attentional encoding

overall in bilinguals have also been observed, with examples of reduced neural activity during various selective attention tasks, observed predominantly in the cortical areas associated with conflict processing. Functional imaging during a Flanker task performed by bilinguals and monolinguals (Abutalebi et al., 2012) revealed significantly lower patterns of activation in the anterior cingulate cortex (ACC) for bilinguals, leading the authors to conclude that 'bilinguals...resolve cognitive conflicts with less neural resource'. A similar fMRI study of a Stroop-like switching task (Garbin et al., 2010) also found that monolinguals activated the ACC during the task, whereas bilinguals did not. An ERP study tracking bilingual and monolinguals' neural responses during a variety of selective attention tasks (Kousaie & Phillips, 2012), predicted superior performance (greater accuracy and faster reaction times) and larger N2 amplitudes for bilinguals relative to monolinguals. On the contrary, behaviour was equivalent between the two groups; and the monolingual group exhibited larger N2 amplitudes than the bilingual group during the Stroop task. In the Simon task, expectations were reversed when monolinguals demonstrated larger P3 amplitudes than bilinguals. In the Flanker task, higher ERN (error-related negativity) amplitudes for bilinguals than monolinguals, instead of corroborating enhanced cognitive control, were due to a longer tail for incongruent trials, indicating a prolonged post-response conflict and slower recovery for bilinguals in these trials.

Another study used fMRI to measure brain activity in passive listening and picturenaming tasks with two groups: monolinguals and early highly-proficient bilinguals (Palomar-García et al., 2015). The fMRI results revealed that monolinguals and bilinguals displayed different patterns of activity, leading to the conclusion that monolinguals were displaying more efficient use of language networks because bilinguals 'utilised a more distributed network, which may imply subtle processing disadvantages' (Palomar-García et al., 2015). Such a conclusion is analogous to the lower indices of attentional encoding in the EEG studies of Experiments 1 and 3 being interpreted as evidence of attentional constraints due to greater demands on bilingual attentional resource. In sum, the interpretation of lower attentional resource is consistent with accounts of lower activation, flatter amplitudes and more distributed processing, from other neural studies.

3. These neural disparities occur alongside optimal behavioural performance, indicating a degenerate system

There are several examples in the literature of equivalent performance despite neural differences between monolinguals and bilinguals. Once again, the dichotic listening study coupled with EEG (Olguin et al., 2019) showed the same result with adults. Other examples include the MEG study to investigate neural differences for a group of monolinguals and two groups of bilinguals performing a Simon task (Bialystok, Craik, et al., 2005). The behavioural results of the Simon test showed no difference between the groups, and analyses of the MEG data showed activity in the left and medial prefrontal areas for all three groups. However, the two bilingual groups showed activity in selected left hemisphere regions, whereas the monolingual group showed activity in the middle frontal regions. This finding was replicated in other tasks such as the flanker (Luk et al., 2010) which used fMRI data to compare activation. Although there was no significant difference between the language groups' response times, a difference was found in the pattern of neural activity between the groups. Similar studies using ERP (Kousaie & Phillips, 2012) (Kousaie & Phillips, 2017) found no behavioural bilingual advantage in groups of young adults in the Stroop, flanker and Simon tasks; and no difference in groups of young and older adults in the Stroop task, but ERP data (2017) revealed different patterns of activation for monolinguals and bilinguals in the tasks.

As per the parallel drawn in the Literature Review chapter, these neural differences in spite of equivalent behavioural performance are also demonstrated in the influence of environmental variables other than bilingualism. SES is a prime example, with a study showing equivalent behavioural performance (comprehension) in a listening task but different magnitudes of response to attended and unattended probes according to groups of higher or lower SES, when age was matched (Hampton Wray et al., 2017). The above examples – of comparable performance despite neural differences – are examples of how environmental variables create degenerate systems. As discussed in the Introduction, degeneracy in the scientific definition comprises structurally diverse components leading to the same output or performing the same function, arguably a desirable characteristic, making systems robust and enabling natural evolution (Whitacre & Bender, 2010; Edelman & Gally, 2001). Degenerate systems are functionally plastic (Mason et al., 2015), therefore the neuroplastic changes effected by bilingualism can be seen as the formation of a degenerate system (Green et al., 2006; Mason et al., 2015; Navarro-Torres et al., 2021). One interpretation discussed in the preceding chapters is that the neural modifications generated by bilingualism may exist precisely in order to overcome the attentional constraints imposed by the management of the additional language and hence create a system that, while showing structural or processing differences, achieves full behavioural performance.

4. However, when the processing load is increased, bilingual children show performance disparities, relative to monolinguals, in the elements of the task with the lowest priority.

There does not appear to be a precedent in the literature comparing monolingual and bilingual children in a dual task paradigm. All the examples cited in Chapter 5 feature adult samples and show inconsistent results, some attributed to poorer lexical access in bilinguals. Therefore this represents a wholly new finding and interpretation. 5. These disparities disappear by adulthood, although some differences remain between bilinguals with different language exposure, indicating differential adaptation of attention in bilinguals, according to maturation and language experience.

There are many studies in the literature indicating differences in bilingual performance as a result of variables such as age of acquisition, proficiency or language exposure, usually revealing that early, highly-proficient bilinguals or those fully immersed, show higher performance across age groups (Durand López, 2021; Bonfieni et al., 2019; Struys et al., 2015; Kaushanskaya & Marian, 2007). In this study, however, low-exposure bilinguals outperformed high-exposure bilinguals in response times to the visual task. This does have some precedent in the literature, namely the study described above comparing monolinguals, fully immersed language learners and classroom learners (Linck et al., 2008), in which the classroom learners outperformed both the monolinguals and immersed language learners in tasks of inhibitory control, such as the Simon task. Additionally, greater L2 proficiency was associated with worse inhibitory control, in contradiction to the studies above showing that greater proficiency and immersion lead to better inhibitory control. In another study, an ANT (attentional network) test performed by early and late bilinguals revealed better conflict resolution by the late bilinguals, indicating age of acquisition is not necessarily linked to better performance in all aspects of selective attention tasks (Tao et al., 2011). Therefore, enhanced performance in some tasks by bilinguals with either lower exposure/immersion (as in Experiment 2), or later age of acquisition, has been found before, and is usually attributed to reduced processing difficulty in code-switching or lexical access (Linck et al., 2008).

In terms of bilingual vs monolingual performance across age groups, Experiment 2 found a bilingual disadvantage in children for speed in the secondary task (despite equivalent comprehension in the primary auditory task and accuracy in the visual task), which disappeared by adulthood. Many of the studies indicating a trajectory in bilingual performance report the opposite effect – i.e., a bilingual advantage in

childhood, which tends to disappear in young adulthood (and then reappears in late adulthood, albeit a finding not relevant to the studies in this thesis). This is often attributed to a "cognitive ceiling" in which young adults have developed peak cognitive ability (Valian, 2015b). In a study comparing four age groups of bilingual and monolinguals participants (Bialystok, Craik, et al., 2005), the interpretation of equivalent behavioural results between monolinguals and bilinguals in the young adult group is that the undergraduate group is already operating at peak cognitive efficiency, (unlike children still in development, or adults in cognitive decline as the result of ageing), which explains the lack of demonstrable bilingual advantage. Other studies have found no difference at all between language groups at different ages (Filippi et al., 2020). The bilingual difference in a secondary task in childhood, which disappears by adulthood, is unprecedented in the literature but, as stated above, there does not appear to be an existing study of monolingual and bilingual children performing a dual-task paradigm against which to compare and contrast results.

7.3 Implications and future directions

7.3.1 Theoretical implications

In the main, the findings from my research corroborate various findings from similar studies in the literature. There are, however, novel findings that add nuance to existing theoretical positions.

One key finding is that equivalence in behavioural performance for both monolinguals and bilinguals, in a primary task, applies to both children and adults; however, this changes when the task is modified beyond what normal attentional capacity can process. In this case, bilingual children reveal deficits in performance of the lowest priority task (speed in the secondary task, in this case), which are overcome by adulthood. This has implications for considerations about the limits of flexible adaptation to the environmental processing demands (i.e., the concept of degeneracy), which can be further broken down into a) a degenerate system for processing within typical boundaries or b) a degenerate system for *all* processing, even demands beyond the typical load. It also implies that degeneracy itself may have a developmental trajectory, i.e., the system in children is degenerate to the point of equivalence in a primary task, but complete degeneracy is not reached until full maturation of the cognitive system.

This is a plausible account from the perspective of both theories of educational psychology (Winne & Nesbit, 2010) and a neuroscientific account of learning (Goswami, 2008) in which the cognitive architecture adapts based either on experience-dependent or maturational factors to form a learning trajectory (Westermann et al., 2006). In the case of bilingualism and selective attention, as a child matures, the interrelation between developmental changes in the brain and cognitive abilities follows a trajectory in which attentional processing reaches its peak in adulthood, and fully accommodates the existing demands of bilingualism.

The second interpretation that contrasts with some existing theoretical positions, is that bilingualism does not lead to an enhanced attentional capacity and therefore a bilingual advantage on selective processing tasks; rather, the lower and flatter neural indices indicate a constrained attentional capacity that needs to 'do more with less' to reach behavioural parity with monolinguals. This interpretation adds a competing account to the dominant view in the literature of 'enhanced' attention (Costa & Sebastián-Gallés, 2014; Bialystok, 2017; Timmer et al., 2021; Comishen et al., 2019; Bialystok, 2015; Kapa & Colombo, 2013) which leads to superior behavioural performance on tasks requiring attentional control. Importantly, however, the interpretation of doing 'more with less' still allows an account of increased flexibility. Thus recruiting fewer neural resources to achieve tasks in no way precludes bilingual optimal performance, but goes some way to explaining the inconsistent behavioural results in the literature, and may mark out a middle way in the heated and polarised debate of a bilingual advantage in certain tasks.

7.3.2 Future lines of enquiry

There are clearly limitations of the studies described in this thesis, which should be addressed by future experiments. The first was that the distractor stream of Musical Rain in EEG Experiment 1 was derived from the same narrative as the target story, resulting in extremely similar speech envelopes, which had the potential to artificially inflate the unattended reconstruction scores in the English-Musical Rain condition. Future studies using a similar design and conditions can avoid this by creating the Musical Rain from a completely unrelated narrative.

A further challenge was the recruitment of a high quality bilingual adult sample on the online dual task study. The adult sample was recruited via Prolific, an online recruitment platform, which sources participants globally and invites a selection to participate according to pre-screening criteria. One advantage of using online recruitment platforms such as Prolific is that they have large, globally-distributed databases of potential participants, allowing researchers to access a larger and more diverse pool of potential participants in a short time, which increases speed of recruitment and may strengthen the ecological validity of the resulting sample. However, critics of online recruitment have raised issues of self-selection biases and lack of context (Dewaele, 2018). In addition, such platforms are known to suffer from potential problems with recruiting genuine participants who will perform the study or task in good faith (Downs et al., 2010), and controlling for participant characteristics when eligibility criteria are self-reported (Chandler & Paolacci, 2017). Issues with fraudulent self-selection, such as participants claiming incorrectly to be eligible for studies, are more pronounced in more specific populations (such as highexposure bilinguals) and are exacerbated by the relative anonymity of online recruitment (Franzen, 2023). Controlling for this risk is extremely important in future studies using similar designs, and can be counteracted through the use of stringent recruitment criteria and careful monitoring of participants throughout the study. A rapid response rate to the study can be slowed down by releasing participant places in batches. This allows time to monitor responses, filter out unsuitable candidates and calibrate the criteria for ongoing recruitment as the study progresses.

Lastly, due to the COVID-19 pandemic, my testing period for EEG Experiment 2 suffered from a protracted interruption in data collection. This possibly led to inconsistent results, due to data being collected in different locations and with different equipment after a 15-month hiatus. A future study could replicate the non-linguistic experimental design and see if it garners more uniform results with consistent testing in one window of time and using the same equipment throughout.

In addition, there are alternative future directions related to these studies. A first potential future direction would be to investigate the composition of the measures of attention generated in Experiment 1 and Experiment 3. The reconstruction accuracy scores used as an index of neural tracking in the EEG experiments were an overall measure, averaged across channels and frequencies (delta, theta, beta, gamma). However, studies have also revealed details of neural oscillations in different frequency bands, which can detail the encoding of different types of lowerlevel features of the signal (e.g. Di Liberto et al., 2015 analysed the processing of phonemes in specific frequency bands). Hence the scores broken down by frequency band for bilinguals and monolinguals would reveal information contained in different frequencies that might have been obscured to date by an averaged measure.

The second future direction could extend results from measuring the processing of lower-level, perceptual information; to the processing of higher-level semantic and syntactic information contained in the speech streams. This can be achieved using similar methods to the EEG experiments in this thesis (using the mTRF toolbox); but changing input to assess how semantic and syntactic vectors computed from the stimulus content are encoded in the neural activation. In addition, this could also be further analysed in different frequency bands (see Direction 1) as previous research suggests that different functions in the processing of speech, such as higher-level semantic and syntactic cues, engage distinct neural components (Keitel et al., 2018; Ding & Simon, 2014), giving an overall score and more detailed breakdown.

In a bilingual context, the above would drill down into the modifications in bilingual processing of the speech signal and test whether they are applicable to all aspects of language processing. It would also reveal more specific differences between monolinguals and bilinguals by frequency band, revealing the composition underlying the overall scores that have been computed so far.

Finally, these studies have focused on children and young adults. Previous studies on age groups across the lifespan have revealed age-related differences in both performance (Bialystok, Martin, et al., 2005) and in neural activation and architecture (Tao et al., 2021). Therefore, in the third future direction, I suggest further investigating the impact of bilingualism on cognitive processing and maturation, by expanding the participant age range to older adults. The concept of "cognitive reserve", and the potential delay to cognitive deterioration as a result of bilingualism, is an area to which the detailed EEG and behavioural studies presented in this thesis could be applied next. This would not only reveal how bilingualism interacts with aging and cognitive decline; but also, when combined with the studies in this thesis, build a fuller picture of the impact of bilingualism on selective attention at different stages across the lifespan.

Thus future directions could build on the research presented in this thesis by investigating how bilingual adaptation affects neural response across different frequency bands, in different linguistic processing contexts (both lower-level and higher-level features), and across the lifespan.

In summary, the results of the experiments presented in this thesis indicate that, in the developing brain, bilingualism modifies the neural mechanisms that support selective auditory attention to natural speech, in conditions of both linguistic and non-linguistic interference. This does not affect behaviour, suggesting bilingualism creates a degenerate system, in which neural modifications lead to equivalent performance on a primary task in childhood. Lower neural indices, however, indicate that bilinguals have constraints on the extent of attentional capacity to meet task demands, due to the inhibition of the omnipresent second language. This

manifests as a limitation on the performance of 'extreme' tasks, with attention economised to meet the demands of highest priority, at the expense of aspects of a secondary task. Any such differences in behaviour on a dual task, relative to monolinguals, dissipates with maturation; although residual processing costs are revealed when adult bilinguals with different L2 exposure are compared. This combination of experiments, bridging research on cognitive development, selective attention, bilingualism and degeneracy, provides new insights into how the developing mind adapts to the demands of bilingualism on finite selective attention capacity and processing of linguistic and non-verbal attention tasks.

Appendix A

Transcripts of linguistic stimuli

Attended Narratives

A. THE BOY AND BIRDS

- 1. Once in Russia, there lived a merchant and his wife.
- 2. Their only son was called Peter.
- 3. Peter was a kind-hearted boy who loved animals.
- 4. He owned a little pet bird in a cage.
- 5. The bird sang beautifully every day.
- 6. Peter wished he knew the meaning of its song.
- 7. One winter, it snowed very heavily.
- 8. Peter was out for a walk in the snow
- 9. When he spotted some baby birds in a tree.
- 10. They were desperately tweeting with cold.
- 11. He took pity on them and fetched a blanket from his room.
- 12. He climbed the tree and put the blanket over the birds.
- 13. They snuggled under it and tweeted happily.
- 14. The next day he received a visit from the birds' mother.
- 15. "Young man" she said "you are so kind
- 16. I would like to reward you with my magical powers.
- 17. Tell me what is your dearest wish?"
- 18. Peter thought hard and then said:
- 19. "I would like to be able to understand birdsong".
- 20. "Very well," said the mother bird with magical powers
- 21. I will teach you the language of the birds".
- 22. So for the next month she gave Peter lessons
- 23. On the tunes, the words and the grammar of the birds.
- 24. He was a quick learner and soon could understand everything.
- 25. Including what his own pet bird was singing.
- 26. To his astonishment, it sang of his future
- 27. It said that Peter would one day be rich
- 28. What is more, he would be a prince
- 29. And his father would be his servant.
- 30. Peter found this very puzzling.
- 31. He would be rich and his father a servant!
- 32. Peter's parents wondered what was bothering him.
- 33. So they asked him what on earth was wrong.
- 34. Peter made the mistake of telling his parents
- 35. The future his pet bird had predicted.
- 36. And they were not pleased at all oh no!

- 37. In fact, they decided he had gone quite mad
- 38. And the best thing to do was get rid of him
- 39. While he was sleeping, they took him down to the seashore
- 40. Then they pushed him out in a little wooden boat
- 41. He quickly drifted out across the wide sea
- 42. And his parents supposed they would never see him again.
- 43. But Peter's little boat was spotted by a ship
- 44. And they rescued him and brought him on board.
- 45. The next morning, he heard some birds flying past.
- 46. "Quick! Quick!" they were squawking to each other
- 47. "A terrible storm in on its way!"
- 48. Peter warned the sailors about the storm
- 49. But they laughed and did not believe him
- 50. Then the storm did come, and it was as fierce as he'd said.
- 51. The next week he heard some geese flying past
- 52. They were also squawking excitedly
- 53. "Those pirates over there are waiting!
- 54. They will attack the next ship that goes past."
- 55. Peter went to warn the ship's captain.
- 56. And this time everyone believed Peter.
- 57. They sailed in the opposite direction as fast as they could.
- 58. And warned every ship that they passed.
- 59. That some pirates were lying in wait
- 60. And so Peter became very useful.

<u>Block 2</u>

- 1 He stayed on the ship for some years
- 2 And warned the crew about dangers ahead
- 3 Always from listening to the birds
- 4 Some years later, he heard about a king
- 5 Who was being bothered by three pesky crows.
- 6 These noisy birds sat on his window sill
- 7 And cawed at him all hours of night and day.
- 8 His servants had tried to shoo them away
- 9 But the birds refused to leave the king's window
- 10 The king was so annoyed by these birds
- 11 That he promised that the person who got rid of them
- 12 Would have half his kingdom as reward!
- 13 Since that proclamation, many people had tried
- 14 But nobody had succeeded in making the crows go away
- 15 The king was beginning to despair
- 16 "Aha" thought Peter. "This is my chance!"
- 17 He made his way to the castle and said
- 18 "Give me a chance, your majesty, to rid you of these crows"
- 19 He was shown to the window where the birds were sitting.
- 20 He listened for a few minutes and then spoke:
- 21 "There are three crows a father, mother and son.
- 22 The mother and father are getting divorced.
- 23 They cannot agree who the son should live with
- 24 So they want to ask the king to decide

- 25 And will not leave until they have heard his ruling".
- 26 The king thought carefully for a moment
- 27 Finally he proclaimed: "Crow family
- 28 I think it is best for the son to stay with the...
- 29 Mother!" and with a great CAW!
- 30 All three birds flew away and left him in peace.
- 31 So Peter then received the reward
- 32 Of half the king's riches and kingdom
- 33 He stayed on to live at the castle
- 34 And advise the king on all matters.
- 35 The king's daughter was full of admiration
- 36 And before long she and Peter fell in love
- 37 They were married, and so now Peter
- 38 Was rich and a prince as his bird had sung.
- 39 Meanwhile, Peter's mother had fallen ill and died.
- 40 His father was old and had lost all his money
- 41 After pirates had attacked his merchant ships
- 42 And stolen all his goods and profits.
- 43 Peter's father decided to go to the king.
- 44 He came to the castle to beg for a job.
- 45 He said, "Your majesty, I will do anything
- 46 Let me be your humble servant
- 47 And work in the castle for your family.
- 48 Now Peter's father had grown very old
- 49 And his eyesight was a bit blurry
- 50 Besides, many years had now passed
- 51 And Peter was a fully-grown man, not a boy
- 52 So he didn't recognise his own son
- 53 Standing there behind the king as chief advisor
- 54 But Peter recognised him and said:
- 55 "Father! It is I, Peter, your son.
- 56 Come stay with me and my family.
- 57 And let us look after you in your old age."
- 58 For Peter was still kind-hearted.
- 59 He looked after his father from that day on.
- 60 And so it was, the bird's song came true.

B. THE TORTOISE AND THE KING

<u>Block 1</u>

- 1. Many centuries ago, there was a King in India.
- 2. He had been king for many many years
- 3. He was a good, kind king, but had one fault
- 4. His fault was this: he talked too much!
- 5. He loved nothing better than the sound of his own voice.
- 6. He had lived a long time, seen many things,
- 7. read many books, and met many fascinating people.
- 8. But his thoughts rambled, and he talked on and on.
- 9. Everyone knew that if you asked him a question

- 10. You would never get a clear and short reply.
- 11. If one of his generals were to ask him what to do
- 12. About difficult soldiers in the army
- 13. he would begin, "What would my grandfather do?
- 14. When I met the old commander 35 years ago,
- 15. he told me that an army marches on its stomach,
- 16. And that reminds me of another thing..."
- 17. and then would launch into an unrelated story.
- 18. By the time he had finished telling the story,
- 19. he had quite forgotten what the question was!
- 20. The king had a beautiful palace
- 21. With wonderful neatly kept gardens all around
- 22. One day, the weather was fine and sunny
- 23. And the king decided to go for a nice walk
- 24. He went for a stroll around his beautiful garden
- 25. But in the garden of his palace, he found a tortoise.
- 26. It was lying on the path and when the king looked closer
- 27. He saw that the poor creature was well and truly dead.
- 28. It had obviously come to a very violent end
- 29. As its shell was smashed into several pieces.
- 30. The king was puzzled by this strange discovery,
- 31. for he was sure that the dead tortoise had a hidden meaning,
- 32. but he struggled to think what it might be.
- 33. "It must be a sign from the gods!" he said
- 34. "But I have no idea what it means!
- 35. I will ask if anyone knows in the palace."
- 36. So one by one, everyone came out
- 37. And looked upon the poor dead tortoise
- 38. But nobody could say what it meant.
- 39. At last, the king sent a messenger
- 40. To the oldest and wisest man in the land,
- 41. and asked him to take a look at the tortoise
- 42. and explain the meaning of this very strange sign.
- 43. On seeing the tortoise, the wise man nodded
- 44. He said, "Master, let me tell you a story.
- 45. This story will explain everything to you
- 46. Once upon a time there lived an old tortoise
- 47. He lived by the side of a beautiful lake.
- 48. He spent his days resting in the shade of the grass,
- 49. and nibbling on the juicy weeds that grew around the lake.
- 50. Lots of animals lived there together.
- 51. And there was enough food and drink for everyone.
- 52. But one year there was no rain all summer
- 53. There was a drought up and down the land.
- 54. The hot sun baked the land drier and drier.
- 55. The weeds and grasses shrivelled away.
- 56. The water in the lake dried first into puddles
- 57. And then started to disappear altogether.
- 58. The animals started to starve.
- 59. There was nothing to eat or drink.
- 60. They were getting weaker and weaker every day.

- 1 The tortoise was friends with some geese nearby.
- 2 One day, two of the geese came to say goodbye,
- 3 for their flock was planning to fly to the mountains
- 4 far, far away to the north of the country
- 5 in search of water and fresh juicy grass.
- 6 "And what will become of me?" asked the tortoise.
- 7 "I am 100 years old, my shell is heavy,
- 8 and my legs are short and stumpy.
- 9 If I plodded all day I would reach no further
- 10 than that boulder that you see over there.
- 11 "I must stay here until I am all dried up.
- 12 Oh, woe is me! What can I do?
- 13 Please take me with you, I beg of you."
- 14 The two geese took pity on the old tortoise
- 15 And they thought hard about how to help him
- 16 They made a plan so he could get away from the lake:
- 17 They would hold a stick between them as they flew.
- 18 The tortoise must hold on to the stick with his mouth
- 19 and they would transport him to a cooler place.
- 20 Off they set the very next morning.
- 21 The tortoise found himself flying through the skies.
- 22 He felt quite sick but was determined to hold on.
- 23 The other geese in the flock thought it was hilarious.
- 24 "Hey look," said one, "I've seen everything now.
- 25 A tortoise has grown wings. Ha ha ha!".
- 26 "That's not what's happened," said another,
- 27 "He's chewing on a stick because he's hungry."
- 28 Then a third goose joined in the teasing.
- 29 "I'm sure he would whizz along a lot faster
- 30 If he didn't have that heavy shell", he said.
- 31 The comments continued, as the geese flew.
- 32 Because they had nothing else to talk about
- 33 Except the tortoise and his unusual travel mode.
- 34 The tortoise could hear every word they said
- 35 He was getting more and more annoyed.
- 36 Eventually the tortoise could stand it no more.
- 37 "See here," he said, "Do you think I'm enjoying this?"
- 38 And that, of course, was when he opened his mouth,
- 39 let go of the stick, and started to fall.
- 40 He fell, and he fell, until he hit the ground,
- 41 Splat! in the garden of your palace, my lord.
- 42 And that is why you have this dead tortoise before you."
- 43 That was the end of the wise man's tale.
- 44 In reply, the king began to talk at length
- 45 about the hibernation patterns of tortoises.
- 46 The wise man listened patiently without interrupting.
- 47 He knew the king had not yet understood his story.
- 48 When the opportunity arose to speak politely

- 49 the wise man walked once again carefully
- 50 around the body of the poor dead tortoise
- 51 and said these following lines very clearly:
- 52 "Oh gracious king, remember well
- 53 To speak wisely and to speak in season
- 54 For to his death the tortoise fell
- 55 He talked too much: that was the reason."
- 56 Suddenly, the king understood what he meant.
- 57 The gods had sent him a sign to not talk so much.
- 58 And from then on, he was careful to think
- 59 Before he opened his mouth to speak.
- 60 And that is how a tortoise stopped a chatterbox king!

C. THE DUMPLING

<u>Block 1</u>

- 1. LONG, LONG ago there lived a woman
- 2. She was an old woman and funny,
- 3. who liked to laugh and to make dumplings to eat.
- 4. One day, while she was preparing some dumplings,
- 5. One fell off the table, then it rolled away.
- 6. It rolled across the floor, through the door and..
- 7. Plop! fell down a hole, and disappeared.
- 8. The old woman looked down the hole
- 9. Then she tried to reach the dumpling by putting her hand down.
- 10. But suddenly the earth started to shake
- 11. With a roar the ground gave way beneath her
- 12. And the old woman fell down the hole.
- 13. She fell down and down and slid all the way to the bottom.
- 14. She wasn't hurt and still wanted to find her dumpling.
- 15. When she got up on her feet again, she was standing on a road.
- 16. It was in a strange land with rolling hills all around.
- 17. Now I can't even tell you how this had happened.
- 18. But the funny old woman had fallen into another country!
- 19. The road she was on went winding down a steep hill
- 20. So she thought that the dumpling must have rolled down.
- 21. She ran down the road to look, saying:
- 22. "My dumpling, my dumpling! Where is my dumpling?"
- 23. After a little while she saw a stone statue.
- 24. The woman stared at the statue and said:
- 25. "Excuse me, did you see my dumpling roll past?"
- 26. The statue answered: "Yes, indeed I did!
- 27. But don't go down the hill to look for it,
- 28. for there is a wicked giant living down there.
- 29. And he will capture you if you go any further."
- 30. But the old woman took no notice of the statue.
- 31. She carried on running down the road saying
- 32. "My dumpling, my dumpling! Where is my dumpling?"
- 33. Soon, she spotted another stone statue.
- 34. The woman asked this statue the same question:

- 35. "Excuse me, have you seen a dumpling roll past?"
- 36. "Yes", replied the statue, "I saw your dumpling.
- 37. But please don't go down any further
- 38. Because a terrible giant lives down there
- 39. Who will take you prisoner if you go near his house."
- 40. But the woman just laughed at the statue.
- 41. Then the statue whispered: "Here comes the giant.
- 42. Sit behind me, and don't make any noise."
- 43. Presently the giant came very close,
- 44. and stopped and bowed to the statue.
- 45. "Good day Mr Statue!" the giant said.
- 46. Then the giant sniffed the air suspiciously,
- 47. And said, "Do you smell what I smell?
- 48. I smell the smell of a nice juicy woman!"
- 49. "Oh no, you are mistaken" said the statue.
- 50. "There's no woman around here."
- 51. But the giant carried on sniffing. Sniff! Sniff!
- 52. "I can definitely smell the smell of a woman!" he said.
- 53. The old woman thought all this was very funny
- 54. And she couldn't help laughing. "Te he-he!"
- 55. The giant immediately reached down his big hand
- 56. and pulled her out, still laughing, "Te-he-he!"
- 57. "Aha!" cried the giant, excitedly.
- 58. "I've got you! I knew I could smell a woman.
- 59. You smell of dumplings and lovely food.
- 60. I will take you home with me to be my cook."

- 1 Then the giant took the old woman down the hill.
- 2 At the end of the winding road was his house.
- 3 It was a big house with lots of huge rooms.
- 4 He led her through to the kitchen, and said
- 5 "This is the kitchen, and you are now our chef
- 6 "Cook some dinner for me and the other giants who live here!"
- 7 The old woman started to get a bit worried
- 8 How was she going to cook enough food for the giants?
- 9 She only knew how to cook for herself.
- 10 However, the giant had not finished.
- 11 He gave her a small wooden spoon, and said:
- 12 "Only put one grain of rice into the pot,
- 13 Then add a little bit of water to the pot
- 14 when you stir that one grain of rice with this spoon,
- 15 the grain will multiply until the pot is full."
- 16 So the woman put just one rice-grain into the pot,
- 17 She then added a little splash of water
- 18 and began to stir it with the wooden spoon; .
- 19 one grain turned to two, then two to four
- 20 And so on and so on and so on.
- 21 Every time she moved the spoon the rice doubled;
- 22 and in a few minutes the great pot was full.

- 23 The old woman was delighted with the magic spoon.
- 24 It made cooking for the giants quick and easy.
- 25 The funny old woman stayed a long time in the house
- 26 and every day cooked food for the giant and his friends.
- 27 The giant never hurt or frightened her,
- 28 and her work was made quite easy by the magic spoon,
- 29 although she did have to cook a LOT of rice,
- 30 because a giant eats much more than any human being eats.
- 31 But she felt lonely, and always wished to go home,
- 32 and make her own tasty dumplings again.
- 33 She missed her own little kitchen
- 34 And wanted to see all her friends again.
- 35 One day, when the giants were all out somewhere,
- 36 She noticed they had left the huge gates unlocked.
- 37 So she thought she would try to run away.
- 38 She first picked up the useful magic spoon,
- 39 and slipped it quickly under her belt;
- 40 then she tiptoed through the unlocked gates
- 41 and ran away up the road as fast as she could.
- 42 Back up the hill she went, huffing and puffing.
- 43 Past the two stone statues who were still standing there.
- 44 But this time she didn't stop to chat.
- 45 She never stopped running until she was home again.
- 46 After that she was very happy to be home.
- 47 She was back in her own little kitchen
- 48 She could make dumplings whenever she pleased.
- 49 Plus, she had the magic spoon to make rice for her.
- 50 She started to sell her dumplings and rice.
- 51 And all her neighbours and friends bought them from her.
- 52 She sold so many of her tasty dumplings
- 53 That in the end the old woman
- 54 Was funny.. and old...and very rich.
- 55 She still laughed all day: "Tee hee hee!"
- 56 But from then on if she accidentally dropped a dumpling
- 57 She left it to roll away and didn't chase it.
- 58 And she was always careful around holes
- 59 She never put her hand down one again.
- 60 For the one adventure down a hole was enough!

D. THE SHEPHERD AND THE TIGER

- 1. Once upon a time there was a good-hearted shepherd
- 2. He was out walking alone one fine morning
- 3. When he heard a strange howling noise.
- 4. It sounded like an animal roaring in distress.
- 5. As he got closer, he saw it was a huge tiger.
- 6. The tiger had been caught in a very strong trap.
- 7. And was rattling the cage door furiously.

- 8. He had been in the trap for hours and hours.
- 9. And despite all his pushing couldn't open the door.
- 10. All at once the tiger saw the shepherd.
- 11. He cried: "Please, I beg you, kind sir!
- 12. You look like an honest and kindly gentleman.
- 13. Would you please let me out of this terrible cage?
- 14. I have been trapped all night and cannot get out!
- 15. All you have to do is open this little door here."
- 16. But the shepherd did not trust the tiger.
- 17. He was worried the tiger would gobble him up
- 18. The tiger was big and ferocious
- 19. And the man was unarmed and all on his own.
- 20. The shepherd replied, "Do I look like an idiot to you?
- 21. If I set you free you will eat me in a flash!"
- 22. "No, no, I promise!" said the tiger
- 23. shaking his head, and smiling at the shepherd.
- 24. "If you let me out, I will be eternally grateful
- 25. I will serve you as a slave for ever!"
- 26. Well, the shepherd thought about his proposal.
- 27. It would be pretty cool to have a tiger as a slave.
- 28. Besides, he was now feeling quite sorry for the tiger.
- 29. Who was looking more and more dejected.
- 30. So he went and opened the door of the cage.
- 31. Creeeeak! The door of the cage swung open.
- 32. Out sprang the tiger and grabbed the shepherd.
- 33. "Aha!" he growled, "You foolish man!
- 34. I am extremely hungry after a night in that cage
- 35. So I think I might just gobble you up right now.
- 36. You look nice and tasty as a snack".
- 37. The shepherd pleaded with the tiger for his life.
- 38. "Please please spare me! Let me go free!
- 39. You cannot break the promise you made!"
- 40. Suddenly the shepherd thought of a plan.
- 41. He suggested this to the tiger:
- 42. The shepherd could ask for the opinion
- 43. of the next three objects he came across
- 44. To decide if the shepherd deserved to live.
- 45. If any of them thought that the shepherd was right,
- 46. The tiger would let him go unharmed.
- 47. But if they all agreed with the tiger's view
- 48. That the man deserved to be eaten
- 49. He would gobble the shepherd up straight away.
- 50. "Hmm" said the tiger "all right
- 51. I agree to this little plan of yours
- 52. But make it quick! My stomach's rumbling
- 53. And I'm sure nobody will side with you."
- 54. Now the shepherd had to find something
- 55. That would save him from the tiger
- 56. He started to look around desperately
- 57. For three things that he could ask
- 58. And hopefully one would agree

- 59. That he, the shepherd, should go free
- 60. And not be eaten by the hungry tiger

- 1 So the shepherd looked around for someone to ask.
- 2 He decided to ask first a nearby tree.
- 3 "Trees are old and wise," he thought to himself.
- 4 "Surely the tree will be on my side"
- 5 and he told the tree his sorry story:
- 6 "Oh, Mr Tree, I have been trapped by a tiger
- 7 Who at first promised to set me free
- 8 And now he wants to gobble me up!
- 9 Do you think this is a fair way to treat me?"
- 10 But the tree looked at him coldly and said:
- 11 "What have you got to complain about?
- 12 I give shade and shelter to everyone.
- 13 And they repay me by cutting my branches off!
- 14 Don't be a wimp be a man!"
- 15 This was not the reply the shepherd wanted.
- 16 He next decided to ask the road the same question.
- 17 "Oh, Mr Road, I have been trapped by a tiger
- 18 Who at first promised to set me free
- 19 And now he wants to gobble me up!
- 20 Do you think this is a fair way to treat me?"
- 21 But the road answered unsympathetically:
- 22 "Well, what do you expect, you silly man?
- 23 Here am I, useful to everyone.
- 24 I take everyone where they want to go.
- 25 But they all trample on me as they go past."
- 26 This wasn't what the shepherd wanted to hear either.
- 27 He was now very worried about being eaten.
- 28 He was walking back towards the tiger and the cage
- 29 Trembling and muttering, when up came a monkey.
- 30 "Hey, Mr Shepherd, what's wrong with you?"
- 31 said the monkey. "You look terrible"
- 32 So the shepherd explained what had happened.
- 33 However, the monkey only scratched his head.
- 34 "I am very confused," he said.
- 35 "I cannot understand what has happened.
- 36 What a strange situation this all seems to be
- 37 I would like to ask the tiger some questions
- 38 And then I can make up my mind whether it's fair."
- 39 So they went back to the tiger and the shepherd said,
- 40 "I asked this monkey but he has some questions."
- 41 "Yes", said the monkey "I'm extremely confused.
- 42 Please explain to me exactly how this all started.
- 43 Was the shepherd in the cage when the tiger...."
- 44 "No, no, no" said the tiger "that's wrong!
- 45 You have got it the wrong way around"
- 46 "Aha!" said the monkey, "I've got it!

- 47 You were both in the cage and then....what happened?"
- 48 "Wrong again!" said the tiger. "It was me!"
- 49 "I was the one who was trapped in the cage!
- 50 "Let me show you, you stupid little creature
- 51 I can see that you don't have any brains."
- 52 The tiger hopped into the cage, where he shouted:
- 53 "Look here! I was the one trapped in the cage!
- 54 "NOW do you understand how it all started?"
- 55 "Perfectly!" said the monkey as he shut and locked the door.
- 56 "And now that is also how it ends! Ha ha!"
- 57 And he smiled a big smile at the shepherd.
- 58 Who realised the monkey was not stupid at all.
- 59 And that is the story of how the monkey
- 60 Helped the shepherd escape from the tiger.

Unattended Narratives

E. THE MAGIC RING

<u>Block 1</u>

- 1. There was once a house that I used to visit.
- 2. I knew the way in very well indeed –
- 3. there was a tiny little door at the back,
- 4. so small that you would probably never notice it,
- 5. but I'm a mouse, and I can squeeze through anything.
- 6. Every time I visited, there were fewer and fewer crumbs.
- 7. It was hardly worth going there anymore.
- 8. There was also the problem of a cat in the house.
- 9. This cat was getting skinnier and skinnier,
- 10. and desperate to catch a mouse to eat.
- 11. One night, I couldn't help overhearing an argument.
- 12. The owners of the house were shouting loudly.
- 13. "It's all your fault we're starving," said the woman.
- 14. "You shouldn't have gone and sold my mother's ring.
- 15. All you got for it was an old horse that died.
- 16. Now we've got nothing, nothing."
- 17. "What's your mother's ring got to do with it?" asked the man.
- 18. "Because I told you a thousand times,
- 19. the person who wears that ring will never go hungry."
- 20. "Well I don't believe in that magic and nonsense,"
- 21. replied the man in a huff.
- 22. "You don't believe in anything I say.
- 23. That's why you're so thin your trousers keep falling down."
- 24. Ooh, that was quite an argument.
- 25. But now I understood why the house was getting so poor.
- 26. The man had sold the woman's magic ring
- 27. A ring that kept the cupboards full of food.
- 28. I wish I had a ring like that, I thought.
- 29. Next thing Bam! Everything went black.

- 30. "Am I dead or alive?" I thought.
- 31. But I soon knew that I was alive because I heard a voice.
- 32. The dog, a scratchy smelly creature, was saying:
- 33. "Hey cat, hang on, don't eat that mouse just yet."
- 34. "Why shouldn't I?" whined the cat. "
- 35. I haven't had a decent meal in days."
- 36. "Neither have I," said the dog,
- 37. "But the mouse will only fill you up for a few hours.
- 38. Let's be smart about this, my friend.
- 39. He can help us fill our stomachs for the rest of our days."
- 40. "You've gone mad from hunger," said the cat.
- 41. "This is a mouse, not a hen. He can't lay eggs."
- 42. "No," woofed the dog, "That's not what I mean.
- 43. The thing about a mouse is that he's small..."
- 44. "And he can slip through the tiniest of holes..."
- 45. "So what?" said the cat.
- 46. "And if we take him to the house where the magic ring is now,
- 47. he can slip inside and get it for us.
- 48. He'll do this for you, because otherwise you'll eat him.
- 49. But if he gets the ring, you will set him free.
- 50. The magpie told me where that house is.
- 51. Come on, let's head off straight away."
- 52. The cat saw that the dog was not quite as stupid as he looked.
- 53. She put me in her mouth and jumped through the window.
- 54. The dog followed and they ran down the alleyway.
- 55. It was hot and smelly in the cat's mouth
- 56. I was bouncing around all over the place,
- 57. It was the worst journey of my life
- 58. but I was alive, thank goodness.
- 59. I just had to find the magic ring
- 60. If I wanted to keep it that way.

- 1. We left the town and reached a river.
- 2. The dog said, "Right-oh, in we go!"
- 3. But the cat put me down and held me with a paw.
- 4. "Not so fast," she said, "I can't swim."
- 5. "Never mind that, " woofed the dog,
- 6. "Jump on my back and I'll carry you over."
- 7. So "Splash!" In we all went.
- 8. The dog paddled quickly to the other side
- 9. And over to the house where the ring was to be found.
- 10. Inside we could see a man and woman,
- 11. both plump, with happy expressions on their faces.
- 12. The dog said: "We've come to the right place.
- 13. Let's hope she takes the ring off when she goes to bed."
- 14. The cat prowled round the house looking for a way in.
- 15. There's usually one, if you search hard enough.
- 16. We found a tiny little hole that only I could squeeze through.
- 17. "In you go," she said, "And bring back that ring!"

- 18. "If you bring us the ring you can live!"
- 19. In I went, still glad to be alive, but quite nervous.
- 20. I can sniff out crumbs, but not gold and silver.
- 21. Fortunately, I saw the ring glinting in the moonlight.
- 22. The lady had left it on the table with her other jewellery.
- 23. I put the ring in my mouth, and slipped back out again.
- 24. "Here it is," I said, "I hope you are happy."
- 25. "Say goodbye to this world, little mouse," replied the cat.
- 26. I trembled for my life, but the dog woofed:
- 27. "Stop right there, cat. A promise is a promise."
- 28. We returned to the river, and swam back like before.
- 29. Soon we were back near where they lived.
- 30. This time, the cat went over the rooftops -
- 31. while the dog ran down the alleyways.
- 32. I went home to my nest for a good sleep.
- 33. Now there is a follow up to this story.
- 34. A few weeks later, I was going past the cat and dog's house.
- 35. The light was on, and I looked through the window.
- 36. The human couple were having a feast,
- 37. and the cat was looking fat and happy.
- 38. "So," I thought, "What they said was true.
- 39. The owners of the magic ring will never go hungry."
- 40. I too deserved a reward for my part in the rescue of the ring,
- 41. and I decided to slip inside and take some crumbs.
- 42. But round the back, I found the dog tied up.
- 43. He was still thin and more miserable than ever.
- 44. "What on earth has happened?" I asked,
- 45. "The house is full of food. Why aren't you tucking in?"
- 46. The dog whined. "Aroooo!
- 47. That filthy lying cat played a trick on me!
- 48. She ran to the house before me, taking a shortcut over the rooftops.
- 49. As soon as she got back, she sprang onto our owners' bed
- 50. and woke them up with the great gift of the ring.
- 51. Oh how delighted they were!
- 52. She is rewarded every day.
- 53. But as for me, they punished me.
- 54. They think I'm just a smelly, useless animal,
- 55. and they tie me up outside.
- 56. I'm lucky if they remember to throw me a scrap.
- 57. GRRRRRR! I shall hate that cat till the day I die,
- 58. and so shall all my puppies hate all cats for ever!"
- 59. That's the story of why cats and dogs are enemies
- 60. They have hated each other ever since!

F. THE ADVENTURES OF PERSEUS

LATIN TRANSCRIPT

- 1. Haec narrantur a poetis de Perseo.
- 2. Perseus filius erat Iovis, maximi deorum;
- 3. avus eius Acrisius appellabatur.
- 4. Acrisius volebat Perseum nepotem suum necare;
- 5. nam propter oraculum puerum timebat.
- 6. Comprehendit igitur Perseum adhuc infantem,
- 7. et cum matre in arca lignea inclusit.
- 8. Tum arcam ipsam in mare coniecit.
- 9. Danae, Persei mater, magnopere territa est;
- 10. tempestas enim magna mare turbabat.
- 11. Perseus autem in sinu matris dormiebat.
- 12. Iuppiter tamen haec omnia vidit,
- 13. et filium suum servare constituit.
- 14. Tranquillum igitur fecit mare,
- 15. et arcam ad insulam Seriphum perduxit.
- 16. Huius insulae Polydectes tum rex erat.
- 17. Postquam arca ad litus appulsa est,
- 18. Danae in harena quietem capiebat.
- 19. Post breve tempus a piscatore quodam reperta est,
- 20. et ad domum regis Polydectis adducta est.
- 21. Ille matrem et puerum benigne excepit,
- 22. et iis sedem tutam in finibus suis dedit.
- 23. Danae hoc donum libenter accepit,
- 24. et pro tanto beneficio regi gratias egit.
- 25. Perseus igitur multos annos ibi habitabat,
- 26. et cum matre sua vitam beatam agebat.
- 27. At Polydectes Danaen magnopere amabat,
- 28. atque eam in matrimonium ducere volebat.
- 29. Hoc tamen consilium Perseo minime gratum erat.
- 30. Polydectes igitur Perseum dimittere constituit.
- 31. Tum iuvenem ad se vocavit et haec dixit:
- 32. "Turpe est hanc ignavam vitam agere;
- 33. iam dudum tu adulescens es.
- 34. Quo usque hic manebis?
- 35. Tempus est arma capere et virtutem praestare.
- 36. Hinc abi, et caput Medusae mihi refer."
- 37. Perseus ubi haec audivit, ex insula discessit,
- 38. et postquam ad continentem venit, Medusam quaesivit.
- 39. Diu frustra quaerebat; namque naturam loci ignorabat.
- 40. Tandem Apollo et Minerva viam demonstraverunt.
- 41. Primum ad Graeas, sorores Medusae, pervenit.
- 42. Ab his talaria et galeam magicam accepit.
- 43. Apollo autem et Minerva falcem et speculum dederunt.

- 44. Tum postquam talaria pedibus induit,
- 45. in aera ascendit. Diu per aera volabat;
- 46. tandem tamen ad eum locum venit
- 47. ubi Medusa cum ceteris Gorgonibus habitabat.
- 48. Gorgones autem monstra erant specie horribili;
- 49. capita enim earum anguibus omnino contecta erant.
- 50. Manus etiam ex aere factae erant.
- 51. Res difficillima erat caput Gorgonis abscidere;
- 52. <u>eius enim conspectu homines in saxum vertebantur.</u>
- 53. Propter hanc causam Minerva speculum Perseo dederat.
- 54. Ille igitur tergum vertit, et in speculum inspiciebat;
- 55. hoc modo ad locum venit ubi Medusa dormiebat.
- 56. Tum falce sua caput eius uno ictu abscidit.
- 57. Ceterae Gorgones statim e somno excitatae sunt,
- 58. et ubi rem viderunt, ira commotae sunt.
- 59. Arma rapuerunt, et Perseum occidere volebant.
- 60. Ille autem dum fugit, galeam magicam induit.

- 1. et ubi hoc fecit, statim e conspectu earum evasit.
- 2. Post haec Perseus in finis Aethiopum venit.
- 3. Ibi Cepheus quidam illo tempore regnabat.
- 4. Hic Neptunum, maris deum, olim offenderat;
- 5. Neptunus autem monstrum saevissimum miserat.
- 6. Hoc cottidie e mari veniebat et homines devorabat.
- 7. Ob hanc causam pavor animos omnium occupaverat.
- 8. Cepheus igitur oraculum dei Hammonis consuluit,
- 9. atque a deo iussus est filiam monstro tradere.
- 10. Eius autem filia, nomine Andromeda,
- 11. virgo formosissima erat.
- 12. Cepheus ubi haec audivit, magnum dolorem percepit.
- 13. Volebat tamen civis suos e tanto periculo extrahere,
- 14. atque ob eam causam imperata Hammonis facere constituit.
- 15. Tum rex diem certam dixit et omnia paravit.
- 16. Ubi ea dies venit, Andromeda ad litus deducta est,
- 17. et in conspectu omnium ad rupem adligata est.
- 18. Omnes fatum eius deplorabant,
- 19. nec lacrimas tenebant. At subito,
- 20. dum monstrum exspectant, Perseus accurrit;
- 21. et ubi lacrimas vidit, causam doloris quaerit.
- 22. Illi rem totam exponunt et puellam demonstrant.
- 23. Dum haec geruntur, fremitus terribilis auditur;
- 24. simul monstrum horribili specie procul conspicitur.
- 25. Eius conspectus timorem maximum omnibus iniecit.
- 26. Monstrum magna celeritate ad litus contendit,
- 27. iamque ad locum appropinquabat ubi puella stabat.
- 28. At Perseus ubi haec vidit, gladium suum eduxit,
- 29. et postquam talaria induit, in aera sublatus est.
- 30. Tum desuper in monstrum impetum subito fecit,

- 31. et gladio suo collum eius graviter vulneravit.
- 32. Monstrum ubi sensit vulnus, fremitum horribilem edidit,
- 33. et sine mora totum corpus in aquam mersit.
- 34. Perseus dum circum litus volat,
- 35. reditum eius exspectabat.
- 36. Mare autem interea undique sanguine inficitur.
- 37. Post breve tempus belua rursus caput sustulit;
- 38. mox tamen a Perseo ictu graviore vulnerata est.
- 39. Tum iterum se in undas mersit,
- 40. neque postea visa est.
- 41. Perseus postquam ad litus descendit,
- 42. primum talaria exuit;
- 43. tum ad rupem venit ubi Andromeda vincta erat.
- 44. Ea autem omnem spem salutis deposuerat,
- 45. et ubi Perseus adiit, terrore paene exanimata erat.
- 46. Ille vincula statim solvit, et puellam patri reddidit.
- 47. Cepheus ob hanc rem maximo gaudio adfectus est.
- 48. Meritam gratiam pro tanto beneficio Perseo rettulit;
- 49. praeterea Andromedam ipsam ei in matrimonium dedit.
- 50. Ille libenter hoc donum accepit et puellam duxit.
- 51. Paucos annos cum uxore sua in ea regione habitabat,
- 52. et in magno honore erat apud omnis Aethiopes.
- 53. Magnopere tamen matrem suam rursus videre cupiebat.
- 54. Tandem igitur cum uxore sua e regno Cephei discessit.
- 55. Postquam Perseus ad insulam navem appulit,
- 56. se ad locum contulit ubi mater olim habitaverat,
- 57. sed domum invenit vacuam et omnino desertam.
- 58. Tris dies per totam insulam matrem quaerebat;
- 59. tandem quarto die ad templum Dianae pervenit.
- 60. Huc Danae refugerat, quod Polydectem timebat.

Appendix B

Comprehension questions

A. THE BOY AND BIRDS

<u>Block 1</u>

- 1. This is a story about a boy called PETER/TOM
- 2. Who was very NASTY / KIND
- 3. And looked after some baby BIRDS / FROGS
- 4. He learnt BIRD LANGUAGE / MATHEMATICS
- 5. His own bird said he was going to be POOR / RICH
- 6. His parents were pleased with the news. YES / NO
- 7. His parents put him in a TRAIN / BOAT
- 8. But he was rescued by a SHIP / PLANE
- 9. Peter heard birds saying some SUBMARINES / PIRATES were ahead
- 10. And this time the sailors thought he was RIGHT / WRONG

Block 2

- 1. Peter then STAYED SOME YEARS / LEFT STRAIGHT AWAY
- 2. He then heard about a QUEEN / KING
- 3. Who was having a problem with some CROWS / DOGS
- 4. Who were too QUIET / NOISY
- 5. The king had promised that if anyone solved the problem he would MAKE

THEM RICH / THROW THEM IN PRISON

- 6. So Peter then lived at the CHURCH / CASTLE
- 7. His FATHER / MOTHER came to beg for a job

- 8. Peter IGNORED / RECOGNISED him
- 9. And told him to GO AWAY / STAY WITH HIM
- 10. For Peter was still EVIL / KIND-HEARTED

B. THE TORTOISE AND THE KING

<u>Block 1</u>

- 1. This story is about a KING / QUEEN
- 2. Who DANCES / TALKS too much
- 3. And finds a RABBIT / TORTOISE
- 4. Who is DEAD / ALIVE
- 5. In his SWIMMING POOL / GARDEN.
- 6. He asks a MAN / WOMAN
- 7. Who is FORGETFUL / WISE
- 8. To EXPLAIN / COOK the tortoise
- 9. He tells a story of a tortoise who lives by a SHED / LAKE
- 10. But one year there is no RAIN / LIGHT

- 1. The tortoise is friendly with some MONKEYS / GEESE
- 2. They are going to FLY AWAY / HAVE A PARTY
- 3. The tortoise wants to STAY WHERE HE IS / GO WITH THEM
- 4. The geese AGREE / REFUSE to help him.
- 5. The tortoise has to hold onto a WING / STICK
- 6. With his TAIL / MOUTH while they fly
- 7. The other geese start to SING / TALK
- 8. The tortoise gets very ANNOYED / PLEASED

- 9. Opens his mouth to talk and FALLS DOWN / STAYS IN THE AIR
- 10. Finally the king DIES / UNDERSTANDS

C. THE DUMPLING

<u>Block 1</u>

- 1. This story is about a funny old MAN / WOMAN
- 2. Who liked to make DUMPLINGS / FAIRY CAKES
- 3. One day, one STARTED SPEAKING / ROLLED AWAY
- 4. The woman went TO BED / DOWN A HOLE
- 5. She found herself on a ROAD / BUS
- 6. That went DOWN / UP a hill
- 7. She saw a WIZARD / STATUE
- 8. Who told her not to go down the hill as a DRAGON / GIANT lived there
- 9. The giant came and SMELLED / KICKED the woman
- 10. He heard the woman SCREAMING / LAUGHING

- 1. The giant took the woman to his SHIP / HOUSE
- 2. He showed her the KITCHEN / BATHROOM
- 3. And gave her a magic wooden FORK / SPOON
- 4. To help her cook POTATOES / RICE
- 5. And make the pot FULL / EMPTY
- 6. After that she stayed to CLEAN / COOK for the giant and his friends
- 7. But after a while she wanted to go HOME / TO THE SHOPS
- 8. She LEFT / TOOK WITH HER the magic spoon
- 9. And RAN / SWAM all the way home
10. After that, she SOLD / THREW AWAY her dumplings

D. THE SHEPHERD AND THE TIGER

Block 1

- 1. This is a story about a BEGGAR / SHEPHERD
- 2. Who finds a TIGER / FISH
- 3. In a LAKE / CAGE
- 4. The tiger wants to DANCE / GET OUT
- 5. At first the shepherd AGREES STRAIGHT AWAY / DOESN'T WANT TO
- 6. But the tiger makes him a PROMISE / SANDWICH
- 7. So eventually he SETS HIM FREE / POKES HIM IN THE EYE
- 8. The tiger then wants to KISS HIM / EAT HIM
- 9. But then AGREES / DISAGREES with the shepherd's plan
- 10. That the shepherd can ASK / KICK three objects for their judgement

Block 2

- 1. First of all the shepherd asks a CLOUD / TREE
- 2. Then the shepherd asks the ROAD / SKY
- 3. Both of these call the shepherd a wimp and silly TRUE / FALSE
- 4. Then the shepherd spots a PARROT / MONKEY
- 5. The monkey says he is CONFUSED / ANGRY
- 6. And wants to see THE SHEPHERD'S HOUSE / THE TIGER AND THE CAGE
- 7. The tiger wants to SHOW WHAT HAPPENED / THROW A BALL TO the monkey
- 8. So the tiger RUNS AWAY / GETS BACK IN THE CAGE
- 9. Then the monkey SHUTS THE DOOR / LETS HIM OUT AGAIN
- 10. And that is how the monkey HELPS / SHOUTS AT the shepherd

References

- Abutalebi, J., Canini, M., Della Rosa, P. A., Green, D. W., & Weekes, B. S. (2015). The neuroprotective effects of bilingualism upon the inferior parietal lobule: A Structural Neuroimaging Study in Aging Chinese Bilinguals. *Journal of Neurolinguistics*, *33*, 3–13. https://doi.org/10.1016/j.jneuroling.2014.09.008
- Abutalebi, J., Canini, M., Della Rosa, P. A., Sheung, L. P., Green, D. W., & Weekes, B. S. (2014). Bilingualism protects anterior temporal lobe integrity in aging. *Neurobiology of Aging*, *35*(9), 2126–2133.

https://doi.org/10.1016/j.neurobiolaging.2014.03.010

Abutalebi, J., Cannini, M., Rosa, P. D., Shueng, L., & Weekes, B. (2013). Second Language Naming Predicts Left Temporal Pole Integrity in Bilinguals. *Procedia* - *Social and Behavioral Sciences*, *94*, 41–42.

https://doi.org/10.1016/j.sbspro.2013.09.017

- Abutalebi, J., Della Rosa, P. A., Green, D. W., Hernandez, M., Scifo, P., Keim, R., Cappa, S. F., & Costa, A. (2012). Bilingualism Tunes the Anterior Cingulate Cortex for Conflict Monitoring. *Cerebral Cortex*, *22*(9), 2076–2086. https://doi.org/10.1093/cercor/bhr287
- Abutalebi, J., & Green, D. W. (2016). Neuroimaging of language control in bilinguals: Neural adaptation and reserve. *Bilingualism: Language and Cognition*, *19*(4), 689–698. https://doi.org/10.1017/S1366728916000225
- Abutalebi, J., Guidi, L., Borsa, V., Canini, M., Della Rosa, P. A., Parris, B. A., & Weekes, B. S. (2015). Bilingualism provides a neural reserve for aging populations.

Neuropsychologia, 69, 201–210.

https://doi.org/10.1016/j.neuropsychologia.2015.01.040

- Aiken, S. J., & Picton, T. W. (2008). Human Cortical Responses to the Speech Envelope. *Ear and Hearing*, *29*(2), 139. https://doi.org/10.1097/AUD.0b013e31816453dc
- Albert, M. L., & Obler, L. K. (1978). *The Bilingual Brain: Neuropsychological and Neurolinguistic Aspects of Bilingualism. Perspectives in Neurolinguistics and Psycholinguistics*.
- Ali, N., Green, D. W., Kherif, F., Devlin, J. T., & Price, C. J. (2010). The Role of the Left Head of Caudate in Suppressing Irrelevant Words. *Journal of Cognitive Neuroscience*, *22*(10), 2369–2386. https://doi.org/10.1162/jocn.2009.21352
- Alladi, S., Bak, T. H., Duggirala, V., Surampudi, B., Shailaja, M., Shukla, A. K.,
 Chaudhuri, J. R., & Kaul, S. (2013). Bilingualism delays age at onset of
 dementia, independent of education and immigration status. *Neurology*, *81*(22), 1938–1944. https://doi.org/10.1212/01.wnl.0000436620.33155.a4
- Alladi, S., Bak, T. H., Mekala, S., Rajan, A., Chaudhuri, J. R., Mioshi, E., Krovvidi, R.,
 Surampudi, B., Duggirala, V., & Kaul, S. (2016). Impact of Bilingualism on
 Cognitive Outcome After Stroke. *Stroke*, *47*(1), 258–261.
 https://doi.org/10.1161/STROKEAHA.115.010418

Andrews Espy, K., Molfese, V., & DiLalla, L. (2001). Effects of Environmental
 Measures on Intelligence in Young Children: Growth Curve Modeling of
 Longitudinal Data. *Merrill-Palmer Quarterly*, 47(1).
 https://digitalcommons.wayne.edu/mpq/vol47/iss1/3

Antón, E., Duñabeitia, J. A., Estévez, A., Hernández, J. A., Castillo, A., Fuentes, L. J.,
Davidson, D. J., & Carreiras, M. (2014). Is there a bilingual advantage in the
ANT task? Evidence from children. *Frontiers in Psychology*, *5*.
https://doi.org/10.3389/fpsyg.2014.00398

- Antoniou, M., Liang, E., Ettlinger, M., & Wong, P. C. M. (2015). The bilingual advantage in phonetic learning. *Bilingualism: Language and Cognition, 18*(4), 683–695. https://doi.org/10.1017/S1366728914000777
- Archila-Suerte, P., Woods, E. A., Chiarello, C., & Hernandez, A. E. (2018).
 Neuroanatomical profiles of bilingual children. *Developmental Science*, *21*(5), e12654. https://doi.org/10.1111/desc.12654
- Arredondo, M. M., Hu, X.-S., Satterfield, T., & Kovelman, I. (2017). Bilingualism alters children's frontal lobe functioning for attentional control. *Developmental Science*, 20(3), e12377. https://doi.org/10.1111/desc.12377
- Astheimer, L. B., & Sanders, L. D. (2012). Temporally selective attention supports speech processing in 3- to 5-year-old children. *Developmental Cognitive Neuroscience*, *2*(1), 120–128. https://doi.org/10.1016/j.dcn.2011.03.002
- Astheimer, L., Janus, M., Moreno, S., & Bialystok, E. (2014). Electrophysiological measures of attention during speech perception predict metalinguistic skills in children. *Developmental Cognitive Neuroscience*, 7, 1–12. https://doi.org/10.1016/j.dcn.2013.10.005
- Attaheri, A., Choisdealbha, Á. N., Liberto, G. M. D., Rocha, S., Brusini, P., Mead, N., Olawole-Scott, H., Boutris, P., Gibbon, S., Williams, I., Grey, C., Flanagan, S., & Goswami, U. (2020). Delta- and theta-band cortical tracking and phase-

amplitude coupling to sung speech by infants. *BioRxiv*, 2020.10.12.329326. https://doi.org/10.1101/2020.10.12.329326

- Baayen, R. H., Davidson, D. J., & Bates, D. M. (2008). Mixed-effects modeling with crossed random effects for subjects and items. *Journal of Memory and Language*, 59(4), 390–412. https://doi.org/10.1016/j.jml.2007.12.005
- Bak, T. H., & Robertson, I. (2017). Biology enters the scene—A new perspective on bilingualism, cognition, and dementia. *Neurobiology of Aging*, *50*, iii–iv. https://doi.org/10.1016/j.neurobiolaging.2016.10.020
- Barac, R., Bialystok, E., Castro, D. C., & Sanchez, M. (2014). The Cognitive
 Development of Young Dual Language Learners: A Critical Review. *Early Childhood Research Quarterly*, 29(4), 699–714.
 https://doi.org/10.1016/j.ecresq.2014.02.003
- Barac, R., Moreno, S., & Bialystok, E. (2016). Behavioral and Electrophysiological
 Differences in Executive Control Between Monolingual and Bilingual Children.
 Child Development, 87(4), 1277–1290. https://doi.org/10.1111/cdev.12538
- Barry, R. J., Clarke, A. R., McCarthy, R., Selikowitz, M., Johnstone, S. J., & Rushby, J. A. (2004). Age and gender effects in EEG coherence: I. Developmental trends in normal children. *Clinical Neurophysiology*, *115*(10), 2252–2258. https://doi.org/10.1016/j.clinph.2004.05.004
- Bartgis, J., Lilly, A. R., & Thomas, D. G. (2003). Event-Related Potential and Behavioral Measures of Attention in 5-, 7-, and 9-Year-Olds. *The Journal of General Psychology*, *130*(3), 311–335. https://doi.org/10.1080/00221300309601162
- Bates, D., Mächler, M., Bolker, B., & Walker, S. (2014). Fitting Linear Mixed-Effects Models using Ime4. *ArXiv:1406.5823 [Stat]*. http://arxiv.org/abs/1406.5823

221

- Baum, S., & Titone, D. (2014). Moving toward a neuroplasticity view of bilingualism, executive control, and aging. *Applied Psycholinguistics*, 35(5), 857–894. https://doi.org/10.1017/S0142716414000174
- Bell, M. A., & Cuevas, K. (2012). Using EEG to Study Cognitive Development: Issues and Practices. Journal of Cognition and Development : Official Journal of the Cognitive Development Society, 13(3), 281–294. https://doi.org/10.1080/15248372.2012.691143
- Benz, S., Sellaro, R., Hommel, B., & Colzato, L. S. (2016). Music Makes the World Go
 Round: The Impact of Musical Training on Non-musical Cognitive Functions—
 A Review. *Frontiers in Psychology*, 6.

https://www.frontiersin.org/articles/10.3389/fpsyg.2015.02023

Berlin, C. I., Hughes, L. F., Lowe-Bell, S. S., & Berlin, H. L. (1973). Dichotic Right Ear Advantage in Children 5 to 131. *Cortex*, *9*(4), 394–402.

https://doi.org/10.1016/S0010-9452(73)80038-3

- Bialystok, E. (1992). Selective Attention in Cognitive Processing: The Bilingual Edge.
 In R. J. Harris (Ed.), Advances in Psychology (Vol. 83, pp. 501–513). NorthHolland. https://doi.org/10.1016/S0166-4115(08)61513-7
- Bialystok, E. (2011). Coordination of executive functions in monolingual and bilingual children. *Journal of Experimental Child Psychology*, *110*(3), 461–468. https://doi.org/10.1016/j.jecp.2011.05.005
- Bialystok, E. (2015). Bilingualism and the Development of Executive Function: The Role of Attention. *Child Development Perspectives*, *9*(2), 117–121. https://doi.org/10.1111/cdep.12116

Bialystok, E. (2017). The Bilingual Adaptation: How Minds Accommodate Experience. *Psychological Bulletin*, *143*(3), 233–262. https://doi.org/10.1037/bul0000099

Bialystok, E., Craik, F. I. M., Grady, C., Chau, W., Ishii, R., Gunji, A., & Pantev, C.
(2005). Effect of bilingualism on cognitive control in the Simon task: Evidence from MEG. *NeuroImage*, *24*(1), 40–49.

https://doi.org/10.1016/j.neuroimage.2004.09.044

- Bialystok, E., Craik, F. I. M., Klein, R., & Viswanathan, M. (2004). Bilingualism, Aging, and Cognitive Control: Evidence From the Simon Task. *Psychology and Aging*, *19*(2), 290–303. https://doi.org/10.1037/0882-7974.19.2.290
- Bialystok, E., Craik, F. I. M., & Luk, G. (2012). Bilingualism: Consequences for mind and brain. *Trends in Cognitive Sciences*, 16(4), 240–250. https://doi.org/10.1016/j.tics.2012.03.001
- Bialystok, E., Craik, F. I. M., & Ruocco, A. C. (2006). Dual-Modality Monitoring in a Classification Task: The Effects of Bilingualism and Ageing. *Quarterly Journal* of Experimental Psychology, 59(11), 1968–1983.

https://doi.org/10.1080/17470210500482955

Bialystok, E., & Martin, M. M. (2004). Attention and inhibition in bilingual children:
Evidence from the dimensional change card sort task. *Developmental Science*, 7(3), 325–339. https://doi.org/10.1111/j.1467-7687.2004.00351.x

Bialystok, E., Martin, M. M., & Viswanathan, M. (2005). Bilingualism across the lifespan: The rise and fall of inhibitory control. *International Journal of Bilingualism*, *9*(1), 103–119.

https://doi.org/10.1177/13670069050090010701

- Bialystok, E., & Shapero, D. (2005). Ambiguous benefits: The effect of bilingualism on reversing ambiguous figures. *Developmental Science*, 8(6), 595–604. https://doi.org/10.1111/j.1467-7687.2005.00451.x
- Blom, E., Küntay, A. C., Messer, M., Verhagen, J., & Leseman, P. (2014). The benefits of being bilingual: Working memory in bilingual Turkish–Dutch children.
 Journal of Experimental Child Psychology, 128, 105–119.
 https://doi.org/10.1016/j.jecp.2014.06.007
- Blumenfeld, H. K., & Marian, V. (2011). Bilingualism influences inhibitory control in auditory comprehension. *Cognition*, *118*(2), 245.
- Blumenfeld, H. K., & Marian, V. (2013). Parallel language activation and cognitive control during spoken word recognition in bilinguals. *Journal of Cognitive Psychology*, 25(5), 547–567. https://doi.org/10.1080/20445911.2013.812093
- Blumenfeld, H. K., Schroeder, S. R., Bobb, S. C., Freeman, M. R., & Marian, V. (2016). Auditory word recognition across the lifespan. *Linguistic Approaches to Bilingualism*, 6(1–2), 119.
- Boeck, P. D., Bakker, M., Zwitser, R., Nivard, M., Hofman, A., Tuerlinckx, F., &
 Partchev, I. (2011). The Estimation of Item Response Models with the Imer
 Function from the Ime4 Package in R. *Journal of Statistical Software*, *39*, 1–
 28. https://doi.org/10.18637/jss.v039.i12

Bonfieni, M., Branigan, H. P., Pickering, M. J., & Sorace, A. (2019). Language
experience modulates bilingual language control: The effect of proficiency,
age of acquisition, and exposure on language switching. *Acta Psychologica*,
193, 160–170. https://doi.org/10.1016/j.actpsy.2018.11.004

- Botvinick, M. M., Cohen, J. D., & Carter, C. S. (2004). Conflict monitoring and anterior cingulate cortex: An update. *Trends in Cognitive Sciences*, 8(12), 539–546. https://doi.org/10.1016/j.tics.2004.10.003
- Bradlow, A. R., & Alexander, J. A. (2007). Semantic and phonetic enhancements for speech-in-noise recognition by native and non-native listeners. *The Journal of the Acoustical Society of America*, *121*(4), 2339–2349.

https://doi.org/10.1121/1.2642103

- Bressler, S. L., Tang, W., Sylvester, C. M., Shulman, G. L., & Corbetta, M. (2008). TopDown Control of Human Visual Cortex by Frontal and Parietal Cortex in
 Anticipatory Visual Spatial Attention. *Journal of Neuroscience*, *28*(40), 10056–
 10061. https://doi.org/10.1523/JNEUROSCI.1776-08.2008
- Briellmann, R. S., Saling, M. M., Connell, A. B., Waites, A. B., Abbott, D. F., & Jackson,
 G. D. (2004). A high-field functional MRI study of quadri-lingual subjects.
 Brain and Language, 89(3), 531–542.

https://doi.org/10.1016/j.bandl.2004.01.008

- Broadbent, D. E. (1958a). CHAPTER 1 INTRODUCTION: HEARING AND BEHAVIOUR. In D. E. Broadbent (Ed.), *Perception and Communication* (pp. 1–10). Pergamon. https://doi.org/10.1016/B978-1-4832-0079-8.50003-7
- Broadbent, D. E. (1958b). CHAPTER 2—SELECTIVE LISTENING TO SPEECH. In D. E. Broadbent (Ed.), *Perception and Communication* (pp. 11–35). Pergamon. https://doi.org/10.1016/B978-1-4832-0079-8.50004-9
- Broadbent, D. E. (1965). Information Processing in the Nervous System. *Science*, *150*(3695), 457–462.

Broderick, M. P., Anderson, A. J., Di Liberto, G. M., Crosse, M. J., & Lalor, E. C. (2018).
Electrophysiological Correlates of Semantic Dissimilarity Reflect the
Comprehension of Natural, Narrative Speech. *Current Biology*, *28*(5), 803-809.e3. https://doi.org/10.1016/j.cub.2018.01.080

Bronkhorst, A. W. (2015). The cocktail-party problem revisited: Early processing and selection of multi-talker speech. *Attention, Perception, & Psychophysics,* 77(5), 1465–1487. https://doi.org/10.3758/s13414-015-0882-9

- Brungart, D. S., & Simpson, B. D. (2007). Effect of target-masker similarity on acrossear interference in a dichotic cocktail-party listening task. *The Journal of the Acoustical Society of America*, *122*(3), 1724–1734. https://doi.org/10.1121/1.2756797
- Bunge, S. A., Dudukovic, N. M., Thomason, M. E., Vaidya, C. J., & Gabrieli, J. D. E.
 (2002). Immature Frontal Lobe Contributions to Cognitive Control in Children:
 Evidence from fMRI. *Neuron*, *33*(2), 301–311. https://doi.org/10.1016/S0896-6273(01)00583-9

Burgaleta, M., Sanjuán, A., Ventura-Campos, N., Sebastian-Galles, N., & Ávila, C. (2016). Bilingualism at the core of the brain. Structural differences between bilinguals and monolinguals revealed by subcortical shape analysis. *NeuroImage*, *125*, 437–445.

https://doi.org/10.1016/j.neuroimage.2015.09.073

Bush, G., Luu, P., & Posner, M. I. (2000). Cognitive and emotional influences in anterior cingulate cortex. *Trends in Cognitive Sciences*, 4(6), 215–222.
https://doi.org/10.1016/S1364-6613(00)01483-2

Byers-Heinlein, K., Burns, T. C., & Werker, J. F. (2010). The Roots of Bilingualism in Newborns. *Psychological Science*, *21*(3), 343–348. https://doi.org/10.1177/0956797609360758

Calvo, A., & Bialystok, E. (2014). Independent Effects of Bilingualism and Socioeconomic Status on Language Ability and Executive Functioning. *Cognition*, *130*(3), 278–288. https://doi.org/10.1016/j.cognition.2013.11.015

- Canbay, O. (2011). Comparing the phonological awareness of bilingual and monolingual pre-school children. *Procedia - Social and Behavioral Sciences*, 15, 976–980. https://doi.org/10.1016/j.sbspro.2011.03.224
- Carlson, S. M., & Meltzoff, A. N. (2008). Bilingual experience and executive functioning in young children. *Developmental Science*, *11*(2), 282–298. https://doi.org/10.1111/j.1467-7687.2008.00675.x
- Casey, B. J., Tottenham, N., Liston, C., & Durston, S. (2005). Imaging the developing brain: What have we learned about cognitive development? *Trends in Cognitive Sciences*, *9*(3), 104–110. https://doi.org/10.1016/j.tics.2005.01.011
- Casey, B. J., Trainor, R. J., Orendi, J. L., Schubert, A. B., Nystrom, L. E., Giedd, J. N.,
 Castellanos, F. X., Haxby, J. V., Noll, D. C., Cohen, J. D., Forman, S. D., Dahl, R.
 E., & Rapoport, J. L. (1997). A Developmental Functional MRI Study of
 Prefrontal Activation during Performance of a Go-No-Go Task. *Journal of Cognitive Neuroscience*, 9(6), 835–847.

https://doi.org/10.1162/jocn.1997.9.6.835

Chandler, J. J., & Paolacci, G. (2017). Lie for a Dime: When Most Prescreening Responses Are Honest but Most Study Participants Are Impostors. *Social* *Psychological and Personality Science*, *8*(5), 500–508.

https://doi.org/10.1177/1948550617698203

Chee, M. W. L., Soon, C. S., Lee, H. L., & Pallier, C. (2004). Left insula activation: A marker for language attainment in bilinguals. *Proceedings of the National Academy of Sciences*, *101*(42), 15265–15270.

https://doi.org/10.1073/pnas.0403703101

- Chee, M. W. L., Tan, E. W. L., & Thiel, T. (1999). Mandarin and English Single Word Processing Studied with Functional Magnetic Resonance Imaging. *Journal of Neuroscience*, *19*(8), 3050–3056. https://doi.org/10.1523/JNEUROSCI.19-08-03050.1999
- Cherry, E. C. (1953). Some Experiments on the Recognition of Speech, with One and with Two Ears. *The Journal of the Acoustical Society of America*, *25*(5), 975– 979. https://doi.org/10.1121/1.1907229
- Clark, C. W., & Dukas, R. (2003). The behavioral ecology of a cognitive constraint: Limited attention. *Behavioral Ecology*, *14*(2), 151–156. https://doi.org/10.1093/beheco/14.2.151
- Coch, D., Sanders, L. D., & Neville, H. J. (2005). An Event-related Potential Study of Selective Auditory Attention in Children and Adults. *Journal of Cognitive Neuroscience*, *17*(4), 605–622. https://doi.org/10.1162/0898929053467631
- Cohen, M. X. (2014). *Analyzing Neural Time Series Data: Theory and Practice*. MIT Press.
- Comalli, P. E., Wapner, S., & Werner, H. (1962). Interference effects of Stroop Color-Word Test in childhood, adulthood, and aging. *The Journal of Genetic Psychology; Provincetown, Mass., Etc., 100*(1), 47–53.

Comishen, K. J., Bialystok, E., & Adler, S. A. (2019). The impact of bilingual environments on selective attention in infancy. *Developmental Science*, *22*(4), e12797. https://doi.org/10.1111/desc.12797

Corbetta, M., Miezin, F. M., Shulman, G. L., & Petersen, S. E. (2007). Selective Attention Modulates Extrastriate Visual Regions in Humans During Visual Feature Discrimination and Recognition. In *Ciba Foundation Symposium 163—Exploring Brain Functional Anatomy with Positron Tomography* (pp. 165–180). John Wiley & Sons, Ltd.

https://doi.org/10.1002/9780470514184.ch10

- Corbetta, M., & Shulman, G. L. (2002). Control of goal-directed and stimulus-driven attention in the brain. *Nature Reviews Neuroscience*, *3*(3), Article 3. https://doi.org/10.1038/nrn755
- Corteen, R. S., & Dunn, D. (1974). Shock-associated words in a nonattended message: A test for momentary awareness. *Journal of Experimental Psychology*, *102*(6), 1143–1144. https://doi.org/10.1037/h0036362
- Costa, A., Hernández, M., & Sebastián-Gallés, N. (2008). Bilingualism aids conflict resolution: Evidence from the ANT task. *Cognition*, *106*(1), 59–86. https://doi.org/10.1016/j.cognition.2006.12.013
- Costa, A., & Sebastián-Gallés, N. (2014). How does the bilingual experience sculpt the brain? *Nature Reviews Neuroscience*, *15*(5), 336–345. https://doi.org/10.1038/nrn3709
- Costa-Faidella, J., Sussman, E. S., & Escera, C. (2017). Selective entrainment of brain oscillations drives auditory perceptual organization. *NeuroImage*, *159*, 195– 206. https://doi.org/10.1016/j.neuroimage.2017.07.056

- Crone, E. A., & Richard Ridderinkhof, K. (2011). The developing brain: From theory to neuroimaging and back. *Developmental Cognitive Neuroscience*, 1(2), 101–109. https://doi.org/10.1016/j.dcn.2010.12.001
- Crone, E. A., & Steinbeis, N. (2017). Neural Perspectives on Cognitive Control Development during Childhood and Adolescence. *Trends in Cognitive Sciences*, *21*(3), 205–215. https://doi.org/10.1016/j.tics.2017.01.003
- Crosse, M. J., Di Liberto, G. M., Bednar, A., & Lalor, E. C. (2016). The Multivariate Temporal Response Function (mTRF) Toolbox: A MATLAB Toolbox for Relating Neural Signals to Continuous Stimuli. *Frontiers in Human Neuroscience*, *10*. https://doi.org/10.3389/fnhum.2016.00604
- Crosse, M. J., Zuk, N. J., Di Liberto, G. M., Nidiffer, A. R., Molholm, S., & Lalor, E. C.
 (2021). Linear Modeling of Neurophysiological Responses to Speech and
 Other Continuous Stimuli: Methodological Considerations for Applied
 Research. *Frontiers in Neuroscience*, *15*, 705621.

https://doi.org/10.3389/fnins.2021.705621

D'Angiulli, A., Herdman, A., Stapells, D., & Hertzman, C. (2008). Children's Event-Related Potentials of Auditory Selective Attention Vary With Their Socioeconomic Status. *Neuropsychology*, *22*(3), 293–300. Scopus. https://doi.org/10.1037/0894-4105.22.3.293

 Darcy, N. T. (1953). A Review of the Literature on the Effects of Bilingualism upon the Measurement of Intelligence. *The Pedagogical Seminary and Journal of Genetic Psychology*, *82*(1), 21–57. https://doi.org/10.1080/08856559.1953.10533654 Davidson, M. C., Amso, D., Anderson, L. C., & Diamond, A. (2006). Development of cognitive control and executive functions from 4 to 13 years: Evidence from manipulations of memory, inhibition, and task switching. *Neuropsychologia*, 44(11), 2037–2078. https://doi.org/10.1016/j.neuropsychologia.2006.02.006

Delorme, A., & Makeig, S. (2004). EEGLAB: An open source toolbox for analysis of single-trial EEG dynamics including independent component analysis. *Journal of Neuroscience Methods*, 134(1), 9–21.

https://doi.org/10.1016/j.jneumeth.2003.10.009

- Dempster, F. N. (1992). The rise and fall of the inhibitory mechanism: Toward a unified theory of cognitive development and aging. *Developmental Review*, *12*(1), 45–75. https://doi.org/10.1016/0273-2297(92)90003-K
- Desjardins, J. L., Bangert, A., & Gomez, N. (2020). What Does Language Have to Do With It? The Impact of Age and Bilingual Experience on Inhibitory Control in an Auditory Dichotic Listening Task. *Journal of Speech, Language, and Hearing Research, 63*(5), 1581–1594. https://doi.org/10.1044/2020_JSLHR-19-00238
- Deutsch, J. A., & Deutsch, D. (1963). Attention: Some theoretical considerations. *Psychological Review*, *70*, 80–90. https://doi.org/10.1037/h0039515
- Dewaele, J.-M. (2018). Online Questionnaires. In A. Phakiti, P. De Costa, L. Plonsky, &
 S. Starfield (Eds.), *The Palgrave Handbook of Applied Linguistics Research Methodology* (pp. 269–286). Palgrave Macmillan UK. https://doi.org/10.1057/978-1-137-59900-1_13

- Di Liberto, G. M., Lalor, E. C., & Millman, R. E. (2018). Causal cortical dynamics of a predictive enhancement of speech intelligibility. *NeuroImage*, *166*, 247–258. https://doi.org/10.1016/j.neuroimage.2017.10.066
- Di Liberto, G. M., O'Sullivan, J. A., & Lalor, E. C. (2015). Low-Frequency Cortical Entrainment to Speech Reflects Phoneme-Level Processing. *Current Biology: CB*, *25*(19), 2457–2465. https://doi.org/10.1016/j.cub.2015.08.030
- Di Liberto, G. M., Pelofi, C., Bianco, R., Patel, P., Mehta, A. D., Herrero, J. L., de Cheveigné, A., Shamma, S., & Mesgarani, N. (2020). Cortical encoding of melodic expectations in human temporal cortex. *ELife*, *9*, e51784. https://doi.org/10.7554/eLife.51784
- Dimitropoulou, M., Duñabeitia, J. A., & Carreiras, M. (2011). Two Words, One Meaning: Evidence of Automatic Co-Activation of Translation Equivalents. *Frontiers in Psychology*, 2.

https://www.frontiersin.org/articles/10.3389/fpsyg.2011.00188

Ding, N., & Simon, J. Z. (2012). Emergence of neural encoding of auditory objects while listening to competing speakers. *Proceedings of the National Academy* of Sciences, 109(29), 11854–11859.

https://doi.org/10.1073/pnas.1205381109

- Ding, N., & Simon, J. Z. (2014). Cortical entrainment to continuous speech: Functional roles and interpretations. *Frontiers in Human Neuroscience*, *8*, 311. https://doi.org/10.3389/fnhum.2014.00311
- Doelling, K. B., Arnal, L. H., Ghitza, O., & Poeppel, D. (2014). Acoustic landmarks drive delta–theta oscillations to enable speech comprehension by facilitating

perceptual parsing. NeuroImage, 85, 761-768.

https://doi.org/10.1016/j.neuroimage.2013.06.035

Donders, F. C. (1868). On the speed of mental processes. *Acta Psychologica*, *30*, 412–431.

Dornic, S. (1980). Language dominance, spare capacity and perceived effort in bilinguals. *Ergonomics*, *23*(4), 369–377. https://doi.org/10.1080/00140138008924750

- Dosenbach, N. U. F., Nardos, B., Cohen, A. L., Fair, D. A., Power, J. D., Church, J. A., Nelson, S. M., Wig, G. S., Vogel, A. C., Lessov-Schlaggar, C. N., Barnes, K. A., Dubis, J. W., Feczko, E., Coalson, R. S., Pruett, J. R., Barch, D. M., Petersen, S. E., & Schlaggar, B. L. (2010). Prediction of Individual Brain Maturity Using fMRI. *Science (New York, N.Y.)*, *329*(5997), 1358–1361. https://doi.org/10.1126/science.1194144
- Downs, J. S., Holbrook, M. B., Sheng, S., & Cranor, L. F. (2010). Are your participants gaming the system? Screening mechanical turk workers. *Proceedings of the SIGCHI Conference on Human Factors in Computing Systems*, 2399–2402. https://doi.org/10.1145/1753326.1753688
- Doyle, A.-B. (1973). Listening to distraction: A developmental study of selective attention. *Journal of Experimental Child Psychology*, *15*(1), 100–115. https://doi.org/10.1016/0022-0965(73)90134-3
- Driver, J., & Baylis, G. C. (1998). Attention and visual object segmentation. In *The attentive brain* (pp. 299–325). The MIT Press.

- Driver, J., & Frackowiak, R. S. J. (2001). Neurobiological measures of human selective attention. *Neuropsychologia*, *39*(12), 1257–1262. https://doi.org/10.1016/S0028-3932(01)00115-4
- Duñabeitia, J. A., Hernández, J. A., Antón, E., Macizo, P., Estévez, A., Fuentes, L. J., & Carreiras, M. (2014). The inhibitory advantage in bilingual children revisited:
 Myth or reality? *Experimental Psychology*, *61*(3), 234–251.
 https://doi.org/10.1027/1618-3169/a000243
- Durand López, E. M. (2021). A bilingual advantage in memory capacity: Assessing the roles of proficiency, number of languages acquired and age of acquisition. *International Journal of Bilingualism*, *25*(3), 606–621. https://doi.org/10.1177/1367006920965714
- Durston, S., Hulshoff pol, H. E., Casey, B. J., Giedd, J. N., Buitelaar, J. K., & Van engeland, H. (2001). Anatomical MRI of the Developing Human Brain: What Have We Learned? *Journal of the American Academy of Child & Adolescent Psychiatry*, *40*(9), 1012–1020. https://doi.org/10.1097/00004583-200109000-00009
- Edelman, G. M., & Gally, J. A. (2001). Degeneracy and complexity in biological systems. *Proceedings of the National Academy of Sciences*, *98*(24), 13763–13768. https://doi.org/10.1073/pnas.231499798
- Edgell, B. (2001). The British Psychological Society. *British Journal of Psychology*, *92*, 3–22. https://doi.org/10.1348/000712601162077
- Engel de Abreu, P. M. J., Cruz-Santos, A., Tourinho, C. J., Martin, R., & Bialystok, E. (2012). Bilingualism Enriches the Poor: Enhanced Cognitive Control in Low-

Income Minority Children. *Psychological Science*, *23*(11), 1364–1371. https://doi.org/10.1177/0956797612443836

- Estanga, A., Ecay-Torres, M., Ibañez, A., Izagirre, A., Villanua, J., Garcia-Sebastian, M., Iglesias Gaspar, M. T., Otaegui-Arrazola, A., Iriondo, A., Clerigue, M., & Martinez-Lage, P. (2017). Beneficial effect of bilingualism on Alzheimer's disease CSF biomarkers and cognition. *Neurobiology of Aging*, *50*, 144–151. https://doi.org/10.1016/j.neurobiolaging.2016.10.013
- Eviatar, Z., & Ibrahim, R. (2000). Bilingual is as bilingual does: Metalinguistic abilities of Arabic-speaking children. *Applied Psycholinguistics*, *21*(4), 451–471.

Fernandes, M. A., Craik, F., Bialystok, E., & Kreuger, S. (2007). Effects of bilingualism, aging, and semantic relatedness on memory under divided attention. *Canadian Journal of Experimental Psychology/Revue Canadienne de Psychologie Expérimentale*, 61(2), 128–141. https://doi.org/10.1037/cjep2007014

Fernandez, N. B., Trost, W. J., & Vuilleumier, P. (2019). Brain networks mediating the influence of background music on selective attention. *Social Cognitive and Affective Neuroscience*, *14*(12), 1441–1452.

https://doi.org/10.1093/scan/nsaa004

- Fiedler, L., Wöstmann, M., Herbst, S. K., & Obleser, J. (2018). Late cortical tracking of ignored speech facilitates neural selectivity in acoustically challenging conditions. *BioRxiv*, 238642. https://doi.org/10.1101/238642
- Filippi, R., Ceccolini, A., Periche-Tomas, E., & Bright, P. (2020). Developmental trajectories of metacognitive processing and executive function from

childhood to older age. *Quarterly Journal of Experimental Psychology*, 73(11), 1757–1773. https://doi.org/10.1177/1747021820931096

- Filippi, R., Leech, R., Thomas, M. S. C., Green, D. W., & Dick, F. (2012). A bilingual advantage in controlling language interference during sentence comprehension. *Bilingualism: Language and Cognition*, 15(4), 858–872. https://doi.org/10.1017/S1366728911000708
- Filippi, R., Morris, J., Richardson, F. M., Bright, P., Thomas, M. S. C., Karmiloff-Smith,
 A., & Marian, V. (2015). Bilingual children show an advantage in controlling
 verbal interference during spoken language comprehension. *Bilingualism: Language and Cognition*, 18(3), 490–501.

https://doi.org/10.1017/S1366728914000686

- Filippi, R., Richardson, F. M., Dick, F., Leech, R., Green, D. W., Thomas, M. S. C., & Price, C. J. (2011). The Right Posterior Paravermis and the Control of Language Interference. *Journal of Neuroscience*, *31*(29), 10732–10740. https://doi.org/10.1523/JNEUROSCI.1783-11.2011
- Franzen, L. (2023). How to Run Behavioural Experiments Online: Best Practice Suggestions for Cognitive Psychology and Neuroscience (No. 1). 3(1), Article 1. https://doi.org/10.5334/spo.34

Friederici, A. D., Steinhauer, K., & Pfeifer, E. (2002). Brain signatures of artificial language processing: Evidence challenging the critical period hypothesis. *Proceedings of the National Academy of Sciences*, *99*(1), 529–534.
https://doi.org/10.1073/pnas.012611199

Frutos-Lucas, J. de, López-Sanz, D., Cuesta, P., Bruña, R., Fuente, S. de la, Serrano, N., López, M. E., Delgado-Losada, M. L., López-Higes, R., Marcos, A., & Maestú, F. (2020). Enhancement of posterior brain functional networks in bilingual older adults. *Bilingualism: Language and Cognition*, *23*(2), 387–400. https://doi.org/10.1017/S1366728919000178

- Garbin, G., Sanjuan, A., Forn, C., Bustamante, J. C., Rodriguez-Pujadas, A., Belloch, V.,
 Hernandez, M., Costa, A., & Ávila, C. (2010). Bridging language and attention:
 Brain basis of the impact of bilingualism on cognitive control. *NeuroImage*,
 53(4), 1272–1278. https://doi.org/10.1016/j.neuroimage.2010.05.078
- Gathercole, S. E., Pickering, S. J., Ambridge, B., & Wearing, H. (2004). The Structure of Working Memory From 4 to 15 Years of Age. *Developmental Psychology*, 40, 177–190. https://doi.org/10.1037/0012-1649.40.2.177
- Gathercole, V. C. M., Thomas, E. M., Kennedy, I., Prys, C., Young, N., Viñas Guasch, N., Roberts, E. J., Hughes, E. K., & Jones, L. (2014). Does language dominance affect cognitive performance in bilinguals? Lifespan evidence from preschoolers through older adults on card sorting, Simon, and metalinguistic tasks. *Frontiers in Psychology*, *5*. https://doi.org/10.3389/fpsyg.2014.00011
- Geffen, G., & Sexton, M. A. (1978). The development of auditory strategies of attention. *Developmental Psychology*, *14*(1), 11–17.

https://doi.org/10.1037/0012-1649.14.1.11

- Geffen, G., & Wale, J. (1979). Development of selective listening and hemispheric asymmetry. *Developmental Psychology*, 15(2), 138–146. https://doi.org/10.1037/0012-1649.15.2.138
- Giedd, J. N., Blumenthal, J., Jeffries, N. O., Castellanos, F. X., Liu, H., Zijdenbos, A., Paus, T., Evans, A. C., & Rapoport, J. L. (1999). Brain development during

childhood and adolescence: A longitudinal MRI study. *Nature Neuroscience*, 2(10), 861. https://doi.org/10.1038/13158

Giraud, A.-L., & Poeppel, D. (2012a). Speech Perception from a Neurophysiological Perspective. In D. Poeppel, T. Overath, A. N. Popper, & R. R. Fay (Eds.), *The Human Auditory Cortex* (pp. 225–260). Springer.

https://doi.org/10.1007/978-1-4614-2314-0_9

- Giraud, A.-L., & Poeppel, D. (2012b). Cortical oscillations and speech processing: Emerging computational principles and operations. *Nature Neuroscience*, *15*(4), 511–517. https://doi.org/10.1038/nn.3063
- Gogtay, N., Giedd, J. N., Lusk, L., Hayashi, K. M., Greenstein, D., Vaituzis, A. C., Nugent, T. F., Herman, D. H., Clasen, L. S., Toga, A. W., Rapoport, J. L., & Thompson, P. M. (2004). Dynamic mapping of human cortical development during childhood through early adulthood. *Proceedings of the National Academy of Sciences*, *101*(21), 8174–8179.

https://doi.org/10.1073/pnas.0402680101

- Gold, B. T. (2015). Lifelong bilingualism and neural reserve against Alzheimer's disease: A review of findings and potential mechanisms. *Behavioural Brain Research*, *281*, 9–15. https://doi.org/10.1016/j.bbr.2014.12.006
- Gold, B. T., Kim, C., Johnson, N. F., Kryscio, R. J., & Smith, C. D. (2013). Lifelong
 Bilingualism Maintains Neural Efficiency for Cognitive Control in Aging.
 Journal of Neuroscience, 33(2), 387–396.
 https://doi.org/10.1523/JNEUROSCI.3837-12.2013

- Goldrick, M., Putnam, M., & Schwarz, L. (2016). Coactivation in bilingual grammars: A computational account of code mixing. *Bilingualism: Language and Cognition*, *19*(5), 857–876. https://doi.org/10.1017/S1366728915000802
- Gomes, H., Duff, M., Barnhardt, J., Barrett, S., & Ritter, W. (2007). Development of auditory selective attention: Event-related potential measures of channel selection and target detection. *Psychophysiology*, *44*(5), 711–727. https://doi.org/10.1111/j.1469-8986.2007.00555.x
- Gopher, D. (1986). In Defence of Resources: On Structures, Energies, Pools and the Allocation of Attention. In G. R. J. Hockey, A. W. K. Gaillard, & M. G. H. Coles (Eds.), *Energetics and Human Information Processing* (pp. 353–371). Springer Netherlands. https://doi.org/10.1007/978-94-009-4448-0_25
- Goswami, U. (2008). Principles of Learning, Implications for Teaching: A Cognitive Neuroscience Perspective. *Journal of Philosophy of Education*, *42*(3–4), 381– 399. https://doi.org/10.1111/j.1467-9752.2008.00639.x

Green, D. W. (1998). Mental control of the bilingual lexico-semantic system. *Bilingualism: Language and Cognition*, 1(2), 67–81. https://doi.org/10.1017/S1366728998000133

- Green, D. W., & Abutalebi, J. (2013). Language control in bilinguals: The adaptive control hypothesis. *Journal of Cognitive Psychology*, 25(5), 515–530. https://doi.org/10.1080/20445911.2013.796377
- Green, D. W., Crinion, J., & Price, C. J. (2006). Convergence, Degeneracy, and Control. *Language Learning*, *56*(s1), 99–125. https://doi.org/10.1111/j.1467-9922.2006.00357.x

Greenberg, L. M., & Waldmant, I. D. (1993). Developmental Normative Data on The Test of Variables of Attention (T.O.V.A.[™]). *Journal of Child Psychology and Psychiatry*, *34*(6), 1019–1030. https://doi.org/10.1111/j.1469-7610.1993.tb01105.x

Haegens, S., & Zion Golumbic, E. (2018). Rhythmic facilitation of sensory processing:
A critical review. *Neuroscience & Biobehavioral Reviews*, *86*, 150–165.
https://doi.org/10.1016/j.neubiorev.2017.12.002

Hakuta, K. (1986). Cognitive Development of Bilingual Children.

https://eric.ed.gov/?id=ED278260

Hämäläinen, S., Sairanen, V., Leminen, A., & Lehtonen, M. (2017). Bilingualism modulates the white matter structure of language-related pathways. *NeuroImage*, *152*, 249–257.

https://doi.org/10.1016/j.neuroimage.2017.02.081

- Hamilton, L. S., & Huth, A. G. (2020). The revolution will not be controlled: Natural stimuli in speech neuroscience. *Language, Cognition and Neuroscience*, 35(5), 573–582. https://doi.org/10.1080/23273798.2018.1499946
- Hampton Wray, A., Stevens, C., Pakulak, E., Isbell, E., Bell, T., & Neville, H. (2017).
 Development of selective attention in preschool-age children from lower socioeconomic status backgrounds. *Developmental Cognitive Neuroscience*, 26, 101–111. https://doi.org/10.1016/j.dcn.2017.06.006
- Har-shai Yahav, P., & Zion Golumbic, E. (2021). Linguistic processing of task-irrelevant speech at a cocktail party. *ELife*, *10*, e65096.

https://doi.org/10.7554/eLife.65096

- Hatzidaki, A., Branigan, H. P., & Pickering, M. J. (2011). Co-activation of syntax in bilingual language production. *Cognitive Psychology*, *62*(2), 123–150.
 https://doi.org/10.1016/j.cogpsych.2010.10.002
- Hawley, M. L., Litovsky, R. Y., & Culling, J. F. (2004). The benefit of binaural hearing in a cocktail party: Effect of location and type of interferer. *The Journal of the Acoustical Society of America*, *115*(2), 833–843.

https://doi.org/10.1121/1.1639908

- Hayakawa, S., & Marian, V. (2019). Consequences of multilingualism for neural architecture. *Behavioral and Brain Functions*, *15*(1).
 https://www.academia.edu/38789240/Consequences_of_multilingualism_for_neural_architecture
- Hilchey, M. D., & Klein, R. M. (2011). Are there bilingual advantages on nonlinguistic interference tasks? Implications for the plasticity of executive control processes. *Psychonomic Bulletin & Review*, *18*(4), 625–658.
 https://doi.org/10.3758/s13423-011-0116-7
- Hiscock, M., & Kinsbourne, M. (1977). Selective listening asymmetry in preschool children. *Developmental Psychology*, *13*(3), 217–224.

https://doi.org/10.1037/0012-1649.13.3.217

- Hiscock, M. & Kinsbourne, M. (1980). Asymmetries of Selective Listening and Attention Switching in Children. *Developmental Psychology*, *16*(1), 70–82.
- Hoff, E. (2003). Causes and consequences of SES-related differences in parent-tochild speech. In *Socioeconomic status, parenting, and child development* (pp. 147–160). Lawrence Erlbaum Associates Publishers.

- Hoff, E., Core, C., Place, S., Rumiche, R., Señor, M., & Parra, M. (2012). Dual language exposure and early bilingual development. *Journal of Child Language*, *39*(1), 1–27. https://doi.org/10.1017/S0305000910000759
- Horton, C., D'Zmura, M., & Srinivasan, R. (2013). Suppression of competing speech through entrainment of cortical oscillations. *Journal of Neurophysiology*, *109*(12), 3082–3093. https://doi.org/10.1152/jn.01026.2012
- Horton, C., Srinivasan, R., & D'Zmura, M. (2014). Envelope responses in single-trial EEG indicate attended speaker in a `cocktail party'. *Journal of Neural Engineering*, *11*(4), 046015. https://doi.org/10.1088/1741-2560/11/4/046015
- Huk, A., Bonnen, K., & He, B. J. (2018). Beyond Trial-Based Paradigms: Continuous
 Behavior, Ongoing Neural Activity, and Natural Stimuli. *Journal of Neuroscience*, *38*(35), 7551–7558. https://doi.org/10.1523/JNEUROSCI.192017.2018
- Hunt, E., Pellegrino, J. W., & Yee, P. L. (1989). Individual Differences in Attention. In
 G. H. Bower (Ed.), *Psychology of Learning and Motivation* (Vol. 24, pp. 285– 310). Academic Press. https://doi.org/10.1016/S0079-7421(08)60540-X
- Illes, J., Francis, W. S., Desmond, J. E., Gabrieli, J. D. E., Glover, G. H., Poldrack, R., Lee, C. J., & Wagner, A. D. (1999). Convergent Cortical Representation of Semantic Processing in Bilinguals. *Brain and Language*, *70*(3), 347–363. https://doi.org/10.1006/brln.1999.2186
- Iniesta, A., Paolieri, D., Serrano, F., & Bajo, M. T. (2021). Bilingual writing coactivation: Lexical and sublexical processing in a word dictation task. *Bilingualism: Language and Cognition*, 24(5), 902–917.
 https://doi.org/10.1017/S1366728921000274

- Isbell, E., Wray, A. H., & Neville, H. J. (2016a). Individual differences in neural mechanisms of selective auditory attention in preschoolers from lower socioeconomic status backgrounds: An event-related potentials study. *Developmental Science*, 19(6), 865–880. https://doi.org/10.1111/desc.12334
- Isbell, E., Wray, A. H., & Neville, H. J. (2016b). Individual differences in neural mechanisms of selective auditory attention in preschoolers from lower socioeconomic status backgrounds: An event-related potentials study. *Developmental Science*, 19(6), 865–880. https://doi.org/10.1111/desc.12334
- Itti, L., Rees, G., & Tsotsos, J. K. (2005). Neurobiology of Attention. Elsevier.
- James, W. (1890). The principles of psychology. New York : Henry Holt, 1890.
- Janic, A., Cavanagh, P., & Rivest, J. (2020). Effect of bilingualism on visual tracking attention and resistance to distraction. *Scientific Reports*, *10*(1), 14263. https://doi.org/10.1038/s41598-020-71185-6
- Janssen, C. P., & Brumby, D. P. (2010). Strategic Adaptation to Performance Objectives in a Dual-Task Setting. *Cognitive Science*, *34*(8), 1548–1560. https://doi.org/10.1111/j.1551-6709.2010.01124.x
- Jasińska, K. K., & Petitto, L. A. (2013). How age of bilingual exposure can change the neural systems for language in the developing brain: A functional near infrared spectroscopy investigation of syntactic processing in monolingual and bilingual children. *Developmental Cognitive Neuroscience*, *6*, 87–101. https://doi.org/10.1016/j.dcn.2013.06.005
- Jasińska, K. K., & Petitto, L. A. (2014). Development of Neural Systems for Reading in the Monolingual and Bilingual Brain: New Insights From Functional Near

Infrared Spectroscopy Neuroimaging. *Developmental Neuropsychology*, *39*(6), 421–439. https://doi.org/10.1080/87565641.2014.939180

Jessen, S., Fiedler, L., Münte, T. F., & Obleser, J. (2019). *Quantifying the individual auditory and visual brain response in 7- month-old infants watching a brief cartoon movie*. Neuroscience. https://doi.org/10.1101/610709

Jessen, S., Obleser, J., & Tune, S. (2021). Neural tracking in infants – An analytical tool for multisensory social processing in development. *Developmental Cognitive Neuroscience*, *52*, 101034.

https://doi.org/10.1016/j.dcn.2021.101034

- Johnson, J. A., & Zatorre, R. J. (2006). Neural substrates for dividing and focusing attention between simultaneous auditory and visual events. *NeuroImage*, *31*(4), 1673–1681. https://doi.org/10.1016/j.neuroimage.2006.02.026
- Johnson, M. H. (2001). Functional brain development in humans. *Nature Reviews Neuroscience*, *2*(7), 475–483. https://doi.org/10.1038/35081509
- Johnston, W. A., & Heinz, S. P. (1978). Flexibility and capacity demands of attention. Journal of Experimental Psychology: General, 107(4), 420–435. https://doi.org/10.1037/0096-3445.107.4.420
- Jones, P. R., Moore, D. R., & Amitay, S. (2015). Development of auditory selective attention: Why children struggle to hear in noisy environments.
 Developmental Psychology, *51*(3), 353–369. Scopus.
 https://doi.org/10.1037/a0038570
- Jones, S. K., Davies-Thompson, J., & Tree, J. (2021). Can Machines Find the Bilingual Advantage? Machine Learning Algorithms Find No Evidence to Differentiate Between Lifelong Bilingual and Monolingual Cognitive Profiles. *Frontiers in*

Human Neuroscience, 15, 621772.

https://doi.org/10.3389/fnhum.2021.621772

- Jones, W. R., & Stewart, W. a. C. (1951). Bilingualism and Verbal Intelligence. *British Journal of Statistical Psychology*, *4*(1), 3–8. https://doi.org/10.1111/j.2044-8317.1951.tb00300.x
- Joshi, N. J., Tononi, G., & Koch, C. (2013). The Minimal Complexity of Adapting Agents Increases with Fitness. *PLOS Computational Biology*, *9*(7), e1003111. https://doi.org/10.1371/journal.pcbi.1003111
- Kahneman, D. (1973). Attention and effort. Englewood Cliffs, N.J. : Prentice-Hall, c1973.
- Kalashnikova, M., Pejovic, J., & Carreiras, M. (2021). The effects of bilingualism on attentional processes in the first year of life. *Developmental Science*, *24*(2), e13011. https://doi.org/10.1111/desc.13011
- Kalashnikova, M., Peter, V., Di Liberto, G. M., Lalor, E. C., & Burnham, D. (2018).
 Infant-directed speech facilitates seven-month-old infants' cortical tracking of speech. *Scientific Reports*, 8(1). https://doi.org/10.1038/s41598-018-32150-6
- Kapa, L. L., & Colombo, J. (2013). Attentional control in early and later bilingual children. *Cognitive Development*, 28(3), 233–246. https://doi.org/10.1016/j.cogdev.2013.01.011

Karns, C. M., Isbell, E., Giuliano, R. J., & Neville, H. J. (2015). Auditory attention in childhood and adolescence: An event-related potential study of spatial selective attention to one of two simultaneous stories. *Developmental Cognitive Neuroscience*, *13*, 53–67.

https://doi.org/10.1016/j.dcn.2015.03.001

- Kasuya-Ueba, Y., Zhao, S., & Toichi, M. (2020). The Effect of Music Intervention on Attention in Children: Experimental Evidence. *Frontiers in Neuroscience*, *14*. https://www.frontiersin.org/articles/10.3389/fnins.2020.00757
- Katsimpokis, D., Hawkins, G. E., & van Maanen, L. (2020). Not all Speed-Accuracy Trade-Off Manipulations Have the Same Psychological Effect. *Computational Brain & Behavior*, *3*(3), 252–268. https://doi.org/10.1007/s42113-020-00074y
- Kaushanskaya, M., Gross, M., & Buac, M. (2014). Effects of classroom bilingualism on task-shifting, verbal memory, and word learning in children. *Developmental Science*, *17*(4), 564–583. https://doi.org/10.1111/desc.12142

Kaushanskaya, M., & Marian, V. (2007). Age-of-Acquisition Effects in the Development of a Bilingual Advantage for Word Learning. 12.

Kaushanskaya, M., & Marian, V. (2009). The bilingual advantage in novel word learning. *Psychonomic Bulletin & Review*, 16(4), 705–710. https://doi.org/10.3758/PBR.16.4.705

- Keitel, A., Gross, J., & Kayser, C. (2018). Perceptually relevant speech tracking in auditory and motor cortex reflects distinct linguistic features. *PLOS Biology*, *16*(3), e2004473. https://doi.org/10.1371/journal.pbio.2004473
- Kerlin, J. R., Shahin, A. J., & Miller, L. M. (2010). Attentional gain control of ongoing cortical speech representations in a 'cocktail party'. *The Journal of Neuroscience: The Official Journal of the Society for Neuroscience*, 30(2), 620–628. https://doi.org/10.1523/JNEUROSCI.3631-09.2010
- Kim, S. Y., Qi, T., Feng, X., Ding, G., Liu, L., & Cao, F. (2016). How does language distance between L1 and L2 affect the L2 brain network? An fMRI study of

Korean–Chinese–English trilinguals. *NeuroImage*, *129*, 25–39.

https://doi.org/10.1016/j.neuroimage.2015.11.068

- Kimura, D. (1967). Functional Asymmetry of the Brain in Dichotic Listening. *Cortex*, *3*(2), 163–178. https://doi.org/10.1016/S0010-9452(67)80010-8
- Klein, D., Milner, B., Zatorre, R. J., Zhao, V., & Nikelski, J. (1999). Cerebral organization in bilinguals: A PET study of Chinese-English verb generation. *NeuroReport*, 10(13), 2841–2845.
- Klein, D., Mok, K., Chen, J.-K., & Watkins, K. E. (2014). Age of language learning shapes brain structure: A cortical thickness study of bilingual and monolingual individuals. *Brain and Language*, *131*, 20–24.
 https://doi.org/10.1016/j.bandl.2013.05.014
- Koch, I., Poljac, E., Müller, H., & Kiesel, A. (2018). Cognitive structure, flexibility, and plasticity in human multitasking—An integrative review of dual-task and taskswitching research. *Psychological Bulletin*, *144*(6), 557–583.

https://doi.org/10.1037/bul0000144

- Konishi, S. (2011). [Frontal lobes and inhibitory function]. *Brain and Nerve = Shinkei Kenkyu No Shinpo*, *63*(12), 1346–1351.
- Kousaie, S., & Phillips, N. A. (2012). Conflict monitoring and resolution: Are two languages better than one? Evidence from reaction time and event-related brain potentials. *Brain Research*, 1446, 71–90. https://doi.org/10.1016/j.brainres.2012.01.052
- Kousaie, S., & Phillips, N. A. (2017). A behavioural and electrophysiological investigation of the effect of bilingualism on aging and cognitive control.

Neuropsychologia, 94, 23–35.

https://doi.org/10.1016/j.neuropsychologia.2016.11.013

Kovács, Á. M., & Mehler, J. (2009). Cognitive gains in 7-month-old bilingual infants. *Proceedings of the National Academy of Sciences*, *106*(16), 6556–6560. https://doi.org/10.1073/pnas.0811323106

Kovelman, I., Baker, S. A., & Petitto, L.-A. (2007). Bilingual and Monolingual Brains
Compared: A Functional Magnetic Resonance Imaging Investigation of
Syntactic Processing and a Possible "Neural Signature" of Bilingualism.
Journal of Cognitive Neuroscience, 20(1), 153–169.

https://doi.org/10.1162/jocn.2008.20011

- Krishnamurthy, N., & Hansen, J. H. L. (2009). Babble Noise: Modeling, Analysis, and Applications. *IEEE Transactions on Audio, Speech, and Language Processing*, *17*(7), 1394–1407. https://doi.org/10.1109/TASL.2009.2015084
- Krizman, J., Bradlow, A., Lam, S., & Kraus, N. (2017). How bilinguals listen in noise:
 Linguistic and non-linguistic factors. *Bilingualism*, *20*(4), 834–843.
 https://doi.org/10.1017/S1366728916000444
- Krizman, J., Marian, V., Shook, A., Skoe, E., & Kraus, N. (2012). Subcortical encoding of sound is enhanced in bilinguals and relates to executive function advantages. *Proceedings of the National Academy of Sciences*, 109(20), 7877–7881. https://doi.org/10.1073/pnas.1201575109
- Krizman, J., Skoe, E., Marian, V., & Kraus, N. (2014). Bilingualism increases neural response consistency and attentional control: Evidence for sensory and cognitive coupling. *Brain and Language*, *128*(1), 34–40. https://doi.org/10.1016/j.bandl.2013.11.006

Kroll, J. F., & Bialystok, E. (2013). Understanding the consequences of bilingualism for language processing and cognition. *Journal of Cognitive Psychology*, 25(5), 497–514. https://doi.org/10.1080/20445911.2013.799170

- Kubanek, J., Brunner, P., Gunduz, A., Poeppel, D., & Schalk, G. (2013). The Tracking of
 Speech Envelope in the Human Cortex. *PLOS ONE*, *8*(1), e53398.
 https://doi.org/10.1371/journal.pone.0053398
- Kuhl, P. K., Stevenson, J., Corrigan, N. M., van den Bosch, J. J. F., Can, D. D., & Richards, T. (2016). Neuroimaging of the bilingual brain: Structural brain correlates of listening and speaking in a second language. *Brain and Language*, *162*, 1–9. https://doi.org/10.1016/j.bandl.2016.07.004
- Kuo, L.-J., & Anderson, R. C. (2012). Effects of early bilingualism on learning phonological regularities in a new language. *Journal of Experimental Child Psychology*, *111*(3), 455–467. https://doi.org/10.1016/j.jecp.2011.08.013
- Kuznetsova, A., Brockhoff, P. B., & Christensen, R. H. B. (2017). ImerTest Package:
 Tests in Linear Mixed Effects Models. *Journal of Statistical Software*, 82(13).
 https://doi.org/10.18637/jss.v082.i13
- Lalor, E. C., & Foxe, J. J. (2010). Neural responses to uninterrupted natural speech can be extracted with precise temporal resolution. *European Journal of Neuroscience*, *31*(1), 189–193. https://doi.org/10.1111/j.1460-9568.2009.07055.x
- Lambert, K., Eisch, A. J., Galea, L. A. M., Kempermann, G., & Merzenich, M. (2019).
 Optimizing brain performance: Identifying mechanisms of adaptive
 neurobiological plasticity. *Neuroscience & Biobehavioral Reviews*, 105, 60–71.
 https://doi.org/10.1016/j.neubiorev.2019.06.033

- Lane, D. M., & Pearson, D. A. (1982). The Development of Selective Attention. *Merrill-Palmer Quarterly*, *28*(3), 317–337.
- Lavie, N. (1995). Perceptual load as a necessary condition for selective attention. Journal of Experimental Psychology: Human Perception and Performance, 21(3), 451–468. https://doi.org/10.1037/0096-1523.21.3.451
- Lavie, N., & Tsal, Y. (1994). Perceptual load as a major determinant of the locus of selection in visual attention. *Perception & Psychophysics*, 56(2), 183–197.
 https://doi.org/10.3758/BF03213897
- Lewis, D. G. (1959). Bilingualism and Non-Verbal Intelligence: A Further Study of Test Results. *British Journal of Educational Psychology*, *29*(1), 17–22. https://doi.org/10.1111/j.2044-8279.1959.tb01470.x
- Li, L., Abutalebi, J., Zou, L., Yan, X., Liu, L., Feng, X., Wang, R., Guo, T., & Ding, G. (2015). Bilingualism alters brain functional connectivity between "control" regions and "language" regions: Evidence from bimodal bilinguals. *Neuropsychologia*, *71*, 236–247.

https://doi.org/10.1016/j.neuropsychologia.2015.04.007

Li, P., Legault, J., & Litcofsky, K. A. (2014). Neuroplasticity as a function of second
 language learning: Anatomical changes in the human brain. *Cortex*, 58, 301–
 324. https://doi.org/10.1016/j.cortex.2014.05.001

Lien, M.-C., Ruthruff, E., Cornett, L., Goodin, Z., & Allen, P. A. (20080526). On the nonautomaticity of visual word processing: Electrophysiological evidence that word processing requires central attention. *Journal of Experimental Psychology: Human Perception and Performance*, *34*(3), 751. https://doi.org/10.1037/0096-1523.34.3.751

- Linck, J. A., Hoshino, N., & Kroll, J. F. (2008). Cross-language lexical processes and inhibitory control. *The Mental Lexicon*, *3*(3), 349–374. https://doi.org/10.1075/ml.3.3.06lin
- Lu, C., & Proctor, R. W. (1995). The influence of irrelevant location information on performance: A review of the Simon and spatial Stroop effects. *Psychonomic Bulletin & Review*, 2(2), 174–207. https://doi.org/10.3758/BF03210959
- Luk, G., Anderson, J. A. E., Craik, F. I. M., Grady, C., & Bialystok, E. (2010). Distinct neural correlates for two types of inhibition in bilinguals: Response inhibition versus interference suppression. *Brain and Cognition*, *74*(3), 347–357. https://doi.org/10.1016/j.bandc.2010.09.004
- Luk, G., Bialystok, E., Craik, F. I. M., & Grady, C. L. (2011). Lifelong Bilingualism
 Maintains White Matter Integrity in Older Adults. *Journal of Neuroscience*, 31(46), 16808–16813. https://doi.org/10.1523/JNEUROSCI.4563-11.2011
- Luk, G., Sa, E. D., & Bialystok, E. (2011). Is there a relation between onset age of bilingualism and enhancement of cognitive control?. *Bilingualism: Language* and Cognition, 14(4), 588–595. https://doi.org/10.1017/S1366728911000010
- Luna, B., Thulborn, K. R., Munoz, D. P., Merriam, E. P., Garver, K. E., Minshew, N. J.,
 Keshavan, M. S., Genovese, C. R., Eddy, W. F., & Sweeney, J. A. (2001).
 Maturation of Widely Distributed Brain Function Subserves Cognitive
 Development. *NeuroImage*, *13*(5), 786–793.
 https://doi.org/10.1006/nimg.2000.0743
- Maccoby, E. E., & Konrad, K. W. (1966). Age trends in selective listening. *Journal of Experimental Child Psychology*, *3*(2), 113–122. https://doi.org/10.1016/0022-0965(66)90086-5

MacDonald, A. W., Cohen, J. D., Stenger, V. A., & Carter, C. S. (2000). Dissociating the Role of the Dorsolateral Prefrontal and Anterior Cingulate Cortex in Cognitive Control. *Science*, *288*(5472), 1835–1838.

https://doi.org/10.1126/science.288.5472.1835

- Machens, C. K. (2004). Linearity of Cortical Receptive Fields Measured with Natural Sounds. *Journal of Neuroscience*, *24*(5), 1089–1100. https://doi.org/10.1523/JNEUROSCI.4445-03.2004
- Mack, A., & Rock, I. (1998). Inattentional blindness: Perception without attention. In *Visual attention* (pp. 55–76).
- Macnamara, J. T. (1966). *Bilingualism and primary education: A study of Irish experience.* Edinburgh : Edinburgh University Press, 1966.
- Mai, G., Minett, J. W., & Wang, W. S.-Y. (2016). Delta, theta, beta, and gamma brain oscillations index levels of auditory sentence processing. *NeuroImage*, 133, 516–528. https://doi.org/10.1016/j.neuroimage.2016.02.064
- Majovski, L. V. (1989). Higher Cortical Functions in Children. In C. R. Reynolds & E. Fletcher-Janzen (Eds.), *Handbook of Clinical Child Neuropsychology* (pp. 41– 67). Springer US. https://doi.org/10.1007/978-1-4899-6807-4_3
- Mangun, G. R., Hillyard, S. A., & Luck, S. J. (1993). Electrocortical substrates of visual selective attention. In *Attention and performance 14: Synergies in experimental psychology, artificial intelligence, and cognitive neuroscience* (pp. 219–243). The MIT Press.
- Manuel, H. T. (1935). A comparison of Spanish-speaking and English-speaking children in reading and arithmetic. *Journal of Applied Psychology*, *19*(2), 189– 202. https://doi.org/10.1037/h0060003
- Mason, P. H., Domínguez D., J. F., Winter, B., & Grignolio, A. (2015). Hidden in plain view: Degeneracy in complex systems. *Biosystems*, 128, 1–8. https://doi.org/10.1016/j.biosystems.2014.12.003
- Matusz, P. J., Dikker, S., Huth, A. G., & Perrodin, C. (2018). Are We Ready for Realworld Neuroscience? *Journal of Cognitive Neuroscience*, *31*(3), 327–338. https://doi.org/10.1162/jocn_e_01276
- Mayo, L. H., Florentine, M., & Buus, S. (1997). Age of Second-Language Acquisition and Perception of Speech in Noise. *J Speech Lang Hear Res*, *40*(3), 686–693. https://doi.org/10.1044/jslhr.4003.686
- McAuley, J. D., Shen, Y., Dec, S., & Kidd, G. R. (2020). Altering the rhythm of target and background talkers differentially affects speech understanding. *Attention, Perception, & Psychophysics, 82*(6), 3222–3233.
 https://doi.org/10.3758/s13414-020-02064-5
- McNealy, K., Mazziotta, J. C., & Dapretto, M. (2011). Age and experience shape developmental changes in the neural basis of language-related learning. *Developmental Science*, 14(6), 1261–1282. https://doi.org/10.1111/j.1467-7687.2011.01075.x
- Mechelli, A., Crinion, J. T., Noppeney, U., O'Doherty, J., Ashburner, J., Frackowiak, R. S., & Price, C. J. (2004). Structural plasticity in the bilingual brain. *Nature*, *431*(7010), 757. https://doi.org/10.1038/431757a
- Mesgarani, N., & Chang, E. F. (2012). Selective cortical representation of attended speaker in multi-talker speech perception. *Nature*, *485*(7397). https://doi.org/10.1038/nature11020

Meyers, L. A., Ancel, F. D., & Lachmann, M. (2005). Evolution of Genetic Potential. PLOS Computational Biology, 1(3), e32.

https://doi.org/10.1371/journal.pcbi.0010032

- Mezzacappa, E. (2004). Alerting, Orienting, and Executive Attention: Developmental Properties and Sociodemographic Correlates in an Epidemiological Sample of Young, Urban Children. *Child Development*, *75*(5), 1373–1386. https://doi.org/10.1111/j.1467-8624.2004.00746.x
- Millman, R. E., Johnson, S. R., & Prendergast, G. (2014). The Role of Phase-locking to the Temporal Envelope of Speech in Auditory Perception and Speech Intelligibility. *Journal of Cognitive Neuroscience*, *27*(3), 533–545. https://doi.org/10.1162/jocn_a_00719
- Mohades, S. G., Struys, E., Van Schuerbeek, P., Mondt, K., Van De Craen, P., & Luypaert, R. (2012). DTI reveals structural differences in white matter tracts between bilingual and monolingual children. *Brain Research*, *1435*, 72–80. https://doi.org/10.1016/j.brainres.2011.12.005
- Moll, K., Snowling, M. J., Göbel, S. M., & Hulme, C. (2015). Early language and executive skills predict variations in number and arithmetic skills in children at family-risk of dyslexia and typically developing controls. *Learning and Instruction, 38*, 53–62. https://doi.org/10.1016/j.learninstruc.2015.03.004
- Monsell, S., & Driver, J. (2000). *Control of Cognitive Processes: Attention and Performance XVIII* (Vol. 18). MIT Press.
- Moray, N. (1959). Attention in dichotic listening: Affective cues and the influence of instructions. *Quarterly Journal of Experimental Psychology*, *11*(1), 56–60. https://doi.org/10.1080/17470215908416289

Moray, N. (1967). Where is capacity limited? A survey and a model. *Acta Psychologica*, *27*, 84–92. https://doi.org/10.1016/0001-6918(67)90048-0

Morton, J. B., & Harper, S. N. (2007). What did Simon say? Revisiting the bilingual advantage. *Developmental Science*, *10*(6), 719–726. https://doi.org/10.1111/j.1467-7687.2007.00623.x

Mowbray, G. H. (1953). Simultaneous vision and audition: The comprehension of prose passages with varying levels of difficulty. *Journal of Experimental Psychology*, *46*(5), 365. https://doi.org/10.1037/h0054574

Mueller, V., Brehmer, Y., von Oertzen, T., Li, S.-C., & Lindenberger, U. (2008).
 Electrophysiological correlates of selective attention: A lifespan comparison.
 BMC Neuroscience, 9(1), 18. https://doi.org/10.1186/1471-2202-9-18

Mulder, H., Pitchford, N. J., Hagger, M. S., & Marlow, N. (2009). Development of Executive Function and Attention in Preterm Children: A Systematic Review. *Developmental Neuropsychology*, *34*(4), 393–421.

https://doi.org/10.1080/87565640902964524

Musso, M., Moro, A., Glauche, V., Rijntjes, M., Reichenbach, J., Büchel, C., & Weiller,
C. (2003). Broca's area and the language instinct. *Nature Neuroscience*, 6(7),
Article 7. https://doi.org/10.1038/nn1077

Nacar Garcia, L., Guerrero-Mosquera, C., Colomer, M., & Sebastian-Galles, N. (2018).
 Evoked and oscillatory EEG activity differentiates language discrimination in young monolingual and bilingual infants. *Scientific Reports*, 8(1), Article 1.
 https://doi.org/10.1038/s41598-018-20824-0

Naeem, K., Filippi, R., Periche-Tomas, E., Papageorgiou, A., & Bright, P. (2018). The Importance of Socioeconomic Status as a Modulator of the Bilingual Advantage in Cognitive Ability. Frontiers in Psychology, 9.

https://www.frontiersin.org/articles/10.3389/fpsyg.2018.01818

- Nagaraj, N. K., Magimairaj, B. M., & Schwartz, S. (2020). Auditory distraction in school-age children relative to individual differences in working memory capacity. *Attention, Perception, & Psychophysics, 82*(7), 3581–3593. https://doi.org/10.3758/s13414-020-02056-5
- Namazi, M., & Thordardottir, E. (2010). A working memory, not bilingual advantage, in controlled attention. *International Journal of Bilingual Education and Bilingualism*, 13(5), 597–616.

https://doi.org/10.1080/13670050.2010.488288

- Navarro-Torres, C. A., Beatty-Martínez, A. L., Kroll, J. F., & Green, D. W. (2021). Research on bilingualism as discovery science. *Brain and Language*, 222, 105014. https://doi.org/10.1016/j.bandl.2021.105014
- Navon, D., & Gopher, D. (1979). On the economy of the human-processing system. *Psychological Review*, *86*(3), 214–255. https://doi.org/10.1037/0033-295X.86.3.214
- Neville, H. J., Stevens, C., Pakulak, E., Bell, T. A., Fanning, J., Klein, S., & Isbell, E.
 (2013). Family-based training program improves brain function, cognition, and behavior in lower socioeconomic status preschoolers. *Proceedings of the National Academy of Sciences of the United States of America*, 110(29), 12138–12143. https://doi.org/10.1073/pnas.1304437110
- Nichols, E. S., & Joanisse, M. F. (2016). Functional activity and white matter microstructure reveal the independent effects of age of acquisition and

proficiency on second-language learning. *NeuroImage*, *143*, 15–25. https://doi.org/10.1016/j.neuroimage.2016.08.053

- Nichols, E. S., Wild, C. J., Stojanoski, B., Battista, M. E., & Owen, A. M. (2020).
 Bilingualism Affords No General Cognitive Advantages: A Population Study of Executive Function in 11,000 People. *Psychological Science*, *31*(5), 548–567.
 https://doi.org/10.1177/0956797620903113
- Norman, D. A. (1968). Toward a theory of memory and attention. *Psychological Review*, *75*, 522–536. https://doi.org/10.1037/h0026699
- Norman, D. A., & Bobrow, D. G. (1975). On data-limited and resource-limited processes. *Cognitive Psychology*, 7(1), 44–64. https://doi.org/10.1016/0010-0285(75)90004-3
- Nunez, P. L., Nunez, E. P. of B. E. P. L., & Srinivasan, R. (2006). *Electric Fields of the Brain: The Neurophysics of EEG*. Oxford University Press.
- Obleser, J., & Kayser, C. (2019). Neural Entrainment and Attentional Selection in the Listening Brain. *Trends in Cognitive Sciences*, *23*(11), 913–926. https://doi.org/10.1016/j.tics.2019.08.004

Oever, S. ten, Schroeder, C. E., Poeppel, D., Atteveldt, N. van, Mehta, A. D.,
 Mégevand, P., Groppe, D. M., & Zion-Golumbic, E. (2017). Low-Frequency
 Cortical Oscillations Entrain to Subthreshold Rhythmic Auditory Stimuli.
 Journal of Neuroscience, *37*(19), 4903–4912.

https://doi.org/10.1523/JNEUROSCI.3658-16.2017

Olguin, A., Bekinschtein, T. A., & Bozic, M. (2018). Neural Encoding of Attended Continuous Speech under Different Types of Interference. *Journal of* *Cognitive Neuroscience*, *30*(11), 1606–1619.

https://doi.org/10.1162/jocn_a_01303

- Olguin, A., Cekic, M., Bekinschtein, T. A., Katsos, N., & Bozic, M. (2019). Bilingualism and language similarity modify the neural mechanisms of selective attention. *Scientific Reports*, *9*(1), 8204. https://doi.org/10.1038/s41598-019-44782-3
- Oliveira, A. M., Castro, S. L., & Sousa, L. de. (1997). Verbal information processing in bilinguals (portuguese/french) in a dichotic listening task (bilingualism). https://repositorio-aberto.up.pt/handle/10216/22066
- Oller, D. K., Pearson, B. Z., & Cobo-Lewis, A. B. (2007). Profile effects in early bilingual language and literacy. *Applied Psycholinguistics*, *28*(2), 191–230. https://doi.org/10.1017/S0142716407070117
- Osman, A., & Moore, C. M. (1993). The locus of dual-task interference: Psychological refractory effects on movement-related brain potentials. *Journal of Experimental Psychology: Human Perception and Performance, 19*(6), 1292– 1312. https://doi.org/10.1037/0096-1523.19.6.1292
- O'Sullivan, A. E., Crosse, M. J., Di Liberto, G. M., & Lalor, E. C. (2017). Visual Cortical Entrainment to Motion and Categorical Speech Features during Silent Lipreading. *Frontiers in Human Neuroscience*, *10*. https://doi.org/10.3389/fnhum.2016.00679

O'Sullivan, J. A., Power, A. J., Mesgarani, N., Rajaram, S., Foxe, J. J., Shinn-Cunningham, B. G., Slaney, M., Shamma, S. A., & Lalor, E. C. (2015). Attentional Selection in a Cocktail Party Environment Can Be Decoded from Single-Trial EEG. *Cerebral Cortex (New York, N.Y.: 1991), 25*(7), 1697–1706. https://doi.org/10.1093/cercor/bht355 Paap, K. R. (2016). The neuroanatomy of bilingualism: Will winds of change lift the fog? Language, Cognition and Neuroscience, 31(3), 331–334. https://doi.org/10.1080/23273798.2015.1082607

- Paap, K. R., & Greenberg, Z. I. (2013). There is no coherent evidence for a bilingual advantage in executive processing. *Cognitive Psychology*, 66(2), 232–258. https://doi.org/10.1016/j.cogpsych.2012.12.002
- Paap, K. R., Johnson, H. A., & Sawi, O. (2014). Are bilingual advantages dependent upon specific tasks or specific bilingual experiences? *Journal of Cognitive Psychology*, *26*(6), 615–639. https://doi.org/10.1080/20445911.2014.944914

Paap, K. R., Johnson, H. A., & Sawi, O. (2015). Bilingual advantages in executive functioning either do not exist or are restricted to very specific and undetermined circumstances. *Cortex*, 69, 265–278. https://doi.org/10.1016/j.cortex.2015.04.014

- Palomar-García, M.-Á., Bueichekú, E., Ávila, C., Sanjuán, A., Strijkers, K., Ventura-Campos, N., & Costa, A. (2015). Do bilinguals show neural differences with monolinguals when processing their native language? *Brain and Language*, *142*, 36–44. https://doi.org/10.1016/j.bandl.2015.01.004
- Paradis, J. (2007). Bilingual children with specific language impairment: Theoretical and applied issues. *Applied Psycholinguistics*, *28*(3), 551–564. https://doi.org/10.1017/S0142716407070300
- Pashler, H. (1994). Dual-task interference in simple tasks: Data and theory. *Psychological Bulletin*, *116*(2), 220. https://doi.org/10.1037/0033-2909.116.2.220

Pashler, H., & Johnston, J. C. (1998). Attentional limitations in dual-task performance. In *Attention* (pp. 155–189). Psychology Press/Erlbaum (UK) Taylor & Francis.

Pasley, B. N., David, S. V., Mesgarani, N., Flinker, A., Shamma, S. A., Crone, N. E.,
Knight, R. T., & Chang, E. F. (2012). Reconstructing Speech from Human
Auditory Cortex. *PLOS Biology*, *10*(1), e1001251.
https://doi.org/10.1371/journal.pbio.1001251

Passler, M. A., Isaac, W., & Hynd, G. W. (1985). Neuropsychological development of behavior attributed to frontal lobe functioning in children. *Developmental Neuropsychology*, 1(4), 349–370.

https://doi.org/10.1080/87565648509540320

- Peal, E., & Lambert, W. E. (1962). The relation of bilingualism to intelligence.
 Psychological Monographs: General and Applied, 76(27), 1–23.
 https://doi.org/10.1037/h0093840
- Pearson, B. Z., Fernández, S. C., & Oller, D. K. (1993). Lexical Development in Bilingual Infants and Toddlers: Comparison to Monolingual Norms. *Language Learning*, 43(1), 93–120. https://doi.org/10.1111/j.1467-1770.1993.tb00174.x
- Peelle, J. E., Gross, J., & Davis, M. H. (2013). Phase-Locked Responses to Speech in Human Auditory Cortex are Enhanced During Comprehension. *Cerebral Cortex*, 23(6), 1378–1387. https://doi.org/10.1093/cercor/bhs118
- Pelham, S. D., & Abrams, L. (2014). Cognitive advantages and disadvantages in early and late bilinguals. *Journal of Experimental Psychology: Learning, Memory, and Cognition*, 40(2), 313–325. https://doi.org/10.1037/a0035224

Perani, D. (2005). The neural basis of language talent in bilinguals. *Trends in Cognitive Sciences*, *9*(5), 211–213. https://doi.org/10.1016/j.tics.2005.03.001

- Perani, D., & Abutalebi, J. (2005). The neural basis of first and second language processing. *Current Opinion in Neurobiology*, *15*(2), 202–206. https://doi.org/10.1016/j.conb.2005.03.007
- Petitto, L. A., Berens, M. S., Kovelman, I., Dubins, M. H., Jasińska, K., & Shalinsky, M.
 (2012). The "Perceptual Wedge Hypothesis" as the basis for bilingual babies' phonetic processing advantage: New insights from fNIRS brain imaging. *Brain and Language*, *121*(2), 130–143. https://doi.org/10.1016/j.bandl.2011.05.003
- Phelps, J., Attaheri, A., & Bozic, M. (2022). How bilingualism modulates selective attention in children. *Scientific Reports*, 12(1), Article 1. https://doi.org/10.1038/s41598-022-09989-x
- Picton, T. W., Bentin, S., Berg, P., Donchin, E., Hillyard, S. A., Johnson, R., Miller, G. A., Ritter, W., Ruchkin, D. S., Rugg, M. D., & Taylor, M. J. (2000). Guidelines for using human event-related potentials to study cognition: Recording standards and publication criteria. *Psychophysiology*, *37*(2), 127–152.
- Pizzagalli, D. A. (2007). Electroencephalography and high-density electrophysiological source localization. In *Handbook of psychophysiology, 3rd ed* (pp. 56–84).
 Cambridge University Press.

https://doi.org/10.1017/CBO9780511546396.003

Pliatsikas, C., Meteyard, L., Veríssimo, J., DeLuca, V., Shattuck, K., & Ullman, M. T.
(2020). The effect of bilingualism on brain development from early childhood to young adulthood. *Brain Structure and Function*, 225(7), 2131–2152.
https://doi.org/10.1007/s00429-020-02115-5

Plude, D. J., Enns, J. T., & Brodeur, D. (1994). The development of selective attention:
A life-span overview. Acta Psychologica, 86(2), 227–272.
https://doi.org/10.1016/0001-6918(94)90004-3

Poarch, G. J., & van Hell, J. G. (2012). Executive functions and inhibitory control in multilingual children: Evidence from second-language learners, bilinguals, and trilinguals. *Journal of Experimental Child Psychology*, *113*(4), 535–551. https://doi.org/10.1016/j.jecp.2012.06.013

Poeppel, D., & Assaneo, M. F. (2020). Speech rhythms and their neural foundations. *Nature Reviews Neuroscience*, *21*(6), 322–334. https://doi.org/10.1038/s41583-020-0304-4

Pollack, I., & Pickett, J. M. (1958). Stereophonic Listening and Speech Intelligibility against Voice Babble. *The Journal of the Acoustical Society of America*, *30*(2), 131–133. https://doi.org/10.1121/1.1909505

Ponton, C., Eggermont, J. J., Khosla, D., Kwong, B., & Don, M. (2002). Maturation of human central auditory system activity: Separating auditory evoked potentials by dipole source modeling. *Clinical Neurophysiology*, *113*(3), 407– 420. https://doi.org/10.1016/S1388-2457(01)00733-7

Ponton, C. W., Eggermont, J. J., Kwong, B., & Don, M. (2000). Maturation of human central auditory system activity: Evidence from multi-channel evoked potentials. *Clinical Neurophysiology*, *111*(2), 220–236. https://doi.org/10.1016/S1388-2457(99)00236-9

Poulin-Dubois, D., Blaye, A., Coutya, J., & Bialystok, E. (2011). The effects of bilingualism on toddlers' executive functioning. *Journal of Experimental Child Psychology*, *108*(3), 567–579. https://doi.org/10.1016/j.jecp.2010.10.009

Power, A. J., Colling, L. J., Mead, N., Barnes, L., & Goswami, U. (2016). Neural encoding of the speech envelope by children with developmental dyslexia. *Brain and Language*, 160, 1–10. https://doi.org/10.1016/j.bandl.2016.06.006

Power, A. J., Foxe, J. J., Forde, E.-J., Reilly, R. B., & Lalor, E. C. (2012). At what time is the cocktail party? A late locus of selective attention to natural speech. *European Journal of Neuroscience*, 35(9), 1497–1503.
https://doi.org/10.1111/j.1460-9568.2012.08060.x

Power, A. J., Mead, N., Barnes, L., & Goswami, U. (2013). Neural entrainment to rhythmic speech in children with developmental dyslexia. *Frontiers in Human Neuroscience*, *7*. https://doi.org/10.3389/fnhum.2013.00777

Prencipe, A., Kesek, A., Cohen, J., Lamm, C., Lewis, M. D., & Zelazo, P. D. (2011).
Development of hot and cool executive function during the transition to adolescence. *Journal of Experimental Child Psychology*, *108*(3), 621–637.
https://doi.org/10.1016/j.jecp.2010.09.008

Prior, A., & Gollan, T. H. (2011). Good Language-Switchers are Good Task-Switchers:
Evidence from Spanish–English and Mandarin–English Bilinguals. *Journal of the International Neuropsychological Society*, *17*(4), 682–691.
https://doi.org/10.1017/S1355617711000580

- Reck, S. G., & Hund, A. M. (2011). Sustained attention and age predict inhibitory control during early childhood. *Journal of Experimental Child Psychology*, *108*(3), 504–512. https://doi.org/10.1016/j.jecp.2010.07.010
- Reiss, A. L., Abrams, M. T., Singer, H. S., Ross, J. L., & Denckla, M. B. (1996). Brain development, gender and IQ in children: a volumetric imaging study. *Brain*, *119*(5), 1763–1774. https://doi.org/10.1093/brain/119.5.1763

- Ridderinkhof, K. R., & van der Stelt, O. (2000). Attention and selection in the growing child: views derived from developmental psychophysiology. *Biological Psychology*, *54*(1), 55–106. https://doi.org/10.1016/S0301-0511(00)00053-3
- Riecke, L., Formisano, E., Sorger, B., Başkent, D., & Gaudrain, E. (2018). Neural
 Entrainment to Speech Modulates Speech Intelligibility. *Current Biology*, 28(2), 161-169.e5. https://doi.org/10.1016/j.cub.2017.11.033
- Rimmele, J. M., Zion Golumbic, E., Schröger, E., & Poeppel, D. (2015). The effects of selective attention and speech acoustics on neural speech-tracking in a multi-talker scene. *Cortex*, *68*, 144–154.

https://doi.org/10.1016/j.cortex.2014.12.014

Ríos-López, P., Molinaro, N., Bourguignon, M., & Lallier, M. (2020). Development of neural oscillatory activity in response to speech in children from 4 to 6 years old. *Developmental Science*, 23(6), e12947.

https://doi.org/10.1111/desc.12947

Roberts, K. L., & Hall, D. A. (2008). Examining a Supramodal Network for Conflict
Processing: A Systematic Review and Novel Functional Magnetic Resonance
Imaging Data for Related Visual and Auditory Stroop Tasks. *Journal of Cognitive Neuroscience*, 20(6), 1063–1078.
https://doi.org/10.1162/jocn.2008.20074

- Rock, I., & Gutman, D. (1981). The effect of inattention on form perception. *Journal* of Experimental Psychology: Human Perception and Performance, 7, 275–285. https://doi.org/10.1037/0096-1523.7.2.275
- Rogers, C. L., Lister, J. J., Febo, D. M., Besing, J. M., & Abrams, H. B. (2006). Effects of bilingualism, noise, and reverberation on speech perception by listeners with

normal hearing. Applied Psycholinguistics, 27(3), 465–485.

https://doi.org/10.1017/S014271640606036X

- Rollins, H. A., & Hendricks, R. (1980). Processing of words presented simultaneously to eye and ear. *Journal of Experimental Psychology: Human Perception and Performance*, 6(1), 99–109. https://doi.org/10.1037/0096-1523.6.1.99
- Román, P., González, J., Ventura-Campos, N., Rodríguez-Pujadas, A., Sanjuán, A., &
 Ávila, C. (2015). Neural differences between monolinguals and early
 bilinguals in their native language during comprehension. *Brain and Language*, 150, 80–89. https://doi.org/10.1016/j.bandl.2015.07.011
- Romine, C. B., & Reynolds, C. R. (2005). A Model of the Development of Frontal Lobe Functioning: Findings From a Meta-Analysis. *Applied Neuropsychology*, *12*(4), 190–201. https://doi.org/10.1207/s15324826an1204 2
- Rueda, M. R., Fan, J., McCandliss, B. D., Halparin, J. D., Gruber, D. B., Lercari, L. P., & Posner, M. I. (2004). Development of attentional networks in childhood. *Neuropsychologia*, *42*(8), 1029–1040.

https://doi.org/10.1016/j.neuropsychologia.2003.12.012

- Ruff, H. A., Capozzoli, M., & Weissberg, R. (1998). Age, individuality, and context as factors in sustained visual attention during the preschool years.
 Developmental Psychology, 34(3), 454–464. https://doi.org/10.1037/0012-1649.34.3.454
- Saer, D. J. (1923). The Effect of Bilingualism on Intellligence. British Journal of Psychology. General Section, 14(1), 25–38. https://doi.org/10.1111/j.2044-8295.1923.tb00110.x

Sakai, K. L., Miura, K., Narafu, N., & Muraishi, Y. (2004). Correlated Functional Changes of the Prefrontal Cortex in Twins Induced by Classroom Education of Second Language. *Cerebral Cortex*, 14(11), 1233–1239. https://doi.org/10.1093/cercor/bhh084

Salo, E., Salmela, V., Salmi, J., Numminen, J., & Alho, K. (2017). Brain activity associated with selective attention, divided attention and distraction. *Brain Research*, *1664*, 25–36. https://doi.org/10.1016/j.brainres.2017.03.021

Samuel, S., Roehr-Brackin, K., Pak, H., & Kim, H. (2018). Cultural Effects Rather Than
 a Bilingual Advantage in Cognition: A Review and an Empirical Study.
 Cognitive Science A Multidisciplinary Journal, 42, 2313–2341.
 https://doi.org/10.1111/COGS.12672

Sanders, L. D., Stevens, C., Coch, D., & Neville, H. J. (2006). Selective auditory attention in 3- to 5-year-old children: An event-related potential study. *Neuropsychologia*, 44(11), 2126–2138.

https://doi.org/10.1016/j.neuropsychologia.2005.10.007

Sassenhagen, J. (2019). How to analyse electrophysiological responses to naturalistic language with time-resolved multiple regression. *Language, Cognition and Neuroscience*, *34*(4), 474–490.

https://doi.org/10.1080/23273798.2018.1502458

Satterthwaite, F. E. (1946). An Approximate Distribution of Estimates of Variance Components. *Biometrics Bulletin*, 2(6), 110–114.

https://doi.org/10.2307/3002019

Schubert, T. (1999). Processing differences between simple and choice reactions affect bottleneck localization in overlapping tasks. *Journal of Experimental* *Psychology: Human Perception and Performance, 25*(2), 408–425. https://doi.org/10.1037/0096-1523.25.2.408

Schubert, T. (2008). The central attentional limitation and executive control. *Frontiers in Bioscience: A Journal and Virtual Library, 13,* 3569–3580. https://doi.org/10.2741/2950

Sebastián-Gallés, N., Albareda-Castellot, B., Weikum, W. M., & Werker, J. F. (2012). A Bilingual Advantage in Visual Language Discrimination in Infancy. *Psychological Science*, 23(9), 994–999.

https://doi.org/10.1177/0956797612436817

Segalowitz, S. J., & Davies, P. L. (2004). Charting the maturation of the frontal lobe: An electrophysiological strategy. *Brain and Cognition*, *55*(1), 116–133. https://doi.org/10.1016/S0278-2626(03)00283-5

Seifert, L., Komar, J., Araújo, D., & Davids, K. (2016). Neurobiological degeneracy: A key property for functional adaptations of perception and action to constraints. *Neuroscience & Biobehavioral Reviews*, 69, 159–165. https://doi.org/10.1016/j.neubiorev.2016.08.006

- Shanna Kousaie, Natalie A. Phillips. (2012). *Ageing and bilingualism: Absence of a "bilingual advantage" in Stroop interference in a nonimmigrant sample*. https://journals.sagepub.com/doi/full/10.1080/17470218.2011.604788?casa _token=IQrAQSCjhmkAAAAA%3Azt0nOfMyCk_IV3oBMKZayD9clnJ9PNSHNE5 uhyT8GI_Mzvk9BMx37GdL7ojr2nLifSr8yXnO4B6i
- Shi, L.-F. (2010). Perception of Acoustically Degraded Sentences in Bilingual Listeners Who Differ in Age of English Acquisition. *Journal of Speech, Language, and*

Hearing Research, 53(4), 821–835. https://doi.org/10.1044/1092-

4388(2010/09-0081)

https://doi.org/10.1016/j.tics.2008.02.003

- Shomstein, S., & Yantis, S. (2004). Control of Attention Shifts between Vision and Audition in Human Cortex. *Journal of Neuroscience*, *24*(47), 10702–10706. https://doi.org/10.1523/JNEUROSCI.2939-04.2004
- Shook, A., & Marian, V. (2019). Covert co-activation of bilinguals' non-target language: Phonological competition from translations. *Linguistic Approaches to Bilingualism*, 9(2), 228–252. https://doi.org/10.1075/lab.17022.sho
- Skoe, E., Burakiewicz, E., Figueiredo, M., & Hardin, M. (2017). Basic neural processing of sound in adults is influenced by bilingual experience. *Neuroscience*, *349*, 278–290. https://doi.org/10.1016/j.neuroscience.2017.02.049
- Sörman, D. E., Josefsson, M., Marsh, J. E., Hansson, P., & Ljungberg, J. K. (2017).
 Longitudinal effects of bilingualism on dual-tasking. *PLOS ONE*, *12*(12),
 e0189299. https://doi.org/10.1371/journal.pone.0189299
- Sowell, E. R., Delis, D., Stiles, J., & Jernigan, T. L. (2001). Improved memory functioning and frontal lobe maturation between childhood and adolescence: A structural MRI study. *Journal of the International Neuropsychological Society*, 7(3), 312–322. https://doi.org/10.1017/S135561770173305X
- Sowell, E. R., Thompson, P. M., Tessner, K. D., & Toga, A. W. (2001). Mapping Continued Brain Growth and Gray Matter Density Reduction in Dorsal Frontal Cortex: Inverse Relationships during Postadolescent Brain Maturation.

Shinn-Cunningham, B. G. (2008). Object-based auditory and visual attention. *Trends in Cognitive Sciences*, *12*(5), 182–186.

Journal of Neuroscience, 21(22), 8819–8829.

https://doi.org/10.1523/JNEUROSCI.21-22-08819.2001

- Sperling, G. (1960). The information available in brief visual presentations. *Psychological Monographs: General and Applied*, 74, 1–29. https://doi.org/10.1037/h0093759
- Standage, D., You, H., Wang, D., & Dorris, M. (2011). Gain Modulation by an Urgency Signal Controls the Speed–Accuracy Trade-Off in a Network Model of a Cortical Decision Circuit. *Frontiers in Computational Neuroscience*, 5. https://www.frontiersin.org/article/10.3389/fncom.2011.00007
- Stevens, C., & Bavelier, D. (2012). The role of selective attention on academic foundations: A cognitive neuroscience perspective. *Developmental Cognitive Neuroscience*, 2, S30–S48. https://doi.org/10.1016/j.dcn.2011.11.001
- Stevens, C., Lauinger, B., & Neville, H. (2009). Differences in the neural mechanisms of selective attention in children from different socioeconomic backgrounds:
 An event-related brain potential study. *Developmental Science*, *12*(4), 634–646. https://doi.org/10.1111/j.1467-7687.2009.00807.x
- Stoet, G. (2010). PsyToolkit: A software package for programming psychological experiments using Linux. *Behavior Research Methods*, 42(4), 1096–1104. https://doi.org/10.3758/BRM.42.4.1096
- Stoet, G. (2017). PsyToolkit: A Novel Web-Based Method for Running Online
 Questionnaires and Reaction-Time Experiments. *Teaching of Psychology*,
 44(1), 24–31. https://doi.org/10.1177/0098628316677643

- Strobach, T., Schütz, A., & Schubert, T. (2015). On the importance of Task 1 and error performance measures in PRP dual-task studies. *Frontiers in Psychology*, *6*. https://www.frontiersin.org/articles/10.3389/fpsyg.2015.00403
- Struys, E., Mohades, G., Bosch, P., & van den Noort, M. (2015). Cognitive control in bilingual children: Disentangling the effects of second-language proficiency and onset age of acquisition. *Swiss Journal of Psychology*, 74, 65–73. https://doi.org/10.1024/1421-0185/a000152
- Stumpf, C. (1895). Hermann Von Helmholtz and the new psychology. *Psychological Review*, *2*, 1–12. https://doi.org/10.1037/h0070633
- Swain, M., & Cummins, J. (1979). Bilingualism, Cognitive Functioning and Education. Language Teaching, 12(1), 4–18. https://doi.org/10.1017/S0261444800003918
- Takio, F., Koivisto, M., Jokiranta, L., Rashid, F., Kallio, J., Tuominen, T., Laukka, S. J., &
 Hämäläinen, H. (2009). The Effect of Age on Attentional Modulation in
 Dichotic Listening. *Developmental Neuropsychology*, *34*(3), 225–239.
 https://doi.org/10.1080/87565640902805669
- Tao, L., Marzecová, A., Taft, M., Asanowicz, D., & Wodniecka, Z. (2011). The
 Efficiency of Attentional Networks in Early and Late Bilinguals: The Role of
 Age of Acquisition. *Frontiers in Psychology*, *2*.
 https://www.frontiersin.org/articles/10.3389/fpsyg.2011.00123
- Tao, L., Wang, G., Zhu, M., & Cai, Q. (2021). Bilingualism and domain-general cognitive functions from a neural perspective: A systematic review. *Neuroscience & Biobehavioral Reviews*, 125, 264–295.
 https://doi.org/10.1016/j.neubiorev.2021.02.029

Telford, C. W. (1931). The refractory phase of voluntary and associative responses. Journal of Experimental Psychology, 14(1), 1–36. https://doi.org/10.1037/h0073262

Telner, J. A., Wiesenthal, D. L., Bialystok, E., & York, M. (2008). Is There a Bilingual Advantage When Driving and Speaking Over a Cellular Telephone? *Proceedings of the Human Factors and Ergonomics Society Annual Meeting*, 52(23), 1905–1909. https://doi.org/10.1177/154193120805202316

Tettamanti, M., Alkadhi, H., Moro, A., Perani, D., Kollias, S., & Weniger, D. (2002). Neural Correlates for the Acquisition of Natural Language Syntax. *NeuroImage*, *17*(2), 700–709. https://doi.org/10.1006/nimg.2002.1201

Thatcher, R. W. (1991). Maturation of the human frontal lobes: Physiological evidence for staging. *Developmental Neuropsychology*, 7(3), 397–419. https://doi.org/10.1080/87565649109540500

Theunissen, F. E., David, S. V., Singh, N. C., Hsu, A., Vinje, W. E., & Gallant, J. L. (2001). Estimating spatio-temporal receptive fields of auditory and visual neurons from their responses to natural stimuli. *Network: Computation in Neural Systems*, 12(3), 289–316. https://doi.org/10.1080/net.12.3.289.316

- Tian, T., Olson, S., Whitacre, J. M., & Harding, A. (2011). The origins of cancer robustness and evolvability. *Integrative Biology*, 3(1), 17–30. https://doi.org/10.1039/c0ib00046a
- Timmer, K., Wodniecka, Z., & Costa, A. (2021). Rapid attentional adaptations due to language (monolingual vs bilingual) context. *Neuropsychologia*, 159, 107946. https://doi.org/10.1016/j.neuropsychologia.2021.107946

Tipper, S. P. (1985). The negative priming effect: Inhibitory priming by ignored objects. *The Quarterly Journal of Experimental Psychology Section A*, 37(4), 571–590. https://doi.org/10.1080/14640748508400920

Titchener, E. B. (1910). Attention as Sensory Clearness. *The Journal of Philosophy, Psychology and Scientific Methods*, 7(7), 180–182.

https://doi.org/10.2307/2010783

Tononi, G., Sporns, O., & Edelman, G. M. (1999). Measures of degeneracy and redundancy in biological networks. *Proceedings of the National Academy of Sciences*, *96*(6), 3257–3262. https://doi.org/10.1073/pnas.96.6.3257

Treisman, A. M. (1960). Contextual cues in selective listening. *Quarterly Journal of Experimental Psychology*, 12(4), 242–248.

https://doi.org/10.1080/17470216008416732

Treisman, A. M. (1969). Strategies and models of selective attention. *Psychological Review*, *76*(3), 282–299. https://doi.org/10.1037/h0027242

Treisman, A. M., & Gelade, G. (1980). A feature-integration theory of attention. *Cognitive Psychology*, *12*(1), 97–136. https://doi.org/10.1016/0010-0285(80)90005-5

Tsujimoto, S. (2008). The Prefrontal Cortex: Functional Neural Development During Early Childhood. *The Neuroscientist*, *14*(4), 345–358.

https://doi.org/10.1177/1073858408316002

Ullman, M. T. (2001). A neurocognitive perspective on language: The declarative/procedural model. *Nature Reviews Neuroscience*, 2(10), Article 10. https://doi.org/10.1038/35094573 Ullman, M. T. (2006). The declarative/procedural model and the shallow structure hypothesis. *Applied Psycholinguistics*, *27*(1), 97–105. https://doi.org/10.1017/S014271640606019X

Uppenkamp, S., Johnsrude, I. S., Norris, D., Marslen-Wilson, W., & Patterson, R. D.
(2006). Locating the initial stages of speech–sound processing in human temporal cortex. *NeuroImage*, *31*(3), 1284–1296. https://doi.org/10.1016/j.neuroimage.2006.01.004

- Valian, V. (2015a). Bilingualism and cognition. *Bilingualism: Language and Cognition*, *18*(1), 3–24. https://doi.org/10.1017/S1366728914000522
- Valian, V. (2015b). Bilingualism and cognition: A focus on mechanisms. *Bilingualism:* Language and Cognition, 18(1), 47–50.

https://doi.org/10.1017/S1366728914000698

van de Weijer-Bergsma, E., Wijnroks, L., & Jongmans, M. J. (2008). Attention development in infants and preschool children born preterm: A review. *Infant Behavior and Development, 31*(3), 333–351.

https://doi.org/10.1016/j.infbeh.2007.12.003

- Veer, I. M., Luyten, H., Mulder, H., van Tuijl, C., & Sleegers, P. J. C. (2017). Selective attention relates to the development of executive functions in 2,5- to 3-yearolds: A longitudinal study. *Early Childhood Research Quarterly*, 41, 84–94. https://doi.org/10.1016/j.ecresq.2017.06.005
- Videsott, G., Herrnberger, B., Hoenig, K., Schilly, E., Grothe, J., Wiater, W., Spitzer,
 M., & Kiefer, M. (2010). Speaking in multiple languages: Neural correlates of
 language proficiency in multilingual word production. *Brain and Language*, *113*(3), 103–112. https://doi.org/10.1016/j.bandl.2010.01.006

- Vince, M. A. (1949). Rapid response sequences and the psychological refractory period. British Journal of Psychology (London, England: 1953), 40(1), 23–40. https://doi.org/10.1111/j.2044-8295.1949.tb00225.x
- von Bastian, C. C., Souza, A. S., & Gade, M. (2016). No Evidence for Bilingual Cognitive Advantages: A Test of Four Hypotheses. *Journal of Experimental Psychology. General*, *145*(2), 246–258. https://doi.org/10.1037/xge0000120
- Von Wright, J. M., Anderson, K., & Stenman, U. (1975). Generalisation of conditioned
 GSRs in dichotic listening. In *Attention and performance: Vol. V* (P. M. A.
 Rabbitt&S. Dornic (Eds.), pp. 194–204). London: Academic Press.
- Walker, D., Greenwood, C., Hart, B., & Carta, J. (1994). Prediction of School
 Outcomes Based on Early Language Production and Socioeconomic Factors. *Child Development*, 65(2), 606–621. https://doi.org/10.1111/j.14678624.1994.tb00771.x
- Walsh, B. J., Buonocore, M. H., Carter, C. S., & Mangun, G. R. (2011). Integrating
 Conflict Detection and Attentional Control Mechanisms. *Journal of Cognitive Neuroscience*, 23(9), 2211–2221. https://doi.org/10.1162/jocn.2010.21595
- Wang, S. L. (1926). A demonstration of the language difficulty involved in comparing racial groups by means of verbal intelligence tests. *Journal of Applied Psychology*, *10*, 102–106. https://doi.org/10.1037/h0074356

Wartenburger, I., Heekeren, H. R., Abutalebi, J., Cappa, S. F., Villringer, A., & Perani,
D. (2003). Early Setting of Grammatical Processing in the Bilingual Brain. *Neuron*, *37*(1), 159–170. https://doi.org/10.1016/S0896-6273(02)01150-9

Wei, M., Joshi, A. A., Zhang, M., Mei, L., Manis, F. R., He, Q., Beattie, R. L., Xue, G., Shattuck, D. W., Leahy, R. M., Xue, F., Houston, S. M., Chen, C., Dong, Q., & Lu, Z.-L. (2015). How age of acquisition influences brain architecture in bilinguals. *Journal of Neurolinguistics*, *36*, 35–55. https://doi.org/10.1016/j.jneuroling.2015.05.001

Welford, A. T. (1952). The 'Psychological Refractory Period' and the Timing of HighSpeed Performance—A Review and a Theory. *British Journal of Psychology. General Section*, 43(1), 2–19. https://doi.org/10.1111/j.20448295.1952.tb00322.x

Welsh, M. C., & Pennington, B. F. (1988). Assessing frontal lobe functioning in children: Views from developmental psychology. *Developmental Neuropsychology*, 4(3), 199–230.

https://doi.org/10.1080/87565648809540405

- Westermann, G., Sirois, S., Shultz, T. R., & Mareschal, D. (2006). Modeling
 developmental cognitive neuroscience. *Trends in Cognitive Sciences*, 10(5),
 227–232. https://doi.org/10.1016/j.tics.2006.03.009
- Whitacre, J., & Bender, A. (2010). Degeneracy: A design principle for achieving robustness and evolvability. *Journal of Theoretical Biology*, 263(1), 143–153. https://doi.org/10.1016/j.jtbi.2009.11.008
- Wickens, C. D. (2007). Attention to the second language. 45(3), 177–191. https://doi.org/10.1515/iral.2007.008
- Wickens, C. D. (2008). Multiple Resources and Mental Workload. *Human Factors*, *50*(3), 449–455. https://doi.org/10.1518/001872008X288394
- Wikman, P., Sahari, E., Salmela, V., Leminen, A., Leminen, M., Laine, M., & Alho, K. (2021). Breaking down the cocktail party: Attentional modulation of cerebral

audiovisual speech processing. *NeuroImage*, 224, 117365. https://doi.org/10.1016/j.neuroimage.2020.117365

Winne, P. H., & Nesbit, J. C. (2010). The Psychology of Academic Achievement. Annual Review of Psychology, 61(1), 653–678. https://doi.org/10.1146/annurev.psych.093008.100348

Wise, L. A., Sutton, J. A., & Gibbons, P. D. (1975). Decrement in Stroop Interference
Time with Age. *Perceptual and Motor Skills*, *41*(1), 149–150.
https://doi.org/10.2466/pms.1975.41.1.149

Wong, D. D. E., Fuglsang, S. A., er, J. H., Ceolini, E., Slaney, M., & Cheveigné, A. de.
 (2018). A Comparison of Regularization Methods in Forward and Backward
 Models for Auditory Attention Decoding. *Front. Neurosci.* https://doi.org/10.3389/fnhum.2016.00604

Wood, H. (2016). Bilingualism is associated with better cognitive outcomes after stroke. *Nature Reviews Neurology*, *12*(1), 4–4.

https://doi.org/10.1038/nrneurol.2015.233

Wundt, W. M. (1912). An Introduction to Psychology, Tr. By R. Pintner.

Yang, J., Gates, K. M., Molenaar, P., & Li, P. (2015). Neural changes underlying successful second language word learning: An fMRI study. *Journal of Neurolinguistics*, 33, 29–49. https://doi.org/10.1016/j.jneuroling.2014.09.004

Zinszer, B. D., Chen, P., Wu, H., Shu, H., & Li, P. (2015). Second language experience modulates neural specialization for first language lexical tones. *Journal of Neurolinguistics*, *33*, 50–66. https://doi.org/10.1016/j.jneuroling.2014.09.005

Zion Golumbic, E. M., Ding, N., Bickel, S., Lakatos, P., Schevon, C. A., McKhann, G. M., Goodman, R. R., Emerson, R., Mehta, A. D., Simon, J. Z., Poeppel, D., & Schroeder, C. E. (2013). Mechanisms Underlying Selective Neuronal Tracking of Attended Speech at a "Cocktail Party". *Neuron*, 77(5), 980–991. https://doi.org/10.1016/j.neuron.2012.12.037

Zoefel, B., Archer-Boyd, A., & Davis, M. H. (2018). Phase Entrainment of Brain
 Oscillations Causally Modulates Neural Responses to Intelligible Speech.
 Current Biology: CB, 28(3), 401-408.e5.
 https://doi.org/10.1016/j.cub.2017.11.071

Zoefel, B., & VanRullen, R. (2015). The Role of High-Level Processes for Oscillatory Phase Entrainment to Speech Sound. *Frontiers in Human Neuroscience*, *9*, 651. https://doi.org/10.3389/fnhum.2015.00651